# A Concise Guide to Personality Disorders

Joel Paris

## Introduction

Personality disorders (PDs) are commonly encountered in practice, but their management is challenging. Patients with these diagnoses can be described as the "stepchildren" of the mental health professions, the patients we have to take care of, even if we would rather not. However, one message of this book is that some forms of PD have a good prognosis and many (but not all) patients can be treated effectively. These conditions are no more mysterious or problematic than psychoses, mood disorders, or any other major group of mental illnesses. However, treating this population requires a wide range of knowledge and skill in biological, psychological, and social domains.

To provide effective therapy, you first have to recognize the problem. In many patients, PD goes undiagnosed. Clinical psychologists may prefer to diagnose anxiety and depression because they have tools designed to

http://dx.doi.org/10.1037/14642-001 A Concise Guide to Personality Disorders, by J. Paris Copyright © 2015 by the American Psychological Association. All rights reserved.

deal with those problems. One has to wonder whether psychiatrists tend to avoid making these diagnoses because they prefer to prescribe medication. Yet all mental health professionals need to understand the difference between transient symptoms and lifelong dysfunction.

This having been said, making a diagnosis of PD requires a certain level of skill. The construct is defined by the *Diagnostic and Statistical Manual of Mental Disorders* (5th edition; *DSM*–5; American Psychiatric Association, 2013) as an enduring, inflexible, and pervasive pattern of inner experience and behavior that begins in late adolescence or early adulthood and that continues to impede functioning in work and relationships over many years. However, clinicians have to make judgment calls, such as how enduring the personality pattern is and the extent to which it impedes functioning. As this book will show, there is no absolute boundary between variations in normal personality and PD. However, some PDs have striking symptoms that may be mistaken for other categories of mental illness.

### WHY I WROTE THIS BOOK

The primary focus of my professional career has been patients with PDs. Some of my colleagues try to avoid these cases—in vain, because they are common in practice (unless, of course, you entirely ignore the role of personality in psychopathology). I also have colleagues who think these patients are just too difficult. I find the problems they present fascinating and challenging, but there is no easy fix, either through medication or psychotherapy.

When I began to treat patients with PD, I was not satisfied with what were then the most influential theories, explaining personality pathology as a result of an unhappy childhood. I saw many patients with traumatic childhoods who grew up to be normal. I also saw patients with severe PDs who had suffered little more than misunderstanding.

I was curious to find out more about PDs, so after almost 15 years of clinical practice and teaching, I started a second career as a researcher. This meant returning to my roots in psychology (my undergraduate major). I was fortunate to be able to collaborate with psychologists trained in

### INTRODUCTION

research. I had the added benefit of having practiced psychiatry on some of the sickest patients that mental health professionals see.

I entered the PD field just as it began to take off. The International Society for the Study of Personality Disorders was founded in 1987 and held its first meeting in 1988 (a 25th anniversary conference was held in 2013). The first issue of the *Journal of Personality Disorders* was published in 1987, and there are now two other print journals that focus on PD (*Personality Disorders: Theory, Research, and Treatment* and *Personality and Mental Health*). Attending scientific meetings over the years, I have met stimulating colleagues from all over the world, from whom I have learned much. Researchers in the same area tend to be spread around but form a kind of "invisible college"—we often see more of each other than colleagues from the same university. There are not many of us, only about 200 active PD researchers worldwide. I am proud to be of these happy few.

As a clinician, a teacher, and a researcher, I am also pleased to say that interest in PDs is definitely on the rise. Research on PDs has become much more active in the past 30 years, moving the field from clinical speculation to solid empirical investigation; since 1987, more than 20,000 articles reporting empirical research have been published. However, clinicians in practice, as well as many academics, continue to be reluctant to recognize PDs or to offer patients the specific forms of treatment they need. In this book, I try to explain why and give reasons why minds need to be changed.

Despite a rapidly developing knowledge base, PD remains mysterious in many ways. The goal of this book is to review what we know, what we don't know, and what the current state of knowledge implies for treatment. I aim to bring clinicians up-to-date with the latest research on PD and to suggest management strategies that are consistent with that evidence base. I will show that even if we cannot always provide definitive treatment for these patients, we can reduce distress and promote functioning.

This book differs in three important ways from previous guides to the management of PDs. First, it provides an evidence-based perspective. It makes no sense to depend entirely on clinical impressions when thousands of research papers have been published. Where there is strong evidence, I review the literature and suggest how it can be integrated into practice.

Where there is little or no literature, I say less and suggest where further investigation is needed. Although it is not possible to write a book on this subject without relying on clinical experience, I make clear what is science and what is opinion.

Second, this book focuses on clinical problems that have been examined in systematic research. Because borderline PD (BPD) is by far the best-studied disorder (nearly 7,000 articles published since 1987), it is the main subject of this book. Antisocial PD also has a large literature (nearly 5,000 articles over the same period), but I still have more to say about BPD because, as we now know, it is usually treatable. BPD is a problem that challenges clinicians, but there is strong and growing research supporting effective therapy of these patients. When a larger body of data emerges on other PDs, we may develop treatment packages for them as well.

Third, this book argues that clinicians need to replace systems of therapy identified by acronyms and associated with charismatic founders. Even if some of these methods are evidence based, we need to adopt integrative, eclectic methods that combine the best ideas from all schools. A single paradigm, called *psychotherapy*, is needed to heal these divisions. An integrative approach is well supported by research and also makes the most practical sense.

### THE STRUCTURE OF THIS BOOK

This book consists of three parts. The first is devoted to general issues about PDs. Chapter 1 addresses thorny problems of definition and the uncertain boundary between PD and personality. Chapter 2 focuses on the relationships between traits and disorders and assesses the advantages and disadvantages of proposals to revised or replace the *DSM* system. Chapter 3 examines etiology and risk factors, and Chapter 4 reviews prevalence and outcome.

The second part of the book is devoted to specific PD categories: Chapter 5 on antisocial PD, Chapter 6 on BPD, Chapter 7 on narcissistic PD, Chapter 8 on other PDs (schizotypal, schizoid, paranoid, histrionic, avoidant, obsessive—compulsive, dependent, and PD, not otherwise specified).

### INTRODUCTION

The third part of the book concerns treatment. Chapter 9 examines the efficacy of pharmacotherapy, and Chapter 10 reviews evidence concerning the various psychotherapies developed for PD. Chapter 11 presents a general approach to management, with emphasis on the borderline category. Finally, Chapter 12 summarizes what we know about PD and what we need to find out.



# Why the Diagnosis of Personality Disorder Is Difficult

This chapter provides an overview of problems in the diagnosis and classification of personality disorders (PDs). Its overall theme is that PDs are both underdiagnosed and underrecognized in clinical practice.

### PDs CARRY STIGMA

The PD construct can meet with a surprising level of resistance, sometimes approaching hostility. At the 2011 Annual Convention of the American Psychiatric Association, a well-known psychoanalyst-researcher told an audience that diagnosing patients with borderline PD (BPD) is no more valid than calling them "jerks." In another session at the same meeting, a biological psychiatrist who prefers to see BPD as a form of "bipolarity" stated that a diagnosis of PD is an insult he would never dream of inflicting on a patient.

http://dx.doi.org/10.1037/14642-002

A Concise Guide to Personality Disorders, by J. Paris
Copyright © 2015 by the American Psychological Association. All rights reserved.

Why does PD arouse so much opposition? Stigma is certainly part of the answer, but isn't there at least as much stigma associated with psychosis or depression? The answer is, not necessarily. Stigma is rooted in the fear of having a mental illness oneself. When I try to explain my work to other professionals, the predictable response is, "How do I know I don't have one of these diagnoses?" Most of us are sure we are not psychotic. We are not so sure we don't have a PD. Rejecting the construct protects us from that possibility.

Mental disorder has always carried stigma and probably always will (Corrigan, 2000). Losing an arm or a leg is sad, but it doesn't affect the integrity of the self. In contrast, losing one's mind is much more threatening. That is why being called "crazy" has always been an insult. The implication is that you are somehow *responsible* for losing your mind.

We all have a personality, but we may wonder, based on feedback from others, whether we have a bad one. The concept of PD is threatening because we are unable to protect ourselves from this specter. According to a concept called a *fundamental attribution bias* (Jones & Harris, 1967), we see other people's misbehavior as reflecting their personality, whereas our own mistakes only reflect difficult situations. This is why PD constructs can become insults. People who are selfish or have big egos may be called narcissistic, whereas perfectionists may be called obsessive—compulsive. Can any of us be sure we do not have a PD? Evidently not.

The other part of the reluctance to diagnose PD is the perception that these disorders are incurable. That is why clinicians prefer to focus on symptoms for which they have effective tools to treat. It is certainly true that PDs begin early in life and affect psychosocial functioning over many years. However, research shows that most patients get better with time (Gunderson et al., 2011). It also shows that patients with BPD, once thought incurable, are much more treatable than previously thought (Paris, 2010a).

I aim to reduce stigma when I explain PD to patients. I tell them that the diagnosis describes people who can't get their life together, either in relationships or at work. This capsule message usually gets across because it fits with what they already know about themselves. It also offers hope. I inform patients that they will probably improve with time but that treatment can make recovery go faster. Both statements are evidence based.

### WHY THE DIAGNOSIS OF PD IS DIFFICULT

The problem is that many, if not most, patients with PDs are misdiagnosed (Zimmerman, Dalrymple, Chelminski, Young, & Galione, 2010). They most often tend to be seen, at least initially, as having mood disorders. PD is only considered when treatment for depression or bipolarity fails.

An even more serious problem is that patients with PDs are being treated inappropriately (and unsuccessfully) with medications. This is not entirely the fault of physicians or of the psychologists who request medical consultation. Unfortunately, by the time patients meet mental health professionals, they may have been socialized to consider themselves as suffering from "chemical imbalances." For this reason, some patients may insist on retaining a mood disorder diagnosis and request a medication cocktail. Another current trend in practice is diagnosing PD patients with attention-deficit/hyperactivity disorder (ADHD), which also leads to a prescription. Those who have failed to respond to several medication trials may be more open to accepting a diagnosis of PD.

I encourage patients, if they have not already done so, to use the Internet to read about PD. Doing so can make them feel validated. Many are actually relieved to receive a diagnosis that corresponds to their life experience. They know intuitively that their personality doesn't work and that depression is the result, not the cause, of their troubles. Many have had the experience of being unsuccessfully treated for other diagnoses with inappropriate forms of treatment.

Successful treatment of PD would be more common if the educated public knew more about the subject, but explaining what *personality disorder* means to nonprofessionals is challenging. On one occasion, I was interviewed on the radio and offered narcissistic PD as an example. My host replied, "But I'm also a bit of a ham!" More recently, I was asked to explain my research to a group of basic scientists working at my hospital. Once again, the almost universal reaction was defensive and fearful: "How do I know that this doesn't describe me?" I have learned to concentrate on the self-harm and suicide attempts associated with BPD so that everyone knows the subject is mental illness, not ordinary life. Even so, stigma gets in the way of the message.

If the fear of losing control of one's own mind is the main cause of the stigma associated with mental disorders, then PDs have to be particularly

threatening. When disorders are exaggerations of normal traits, we recognize ourselves, albeit in a distorting mirror.

Even when PD presents as an alien state, readily recognized as an illness, the construct does not always meet with acceptance. For example, emergency department doctors sometimes hate patients with BPD. I have heard them say, "I am already busy with so many people who are sick for no fault of their own, so why should I have to spend time treating patients whose illness is self-inflicted?" What these professionals are missing is that these patients experience intense suffering, which they express through impulsive actions.

# WHY PD CANNOT BE EASILY EXPLAINED AT THE LEVEL OF NEUROSCIENCE

The neuroscience model that dominates the contemporary practice of psychiatry works against the recognition and treatment of PDs. Ironically, reliance on brain research was a strategy developed by the mental health professions to get around stigma. Psychiatry now views mental disorders as brain disorders, and its most influential paradigm redefines practice as the clinical application of neuroscience (Insel & Quirion, 2005). Yet this point of view leads to problems, most obviously to the tendency to prescribe drugs to all patients.

Clinical psychology has also been influenced by cognitive neuroscience, which has provided evidence that therapy works by changing brain function (Trull & Prinstein, 2012). Unfortunately, some clinical psychologists have adopted the same point of view as psychiatrists, sending more and more patients in therapy to physicians for consultation—with the almost guaranteed result that medication will be prescribed.

Although it is true that mental illness is a product of the brain, it does not follow that disorders can readily be explained on a neural level. As later chapters in this book show, there are biological risk factors for PDs, but their effects depend on their interaction with the environment. Biological reductionism fails to acknowledge the importance of psychosocial factors in development. A widely quoted article by Insel and Quirion (2005) suggested that psychiatry abolish itself by becoming a part of neurology, but

### WHY THE DIAGNOSIS OF PD IS DIFFICULT

the words *psychology* or *psychotherapy* cannot be found in the text. So much for the biopsychosocial model!

Reductionistic theories, in which mental disorders are seen as "nothing but" brain disorders, remove agency from patients. In a fully deterministic model, psychotherapy would be next to impossible. Patients may not be considered responsible for getting better but are viewed as victims of chemical imbalances and aberrant neurocircuitry. To recover, patients would only be asked to be passive recipients of pharmacological treatment.

This model is reasonably appropriate for the psychoses, in which theory and practice would benefit from more, not less, data based on neurobiology. It is wrong, however, for other mental disorders, such as mild to moderate depression, substance abuse, and PDs. In each of these conditions, patients are not the victims of chemistry but of choices. People who fail at work and relationships have cognitive misconceptions, difficult emotions, and problematic behaviors that cannot be reduced to the activity of neurotransmitters. It is also wrong from a philosophical point of view. Therapists cannot work effectively with patients without believing in the existence of free will.

Our current knowledge base about psychopathology is, unfortunately, insufficient to explain mental illness (Uher & Rutter, 2012). That conclusion certainly applies to PD, which remains in so many ways a mystery. However, we do know enough to say that PDs, although partly shaped by biological factors, are strongly influenced by psychological and social risks. The attempt to take stigma away from PD by making it an epiphenomenon of neurobiology oversimplifies an enormously complex problem.

### PDs ARE OFTEN MISDIAGNOSED

Despite its clinical importance, the diagnosis of PD is often missed. If practitioners are not well trained to manage these patients, they may prefer to call them something else. Another reason for missing PD is that clinicians tend to have a preference for diagnoses that are currently "hot."

Let us consider some examples. First, because most patients who see therapists are notably unhappy, it is tempting to diagnose them with depression. The *Diagnostic and Statistical Manual of Mental Disorders* 

(5th edition; *DSM*–5; American Psychiatric Association, 2013) criteria set the bar much too low for a diagnosis of major depression (five of nine symptoms for only 2 weeks), greatly encouraging overdiagnosis (Paris, 2013a). Focusing on episodes of low mood in patients with PD ignores more important long-term behavioral patterns. By definition, these patients have had problems for most of their lives.

Second, the faddishly popular diagnosis of bipolar disorder has been used to account for many of the phenomena associated with PD, most particularly moodiness (Paris, 2012). If you follow the guidelines in *DSM*–5, bipolarity requires clear periods of mania or hypomania, but clinicians may invoke this diagnosis in all patients who only have a mood instability that indicates emotion dysregulation.

Third, the diagnosis of posttraumatic stress disorder (PTSD) has been applied to patients who have been exposed to significant trauma. Yet this preference fails to take into account that PTSD has a characteristic symptom profile and that most people who have experienced trauma do not develop the disorder (McNally, 2003). Applying this label to every patient who has experienced significant adversity strays from diagnostic precision and is misleading.

Fourth, ADHD, another diagnosis that has become a fad, is currently being used to explain a wide variety of clinical problems (Batstra & Frances, 2012). Even though many other conditions, such as anxiety and depression, affect attention, it is tempting to make a diagnosis believed to respond predictably to stimulant medication.

Specialists in any group of mental disorders tend to be enthusiastic about their ideas. They may see the conditions they study everywhere and diagnose them all too freely. This is the case for all the currently popular diagnoses just mentioned. As a PD specialist, I try to bend over backward to avoid overdiagnosis, particularly when problems are situational rather than lifelong. If everyone who has serious trouble with work or relationships is seen as having a PD, at least half of us could meet criteria, making the concept so broad that it would lose meaning.

The misdiagnosis of PD feeds into the reluctance to make these diagnoses, even when they are obvious. PDs may be complex, but they can describe clinical pictures parsimoniously. Why give patients multiple

### WHY THE DIAGNOSIS OF PD IS DIFFICULT

symptomatic diagnoses when a single PD category covers most of the same ground? Moreover, some PDs, most particularly BPD, describe patients who are treatable. Making the diagnosis provides information that can be used to guide treatment planning.

### PDs ARE COMPLEX

One of the main sources of resistance to diagnosing PD is the complexity of the construct. Mood is simple, but personality is complicated. By and large, simplicity trumps complexity and leads to ideas that are clinically seductive.

The ideology of neuroscience is a good example. It favors disordered chemistry or neuroconnectivity (or a combination of these) as explanations for mental illness. Although these mechanisms are not really simple, they simplify the problem by being reductionistic. Considering interactions between brain mechanisms and environmental factors introduces a much greater level of complexity. We now know that neurobiology is not just a given and that experience changes the brain (Rutter, 2006).

Even at a neural level, the task of accounting for behavior is, to say the least, formidable. The brain is unbelievably complex, with 100 billion neurons and trillions of synapses. This makes the utility of reducing mental disorder to cellular mechanisms doubtful. It makes more sense to study the mind at its own level while keeping neural mechanisms well in mind.

PD can only be understood at multiple levels of analysis by examining complex interactions between temperament and life experiences. The attempt to turn complexity into simplicity leads to serious clinical problems. A biomedical approach, offering drugs for every symptom, ignores the cognitive, emotional, and behavioral patterns that make patients unhappy. Psychologists who focus only on symptoms also do so at the expense of understanding life histories. Thus, even if patients with PDs are sad, moody, or fearful—as they often are, given their troubled lives—they cannot be treated with standard methods developed for common mental disorders.

One often hears that PDs may remit when mood disorders are treated, so one should "defer" making these diagnoses when patients are depressed.

Actually, several lines of evidence refute that conclusion. One is that patients who also have a PD do not respond well to standard treatment for depression (Newton-Howes, Tyrer, & Johnson, 2006). Another is that patients with PDs almost always continue to meet criteria even when depression remits (Lopez-Castroman et al., 2012). These findings contradict the idea that PDs are nothing but atypical mood disorders, as some researchers have proposed (Akiskal, 2004). If only it were true that antidepressants and mood stabilizers could put PDs into remission! As a recent Cochrane report showed, research consistently indicates that they cannot (Stoffers et al., 2010). It is therefore better to use a complex construct like PD that does justice to clinical problems than to replace it with a simpler one, like mood disorder, that does not.

To differentiate PD from mood disorder, one needs to take a careful history of life events over many years. Patients with mood disorders have episodes of illness that derail their functioning for defined periods of time. In contrast, patients with PDs have had serious problems continuously for years, usually beginning in adolescence (or earlier). Patients with mood disorders suffer from depressive (or hypomanic) episodes that temporarily interfere with their lives but leave the self intact. Patients with PDs have pathology that lies in the self and may never have been "normal" enough to get "back to normal."

# SEPARATING MILD PDs FROM NORMAL VARIATIONS IN PERSONALITY

For PD to be considered a legitimate diagnosis, it must be as well defined as any other mental disorder and distinguished from normal variations in personality. Strange to say, the *DSM* did not include a formal overall definition of PD until its fourth edition (American Psychiatric Association, 1994). That definition, retained in *DSM*–5, emphasizes that PD is chronic, beginning in late adolescence or early adulthood, and impedes functioning in work and relationships over many years. Unfortunately, language such as *enduring*, *inflexible*, and *pervasive* gives the impression of incurability. As this book shows, that is far from the case.

### WHY THE DIAGNOSIS OF PD IS DIFFICULT

One of the problems with the DSM system is the way it defines PDs as categories. There are 10 of them in the DSM-5, of which only two or three (borderline, antisocial, and schizotypal) have solid research behind them. Moreover, anyone who has a PD that does not fit any of the 10 has to be diagnosed as personality disorder, not otherwise specified (now called in DSM-5, PD, Unspecified). This is the most common diagnosis in practice (Coccaro, Nayyer, & McCloskey, 2012), because only about half of all patients who meet overall PD criteria meet specific criteria for the currently listed categories (Zimmerman, Rothschild, & Chelminski, 2005). Of those who do meet criteria, patients in clinical settings often meet criteria for more than one (Nurnberg et al., 1991). This shows that the current categorical system is not valid. Although every clinician can recognize diagnoses when they are typical and paradigmatic, many if not most patients fall between the cracks of the system. Moreover, given the polythetic nature of the system (i.e., requiring only a limited number of symptoms from a longer list), people with the same diagnosis can be quite different from each other.

One solution to this problem is to view PDs as *dimensional*, that is, as pathological amplifications of traits with a continuous relationship to normality. Trait psychology sees PDs in this way, rooted in the domains identified by research on normal personality. Using these models, one can quantify personality dimensions so that diagnosis depends on a cutoff that defines a point at which trait amplification reaches a pathological level. The most widely used schema is the five-factor model (FFM; Widiger & Costa, 2013), which has been the subject of an enormous amount of research. If PD represents an exaggerated set of personality traits that lead to dysfunction, then the FFM can be adapted for clinical use to describe patients with more precision than categorical diagnoses.

Why have clinicians been somewhat hesitant about adopting dimensional assessment as a diagnostic tool? The first reason is that practitioners are used to the algorithm-based categories of the *DSM*. Some of these categories have a strong research base, and diagnosis can be a distinct help in guiding treatment. A second is that even though quantitative data provide more information, clinical decisions are categorical. A third is that given

the time restrictions of practice, it may not always be practical to give patients questionnaires, even in computerized forms.

The dimensional approach has a large body of research in normal populations, but it may or may not be a useful guide to treatment in clinical settings. Almost all research thus far on the treatment of PDs has concerned patients with categorically diagnosable disorders (e.g., BPD), although it is possible that specific psychotherapies based on dimensions could eventually be developed. Categorical diagnoses of PD do not tell us how to conduct therapy but often point to the likelihood of failure to respond to certain types of interventions, in both pharmacological treatment and psychotherapy (Newton-Howes et al., 2006).

The larger problem is that all mental disorders require a cutoff between normality and pathology. Sometimes the difference is obvious, as when patients are psychotic or suffer from a profound depression, but in less severe disorders, the cutoff for diagnosis may not be so clear. In such cases, the question arises as to whether one is looking at a mental illness or a normal psychological reaction.

Wakefield (1992) proposed a way to cut this Gordian knot. His idea was that patients with mental disorders must be characterized by *harmful dysfunction*. In other words, problems have to involve distress or impairment, as well as the failure of a psychological mechanism to perform its naturally designed function. Determining harm and dysfunction still requires judgment calls, however, based more on clinical experience than on empirical data demonstrating construct validity.

This is also a problem in the clinical applicability of trait dimensional models. Almost anyone might be diagnosed with a PD if he or she has significant dysfunction in a major area of life. (Most of us have at least some degree of dysfunction in work or relationships.) The more broad the definition, the more likely it is that people with normal variation will be labeled as disordered. A definition of disorder that is too broad would also not be a useful guide to treatment.

A more precise definition of PD would be needed to describe homogeneous groups of patients. For example, the decision as to whether traits interfere with function in a clinically significant way could be arbitrary.

### WHY THE DIAGNOSIS OF PD IS DIFFICULT

Keeping cutoff points conservative helps to prevent diagnostic inflation. A narrower definition of PD would also be important for research. If criteria are applied scrupulously, then patient groups become more homogeneous, or at least have a heterogeneity that can be described systematically. Yet when patients suffer from only mild to moderate dysfunction, establishing a diagnosis is much more difficult. I return to this problem in Chapter 3, which critically examines research on the prevalence of PD. In summary, separating PD from normal variations in personality is problematic. When dysfunction is severe, the diagnosis will be clear. It is not so clear when dysfunction is mild to moderate. These problems are examined in more detail in the next chapter.

# Traits, Disorders, and the *DSM*–5

This chapter examines where the boundaries lie between personality traits and disorders, whether categorical or dimensional systems are more valid, and whether the proposals that were made for the *Diagnostic* and Statistical Manual of Mental Disorders (5th ed.; DSM–5; American Psychiatric Association, 2013) would have had clinical benefit.

### BOUNDARIES BETWEEN TRAITS AND DISORDERS

Everyone has a personality. *Trait* profiles describe behaviors, emotions, and cognitions that are unique to individuals. There is a good evolutionary reason why these characteristics should vary from one person to another. Like physical variations, traits are designed by natural selection to make survival more likely. However, the environment varies so much that traits that are adaptive in one setting will be maladaptive in others. Variation in nature is a hedge that makes survival possible in a range of environments.

http://dx.doi.org/10.1037/14642-003 A Concise Guide to Personality Disorders, by J. Paris Copyright © 2015 by the American Psychological Association. All rights reserved.

Thus, each of us has a unique personality profile, which is functional under some conditions but not others (Beck & Freeman, 2002). In other species (Réale, Reader, Sol, McDougall, & Dingemanse, 2007), aggressiveness and risk taking, or timidity and behavioral inhibition, can either be adaptive or maladaptive, depending on circumstances. The same principle applies to human beings. People who develop antisocial personality disorder (ASPD) in a deprived urban environment may have traits that could, in another context, make them into good soldiers. People whose shyness prevents them from finding a mate in modern society might do perfectly well in cultures where marriage is arranged.

Personality traits are usually measured using self-report questionnaires that have been validated in large community samples. There are several currently in use, but the largest body of research by far is on the five-factor model (FFM), which has also been adapted in a briefer version for clinical use (Widiger & Costa, 2013). The Big Five factors are *Neuroticism*, *Extraversion*, *Agreeableness*, *Conscientiousness*, and *Openness to Experience*, each of which can be divided into more specific facets. The Big Five have been shown to be valid in both community and clinical populations, in all social classes, and across cultures.

The main limitation of instruments that measure traits is their reliance on self-report. One cannot simply assume that people can describe themselves accurately, and some researchers have used informants to confirm these reports (Widiger & Costa, 2013). Kagan (2012) suggested that personality traits would best be described by using a convergence of multiple measures—questionnaires, ratings by other people, direct observation, and biological markers. Because multiple procedures are expensive, however, almost no one carries out research in that way.

Trait psychology also assumes there is no clear boundary to determine when traits end and disorder begins. In this view, personality disorder (PD) reflects an extreme and dysfunctional trait profile (Livesley, 2003). Thus, PD would be an analogue to medical phenomena such as the relation of blood sugar to diabetes or of blood pressure to hypertension. In these cases, one can mark a threshold for disorder on the basis of an increased likelihood of dysfunctional complications. However, there is no way to establish a definite cutoff between traits and disorders with external validity.

Another problem is that, as discussed in the previous chapter, if a dimensional approach defines PD so broadly, it might be hard to see who wouldn't have one. This would undermine the goal of getting clinicians to recognize these disorders. This is the same problem I ran into trying to explain PD to nonpsychiatric colleagues (maybe they had a point!). It is analogous to the question of the difference between sadness and depression (A. V. Horwitz & Wakefield, 2007) or between fear and anxiety (A. V. Horwitz & Wakefield, 2012). If we accept the broad definition of major depression used in the *DSM* system, in which 2 weeks of distress would be sufficient for a diagnosis that would affect half of the general population over a lifetime (Moffitt et al., 2010). This shows just how difficult it is to separate psychopathology from normality. Differences in degree eventually become differences in kind, and clinicians need categories to guide their treatment choices.

# EGOSYNTONIC AND EGODYSTONIC ASPECTS OF PD

Much ink has been spilled, and much emotion expressed, as to whether PDs should be classified in terms of categories or dimensions. Those favoring categories see the dimensional alternative as obscure and lacking in clinical relevance. Those favoring dimensions see categories as arbitrary and unscientific. My own view is intermediate: Both systems have value, but we do not know enough to choose between them. At this point, although traits provide more information, some categories (borderline and antisocial) capture clinical syndromes that need to be identified.

Some of the confusion about the best way to classify PDs derives from the fact that there are two distinct aspects of PD. This distinction is based on the well-known description of mental disorders as either *egodystonic*, in which patients have symptoms they find troubling and alien, or as *egosyntonic*, in which patients consider their traits or symptoms to be normal (Hirschfeld, 1993). For example, obsessive—compulsive disorder (as opposed to obsessive—compulsive PD [OCPD]) is characterized by highly egodystonic thoughts, whereas patients with anorexia nervosa have an egosyntonic belief that their obsession with thinness is perfectly normal.

Applying this model to PDs, the more egosyntonic a disorder, the more readily it is classified dimensionally; the more egodystonic a disorder is, the more one should frame its psychopathology as a category. Thus, some disorders are almost entirely based on traits that are exaggerated but egosyntonic. The best examples are narcissistic PD (NPD) and OCPD. These categories describe people with amplified traits that get them into trouble in life but who do not usually recognize the nature of the problem. Patients with these diagnoses are insufficiently self-critical, often wondering why other people are not more like them. As described later in this book, in both disorders, there is a smooth and imperceptible transition between trait domains and diagnosable psychopathology.

In contrast, other PDs have predominant egodystonic features that are not fully explained by a trait profile. These symptoms, although associated with pathological traits, can describe a clinically important difference between trait domains and diagnosable disorders. The best example is borderline PD (BPD), in which patients are high in negative affectivity and impulsivity (Siever & Davis, 1991) and suffer from a wide variety of symptoms. Another example is avoidant PD, in which social anxiety produces a painful loneliness that leads patients to seek treatment (Sanislow, da Cruz, Gianoli, & Reagan, 2012).

There is some evidence that PDs with egodystonic features are, at least partially, discontinuous from their underlying trait profiles. In principle, the symptoms of any mental disorder can be scored dimensionally by considering them as traits (Krueger & Tackett, 2006). These models of psychopathology, in contrast to categorical systems, suggest a smooth progression from normality to pathology, and the absence of points of rarity among traits, subclinical phenomena, and diagnosable disorders (Kendell, 2002). Yet dimensional scores, such as the measurement of blood pressure in medicine, are still subject to categorical cutoffs. This is because clinical decision making in mental health practice remains a dichotomous procedure in that treatment decisions are usually based on the presence or absence of a defined condition. Moreover, severe mental disorders are generally heritable (Kendler & Prescott, 2006), have a more severe outcome than subclinical syndromes (Bedhiran & Sartorius, 1995), and often have a more specific response to treatment.

Because traits have stronger correlations than categories with neurobiological measures (Insel et al., 2010), patients with PDs were considered as "poster children" for a dimensional approach in the proposals for *DSM*–5. However, there are no known consistent biomarkers for personality traits.

Moreover, studies of trait dimensions show that they sometimes capture only a portion of the variance in PD. Using data from the Longitudinal Study of Personality Disorders (a large sample of college students followed prospectively), Wright, Pincus, and Lenzenweger (2012) conducted regressions of traits on PD symptoms The results showed that considering symptoms as normally distributed resulted in a violation of model assumptions (too many subjects scored zero) and that results differed according to whether one was predicting the presence of PD or its severity.

Similarly, in a study in which PD diagnosis was scored by clinical ratings of *DSM* criteria (Zimmerman, Chelminski, Young, Dalrymple, & Martinez, 2013), dimensional scoring, based on counts of *DSM* criteria, added information in subclinical cases but not in cases without a diagnosis. This finding contradicts the assumption of a smooth progression from trait profiles to clinical pathology and suggests discontinuities between traits and PDs.

Also, the FFM may not be sufficiently sensitive to symptoms that are relatively uncommon in community samples, even when they are clinically important (Clark, 2007). Two studies using this system (Bagby, Costa, Widiger, Ryder, & Marshall, 2005; Morey et al., 2007) found that trait domains only partially accounted for the clinical features of PDs, particularly in disorders such as BPD, which are marked by egodystonic symptoms. Morey and Zanarini (2000), who also studied patients with BPD, reported that although the FFM's neuroticism scale accounted for some of the variance, it did not account for the most symptomatic aspects of the disorder (e.g., a wide range of impulsive behaviors).

All PDs share a common profile at the level of FFM trait domains, with high Neuroticism, low Conscientiousness, and low Agreeableness (Morey et al., 2007; Saulsman & Page, 2004). A more complex analysis of trait facets could be needed to account for clinical symptoms associated with PDs (Widiger & Mullins-Sweatt, 2009). We do not know if this approach is feasible or has clinical utility.

To resolve this problem, self-report instruments have been specifically designed to measure the trait dimensions that underlie PD. The Schedule for Nonadaptive and Adaptive Personality (Simms & Clark, 2006) and the Diagnostic Assessment of Personality Pathology (DAPP; Livesley, Jang, & Vernon, 1998) overlap with the FFM in most respects, and the DAPP has been factor analyzed into four domains that closely resemble the FFM. Both instruments differ from the FFM in that they make a point of assessing egodystonic symptoms such as self-harm. That behavioral pattern can be found in community populations (Kessler, Berglund, Borges, Nock, & Wang, 2005) but is more intermittent and not associated with the development of PD (Klonsky, 2007).

Measures of traits based entirely on self-report may not be ideal to assess features of PDs that are not seen as problematic by patients. For example, Cooper, Balsis, and Oltmanns (2012) found that a diagnosis of NPD was more accurate if informants were interviewed about patterns that patients themselves failed to recognize. Sometimes, observations scored by trained observers can pick up aspects of psychopathology that questionnaires miss, as Perry (1992) found when comparing of self-report instruments with semistructured interviews for diagnosis of PDs.

Other research supports the overall conclusion that PDs are an amalgam of traits and symptoms (Skodol et al., 2005). Long-term follow-up of patients with several PDs over 10 years (Gunderson et al., 2011) and of patients with BPD over 14 years (Zanarini, Frankenburg, Reich, & Fitzmaurice, 2012) has shown that these disorders have a unique trajectory over time in that traits remain stable, whereas symptoms tend to remit.

Egosyntonic and egodystonic features are overlapping but conceptually distinct domains of PD that are different in different categories of disorder. For example, BPD is a complex form of psychopathology that reflects multiple endophenotypes (Paris, 2007a) and that straddles externalizing and internalizing spectra of mental disorders (Røysamb et al., 2011). Thus, although BPD is rooted in trait profiles, it cannot be entirely reduced to them. The affective instability and impulsivity associated with this disorder does not explain why patients end up overdosing or wrist cutting. The construct of BPD goes beyond the quirks that differentiate

one person from another. It describes problems that interfere with getting a life and keeping one.

Similarly, patients with avoidant personality have striking egodystonic features. Given their sensitivity, these patients feel they are right to fear social rejection. Even so, they may ask for help to overcome loneliness and isolation (Sanislow, da Cruz, Gianoli, & Reagan, 2012).

In contrast, let us consider two examples of PD categories that reflect extreme levels of traits, with few egodystonic symptoms. These categories can be understood as amplified and dysfunctional trait profiles, just as a dimensional model would predict.

The best example is NPD. Research suggests that high levels of trait narcissism are essentially equivalent to a diagnosis of NPD (Paris, 2013b; Pincus & Lukowitsky, 2010). The disorder is characterized by a grandiose view of the self, and patients meeting these criteria tend to maintain their self-image by blaming others for their problems. This makes the disorder highly egosyntonic, although patients can eventually be unhappy enough to seek treatment (Ronningstam, 2010).

A second example is OCPD. This construct is characterized by high levels of conscientiousness and perfectionism, and the disorder is essentially equivalent to high levels of compulsive traits (de Rues & Emmelkamp, 2012). OCPD patients may consider their own personality normal, regarding others as careless or sloppy (Samuel & Widiger, 2011). Thus the disorder tends to be egosyntonic, but because perfection is impossible, compulsive traits can create interpersonal difficulties.

Similar principles can be applied to other PDs with prominent egosyntonicity. Patients with paranoid PD believe they are perfectly right to be suspicious, and patients with schizoid PD are comfortable with being socially isolated. Patients with dependent PD believe they have a right to ask other people to take care of them. All these categories are exaggerated and dysfunctional traits that can readily be dimensionalized (Widiger & Costa, 2013).

ASPD is difficult to place in this model: It has mainly egosyntonic features, yet some of its egodystonic features resemble BPD (Gunderson, 2013; Paris, Chenard-Poirier, & Biskin, 2013). Thus, comorbid anxiety and mood

disorders are associated with a significant risk (approximately 5%) for completed suicide (Ullrich & Coid, 2009). Patients with ASPD are well known for not seeking help and for justifying the most problematic behaviors. In its most severe form, psychopathy (Hare, 1999), there is little suffering. Even so, ASPD is a complex disorder, describing amplified trait dimensions (low conscientiousness, low agreeableness) as well as features (chronic criminality) that are qualitatively different from behaviors in normal population (Black, 2013a).

### DSM-5: THE STORY OF A CONTROVERSY

The classification of PDs has long been problematic. Most patients don't fit existing categories, but dimensional scores don't always do justice to severe psychopathology.

Years before the *DSM*–5 process was launched, the battle lines were already drawn. In 2004, I attended a 2-day conference in Washington, DC, at which the American Psychiatric Association asked a group of researchers to come up with a dimensional model for PD. Nobody could agree on how to do this. Some were proponents of the FFM, others favored dimensional models that included more measures of psychopathology, and others felt that the most well-researched categories should be retained. My view was that we didn't yet know enough to replace a familiar model, however flawed, with a new system that would have uncertain validity and clinical utility.

In the absence of any consensus, the next 9 years were marked by controversy, leading to an epic struggle. The *DSM*–5 Task Force, led by a mood disorder researcher (David Kupfer from Pittsburgh) and an epidemiologist (Darryl Regier from the American Psychiatric Association), wanted a dimensional system of diagnosis because they believed that quantitative scores are closer to neurobiology. Trait psychologists had been arguing for such a system for decades.

The *DSM*–5 Personality Disorders Work Group was therefore encouraged to come up with a dimensional model, but it could not ignore the large number of researchers who favored categories and who had based their careers on studying categories. The group also made some political

errors. One was that it excluded the people who had worked on the fourth edition of the *DSM* (*DSM–IV*; American Psychiatric Association, 1994), most of whom weren't even consulted. (When *DSM–IV* was being prepared, I wasn't on the committee but, along with many others, was sent a copy of the draft and invited to make comments. Although I am sure this process had no impact on the outcome, it made me feel included.)

Total exclusion forced those who were not on committees to state their objections in journals. The most important example was a paper in *The American Journal of Psychiatry* (Shedler et al., 2010), which made it clear that all these experts considered a dimensional system a potential disaster. The fact that so many of the leading researchers on PD were against the work group's proposals was a major factor in its ultimate rejection.

Who, then, was included in the *DSM*–5 process? Membership on the work group was wide ranging, including neuroscientists, clinical psychiatrists, and trait psychologists. I have great respect for the members of this group, although not all had extensive clinical experience. Another twist in the story is that some members, including the chair, Andrew Skodol, had been involved with the Collaborative Longitudinal Study of Personality Disorders, a multimillion-dollar project funded by the National Institute of Mental Health (Skodol et al., 2005). This was the largest study ever conducted on PD, and it examined long-term outcomes for four categories (BPD, schizotypal PD, avoidant PD, and OCPD). It may not have been an accident that none of these were ever slated for removal.

To the credit of the *DSM*–5 Task Force, all proposals for change in the manual were posted on the Internet in February 2010. This procedure allowed nonmembers to comment, either in print or among themselves, on these ideas. Along with a number of others, I cosigned a letter to the PD work group outlining the problems with the proposal. However, this kind of feedback was totally ineffective. Committees have a life of their own, and their move to consensus has sometimes been called "groupthink." Members whom I had previously known to be agnostic about simple solutions to the problems of PD classification became almost religiously attached to the proposals developed by the work group and were bitterly disappointed when they failed.

### THE DSM-5 HYBRID PROPOSAL

A great deal of work went into creating a new classification of PDs for *DSM*–5 that would be based on a clinical assessment of personality traits. It was proposed that diagnosis should be rooted in two basic features that can be scored dimensionally: significant impairments in self (identity or self-direction) and interpersonal (empathy or intimacy) functioning, as well as one or more pathological personality trait domains or trait facets. As in *DSM*–*IV*, impairments in personality functioning and the individual's personality trait profile would have to be relatively stable across time and consistent across situations. There was a problem, however: Although most of the research on trait dimensions has used self-report, *DSM*–5 proposed that *clinicians* would be the ones to rate both traits and functioning, using a set of standard scales. Yet such ratings might not be reliable. The proposal was also a hybrid in that it allowed for dimensional diagnosis but allowed clinicians to infer categorical diagnoses from trait profiles.

The work group had not wanted to take sides in the long-running categorical—dimensional dispute, so a hybrid system attempted to combine both approaches, building categories on ratings of trait dimensions. Most of the procedures were dimensional—scores for PD characteristics as a whole, scores for functional levels, and ratings of trait profiles to define specific disorders. Six of the 10 categories listed in *DSM–IV* were retained, with two (schizoid and paranoid) folded into schizotypal PD and another two (histrionic and dependent) dropped entirely. Those that remained would be defined on the basis of trait domains to be rated by clinicians (Skodol et al., 2011), rather than on *DSM*-style algorithms (a list of characteristic features to be scored categorically).

Thus, the hybrid proposal attempted to satisfy both trait psychologists and the BPD researchers who wrote the article that appeared in *The American Journal of Psychiatry* (Shedler et al., 2010). In the end, as is the case for many compromises, the proposal satisfied no one. The word also went around that the work group was dysfunctional. John Livesley, a strong proponent of dimensional classification, resigned and published an article explaining why the hybrid system was incoherent and unscientific (Livesley, 2010).

The categorical—dimensional dispute was not the only problem for the work group's proposal. The *DSM*–5 field trials, by design, avoided making direct comparisons between *DSM*–*IV* and *DSM*–5. Thus, there was no clear evidence of superiority for the new system. Furthermore, the only field trial that examined the reliability of diagnoses in *DSM*–5 came up with equivocal results (Regier et al., 2013). Results at two sites showed that BPD was reliable in Toronto but not in Houston, a finding that was hardly reassuring.

Opinion was sharply divided on the merits of the hybrid system. Some felt that categories were getting short shrift (Gunderson, 2013). For others, like John Livesley, the system was not dimensional enough and should have dispensed with categories entirely. For still others, particularly members of the work group, it was the ideal solution to a complex problem, and its failure was a tragedy. For myself, the hybrid system was overly complex, lacking in both clinical utility and empirical support. The jury is still out on this proposal, which is now in Section III of *DSM*–5 as an "alternative model" considered to require further research.

Because many experts supported the alternative model, let us examine it in a little more detail. It has a series of stages and procedures (American Psychiatric Association, 2013): (a) determining whether impairment in self (identity or self-direction) and interpersonal (empathy or intimacy) functioning is present; (b) rating the level of impairment on a Levels of Personality Functioning Scale; (c) determining whether one of the six defined types is present and recording the type and the severity of impairment; (d) if none of these are applicable, to record PD traits—specified (PDTS), list the trait domains that are applicable, and record the severity of impairment; and (e) if a more detailed personality profile is desired, evaluate the trait facets. At each of these steps, clinicians would be asked to use a Likert scale (1–5) for scoring. Thus, they could diagnose a PD, rate its severity, and then score levels of personality functioning.

All patients with PD would be rated on five personality trait domains: negative affectivity, detachment, antagonism, disinhibition versus compulsivity, and psychoticism. The first four resemble four of those described in the FFM, as well as by other personality schema. (The fifth factor in the

FFM, Openness to Experience, was omitted because it has little significance for psychopathology.) The fifth domain in *DSM*–5, psychoticism, does not appear in most trait models because it describes problems that are less common in community populations, but it is needed to describe patients with PD.

This was an ambitious undertaking that would probably require more time than most practitioners currently devote to diagnostic assessment. Moreover, each of these ratings required judgment calls to determine what is normal, what is extreme, and what is truly dysfunctional. There was also some unfamiliar terminology (e.g., integrity of self-concept, identity integration, self-directedness, complex and integrated representations of others). I have to wonder whether busy clinicians, who have been ignoring the precise instructions of the *DSM* system for the past three decades, could carry out such a demanding procedure.

My own opinion was that although many of the PD categories listed in *DSM–IV* suffer from serious problems in validity, the hybrid system could have made a bad situation even worse. I had several concerns. First, the system requires multiple ratings that even experts would have difficulty carrying out in a reliable way. Second, it uses obscure terminology (after 40 years in psychiatry, I am still hard put to say what the term *self* means). Third, overly complex concepts that appeal more to researchers might have made it harder to convince skeptical clinicians that patients suffer from PDs. The more difficult it is to make a diagnosis, the more likely it is that PDs will be ignored. Finally, because the hybrid system is only beginning to be examined in systematic research, it would be premature to adopt it unless it had the kind of empirical support that underpins the FFM.

The most serious problem for the hybrid system concerns clinical utility (Paris, 2013a). Although the proposal has received good ratings by selected groups of clinicians (Morey, Krueger, & Skodol, 2013), reliable multiple scoring on multiple dimensions would probably require intensive training. We already know that practitioners do not follow *DSM* in any systematic way. Practitioners neither remember nor consistently apply criteria listed in the *DSM* manual but make diagnoses that correspond to prototypes in their mind rather than to formal algorithms (Zimmerman & Mattia, 1999).

Another serious problem, one that has escaped many observers, is that clinician ratings of psychosocial dysfunction are particularly likely to be unreliable. These ratings would have been necessary to distinguish personality and PD. When you use self-report data that has been subject to psychometric investigation for decades or use structured interviews with established reliability, you can empirically determine cutoff points. Clinicians cannot be expected to get these issues right, however, and may either underestimate or overestimate the likelihood of a PD. Basing a diagnostic classification on clinical impressions could have been problematic.

Had the scientific case for the new system been stronger, it would have motivated all of us to do diagnosis differently, as happened when the third edition of the *DSM* (*DSM–III*; American Psychiatric Association, 1980) was published in 1980. Clinicians could be willing to learn complicated procedures, but they first have to be convinced that the evidence supporting them is unimpeachable. I have taught the *DSM–III* and *DSM–IV* for several decades and can attest to a lack of success in getting trainees to make reliable clinical ratings using either algorithms or quantitative scales.

In my view, the hybrid system was a noble experiment, but the evidence for it was insufficient. I will be happy to change my mind if more data come in to support this proposal. A scientific committee appointed by the American Psychiatric Association also concluded that the proposal had not been sufficiently validated. In the end, radical changes in classification require strong evidence. Thus, in December 2012, it was decided to repeat the *DSM–IV* criteria for PDs, word for word, in *DSM–5*. Although the alternative system was consigned to Section III of the published manual, several members of the work group are promoting the rejected model. It remains to be seen whether further research will interest practitioners in using this system. Furthermore, because preparation of the sixth edition of the *DSM* could take another 15 years, the ultimate fate of this proposal remains uncertain.

The only substantive change for PD classification in *DSM*–5 was the demise of the five-axis system introduced in *DSM*–III; one can no longer speak of disorders as being "on Axis II." In *DSM*–5, all diagnoses are made on a single axis. This is an important and positive development. Putting PD diagnoses on a separate axis never succeeded in giving them more

attention and, if anything, provided another reason to ignore them. (How many times have you seen a diagnosis of "Axis II, deferred"?)

Finally, a diagnostic system that is too complex could have hurt patients with PDs by discouraging busy clinicians from recognizing them—on top of an already strong reluctance. This is why some researchers viewed the rejection of the hybrid proposal with palpable relief (Black, 2013b). As Zimmerman (2012) pointed out, even if the current system is bad, one should not make radical changes without strong evidence. My greatest concern was that the people who would have most been hurt by this system would have been the patients who most need our help.

### THE FUTURE OF PD DIAGNOSIS

The current categorical system for PDs is nothing to be proud of, but *DSM*–5 has retained it, so this is the way that PDs will continue to be diagnosed for some time to come. The "paradigm shift" in classification that Kupfer and Regier (2011) had hoped for, in which all categories of mental illness would eventually be replaced by dimensional scoring, is a dead letter for now. Finally, the belief that quantitative measures drawn from factor analysis of self-report data might correspond to biomarkers that could be identified by neurobiological research remains just that—a belief.

However, a dimensional model has also been proposed for the World Health Organization (WHO), which will be publishing the 11th edition of the *International Classification of Diseases (ICD–11)* by 2017. This system (Tyrer, Crawford, Mulder, & the *ICD–11* Working Group for the Revision of Classification of Personality Disorders, 2011) has one great advantage: It is rather simple. Clinicians would be asked to rate patients on a scale of 5 points: no PD, personality difficulty, PD, complex PD, and severe PD. There would be no specific categories, although clinicians would also be asked to score patients on five trait domains: asocial, dissocial (i.e., antisocial), anankastic (i.e., obsessional), anxious—dependent, and emotionally unstable.

Assuming that WHO accepts this system, we could have two competing and incompatible systems for PD classification. I have the same concern

about *ICD-11* as about the hybrid system: Can clinicians be trained to make reliable and valid ratings of trait profiles? Even with a simpler system, they would need a lot of training, which would in turn be expensive.

A much more radical (but also more complex) proposal to eliminate categorical diagnosis in psychiatry entirely has been promoted by the National Institute of Mental Health (NIMH). Thomas Insel, the director of the NIMH, has proposed a system to replace *DSM* called Research Domain Criteria (RDoC; Insel et al., 2010). This model is already being applied to the assessment of research grants. RDoC is a matrix of theoretical dimensions of psychopathology across many levels of analysis. It is based on the idea that mental illnesses fall along a spectrum and that quantitative measures will shed light on the endophenotypes underlying mental disorders.

The RDoC system attempts to eliminate all categories of mental illness in favor of scoring procedures believed to correspond more closely to neurobiology. Although some consider it the wave of the future, at our present state of knowledge, it is overambitious. Mental health practitioners will continue to use *DSM* diagnoses because they are familiar ways to allow communication; practitioners would have to be convinced that alternatives are scientifically or clinically superior.

RDoC would not just dimensionalize PD but make it disappear completely, in favor of ratings of variations in cognition, emotion, and behavior. In my opinion, the RDoC matrix is, at this point, full of holes where solid research should be. Furthermore, this ideological proposal comes from an NIMH director who is on record as wanting to abolish psychiatry by combining it with neurology (Insel & Quirion, 2005). RDoC are a manifesto for neurobiology that also takes little or no account of psychology, effectively dismissing any research not based on, or linked to neurochemistry, neurophysiology, or brain imaging. The idea that mental disorders are brain disorders goes along with the mindless reductionism of contemporary psychiatry (Paris, 2008b).

The construct of a PD will survive this onslaught. Clinical psychologists, social workers, and psychiatrists work with real people and do not spend their time in laboratories. They are not ready to reject the concept of mind or view a person as a neural network. For better or for worse,

our current PD categories are associated with a personological approach, consistent with the practice of psychotherapy.

### A LOST OPPORTUNITY

The failure to revise procedures for PD diagnosis in *DSM*–5 was in many ways a lost opportunity. We still need a better general definition of PD, even if the hybrid system produced one that was too complicated. Another loss was the need to repair flaws in the criteria for specific PDs. After 20 years of additional research, the work group might have come up with a better set of algorithms than those found in *DSM*–*IV*. The current criteria continue to be sorely deficient in discriminant function, which is why patients who meet criteria for one PD will often meet criteria for others. Moreover, some of the categories in *DSM*–5 are next to useless because they describe a single trait.

The most serious loss concerned patients who fit the general description of a PD but do not meet criteria for any of the 10 categories listed. They still have to be diagnosed as "personality disorder, unspecified," with about half of those who meet overall PD criteria falling into this group (Zimmerman, Rothschild, & Chelminski, 2005). The fact that the most frequent diagnosis in practice cannot be specified is a serious indictment of the current system. A better description of trait profiles in patients with this diagnosis would have allowed a vague category to be replaced with a more precise assessment of the traits that characterize individual patients, but I cannot recommend going to Section III of the manual to do so.

A 2013 issue of the journal *Personality Disorders: Theory, Research, and Treatment* presented a series of target papers and commentaries that reviewed the story of *DSM*–5's approach to PDs. The most sensible, in my view, was a paper by Tom Widiger (a prominent player in the *DSM*–*IV* process who was not consulted about *DSM*–5). Widiger (2013) pointed out that it was a mistake to create a new and untested system when the FFM was already available and could have been applied for clinical use. Widiger was in favor of retaining some categories and noted that no one had ever taken the trouble to check out the discriminant validity of their diagnostic criteria, almost all of which could have been improved.

One does not need to be a psychometrician to understand that a system with problems like those of the *DSM* badly needed revision. One might conclude that being stuck with the same system as in previous editions is dispiriting. Yet despite all the research in recent decades, none of the proposed alternatives was sufficiently better than the status quo to gain wide support. It should be kept in mind, however, that the situation for PD diagnosis is not much worse than any of the other groupings in *DSM*–5. Until we know much more about mental disorders, any attempt to develop a scientific classification is premature and bound to fail.

I support Widiger's views for two reasons. First, they would allow a degree of continuity for research communities studying well-investigated categories such as BPD and ASPD. Second, adapting the FFM for clinical use, most probably using a brief self-report instrument, would be superior to developing a new and untested system.

In summary, PDs as a whole do not quite fit with either classical trait psychology or a categorical medical model. The current categories of PD are problematic and can only be regarded as provisional. Eventually, when we know a good deal more about their endophenotypes, they will be replaced. For now, there may be value in retaining some categories that are familiar to clinicians and that have a strong basis in research. However, given the passion that has marked the categorical versus dimensional controversy, I cannot be optimistic that these principles will be adopted any time soon.

# Etiology

In this chapter, I develop a general theory on the etiology of PDs. I propose that only interactions among biological temperament, psychological adversities, and social factors are sufficient conditions for the development of this form of psychopathology.

### **TEMPERAMENT**

Every child has a unique temperament. Heritability accounts for about half the variance in traits that eventually emerge and also determines the type of PD that can develop (Paris, 1998). Although an abnormal temperament need not by itself lead to a PD, it can be a risk factor for psychopathology.

Almost 30 years ago, the great British child psychiatrist Michael Rutter (1987) wrote a classic and seminal paper describing the hierarchical relationship among temperament, traits, and PDs. In this model, personality

http://dx.doi.org/10.1037/14642-004

A Concise Guide to Personality Disorders, by J. Paris
Copyright © 2015 by the American Psychological Association. All rights reserved.

is rooted in temperament (heritable biological factors producing individual differences in emotion, cognition, and behavior). Personality traits (stable characteristics of emotion, cognition, and behavior) are amalgams of temperament and life experience, and disorders are pathological exaggerations of traits.

Temperament, although present at birth, is difficult to measure directly. Researchers have studied infant behavior, measuring differences on dimensions of positive affect, fear, frustration or anger, and effortful control (Rothbart, 2007). However, there is a paucity of research in which children with problematic temperaments have been followed into adulthood to determine the risk for PD or other mental disorders. Kagan (2012) followed a cohort of infants with behavioral inhibition, a likely precursor of anxious PDs, but only into adolescence. Published long-term follow-ups have usually started in middle childhood, although some have followed their subjects for decades (Caspi & Roberts, 1999; Cohen, Crawford, Johnson, & Kasen, 2005; Tremblay, 2006). These studies did not measure early temperament, however. Moreover, birth cohort studies that have followed subjects from "the cradle to the grave" do not have a high enough rate of severe psychopathology in adulthood to produce that kind of data.

Another problem is the absence of biological markers associated with temperamental variations (Rettew & McKee, 2005). Given the strong evidence for heritability of personality traits (Plomin, DeFries, Knopik, & Neiderhiser, 2012), one might expect to find such relationships, but researchers will need many decades to correlate the enormous complexity of brain structure and function with temperament. It is unlikely that advances in brain chemistry or connectivity will be able to address this problem in the foreseeable future.

### PERSONALITY TRAITS

Personality traits are an amalgam of temperament and life experience, and thus they reflect the effects of interactions between inborn tendencies and the psychosocial environment (M. Rutter, 1987). Most have a heritability close to 0.5, are reasonably stable by late childhood, and

#### ETIOLOGY

remain fairly stable over the life course (Widiger & Costa, 2013). This is not to say that personality cannot change but that traits set limits on the extent of change.

The stability of traits is an important observation for the practice of psychotherapy. We cannot change personality, but we can help patients to function better using traits more effectively (Paris, 1998). For example, although introverts may always remain shy, there are ways, particularly in an age of computers, to use that trait adaptively. Similarly, neuroticism can be adaptive if associated with emotional responsiveness rather than with pathology, particularly when people learn how to control their emotions (Gross, 2013).

The widely believed principle that the quality of parenting shapes personality traits is partly right and partly wrong. There are statistical relationships between poor parenting and outcome, as well as clear-cut negative effects from abuse and neglect, but these risks do not predictably or necessarily produce major psychopathology in adulthood (Fergusson & Mullen, 1999; M. Rutter & Rutter, 1993).

By and large, it is easier to ruin a child than to raise one. Many books have been written about the right and wrong way to be a parent. Yet most children develop in their own way, based on their temperamental characteristics. Moreover, the impact of the psychosocial environment is itself genetically mediated (Belsky & Pluess, 2013)—that is, people with different temperaments respond to the same environmental events in unique ways. For example, those who are environmentally sensitive (i.e., highly neurotic) are influenced more by both positive and negative life events, benefiting from a good environment more than those who are relatively insensitive (Belsky & Pluess, 2009). Moreover, positive temperamental characteristics lead to gene—environment interactions that help explain high levels of resilience to traumatic events during childhood (M. Rutter, 2012).

A great deal of effort has gone into the precise measurement of personality traits. Researchers almost always use self-report questionnaires with established reliability and external validity and in which trait domains are identified by factor analysis. The most widely used instruments are based on the five-factor model (FFM), a theory of personality that describes five

broad domains (see Chapters 1 and 2, this volume). The main limitation of the FFM is that it depends on self-report. People are not always accurate when reporting negative traits so questionnaires should ideally be supplemented by direct observations, peer ratings, and (eventually) biomarkers (Kagan, 2012). Nonetheless, given the problems of clinical ratings, self-report remains the most empirically validated option.

Another limitation is that the traits identified in community samples do not account for all features of severe psychopathology. As discussed in Chapter 2, there are several alternatives to the FFM that have specifically set out to measure behavioral patterns seen in clinical settings.

Yet however traits are measured, heritability always accounts for nearly half the variance (Plomin, DeFries, Knopik, & Neiderhiser, 2012). The method most often used to quantify heritability is *behavioral genetics*, in which monozygotic (MZ) and dizygotic (DZ) twins can be compared for concordance in diagnosis or in trait dimensions. A trait can be determined to be heritable to the extent that concordance is higher in MZ than in DZ twins. Other methods, such as adoption studies, or studies of twins separated at birth, are not often practical but yield similar results.

Behavioral genetic methods also yield an estimate of environmental influence, which accounts for the other half of the variance (Plomin et al., 2012). Children who grow up in the same family are no more similar than perfect strangers (Dunn & Plomin, 1990). Thus, the environmental variance in personality is not shared (i.e., the result of being brought up in a particular family), as past theories would have predicted. Instead, it is almost entirely unshared (i.e., affected by environmental factors outside the family or by unique experiences within a family). This finding, which overturns many long-held assumptions of developmental psychology, has been one of the biggest surprises in psychological research over the past 50 years.

It would be useful to find biomarkers associated with personality traits—either the broad dimensions of the FFM or its more narrowly defined facets—but no such relationships have been found. There is some evidence that abnormalities in serotonin activity correlate with a tendency to be impulsive (Siever & Davis, 1991), yet even that relationship is not consistent (Carver & Miller, 2006). Although establishing biomarkers has

#### ETIOLOGY

to be a goal for further research, in the current state of knowledge, neuro-scientists don't know what to look for. Trait dimensions as currently defined are consistent, but they may or may not be biological constructs.

# BIOLOGICAL FACTORS IN PDs

We know that PDs are at least partially heritable. What we do not know is how genes shape the brains of people who develop such disorders. The heritability of PDs has been demonstrated by a series of studies, almost all of which were conducted in Scandinavia (Kendler et al., 2008; Reichborn-Kjennerud et al., 2013; Torgersen, Kringlen, & Cramer, 2001). The level of heritability is in the same range as traits, about half the variance for most disorders. Borderline PD (BPD), long thought to be an environmental condition, has a single heritable factor that accounts for 55% of the variance in all nine *Diagnostic and Statistical Manual of Mental Disorders* (5th ed.; *DSM*–5; American Psychiatric Association, 2013) criteria (Reichborn-Kjennerud et al., 2013). These findings overturn a good deal of conventional wisdom. Yet despite intensive research, no biomarkers for any type of PD have been discovered. Researchers should not feel apologetic, however; there are no biomarkers for any major mental disorder (Hyman, 2010).

What heritability determines is not PD itself but who is and who is not vulnerable to PD. A vulnerable temperament makes the development of a PD more likely, but this outcome can only be explained by complex interactions between traits and environmental stressors (Paris, 1998).

Temperament also determines the type of PD that can develop in any person. A highly introverted person will not develop narcissistic PD, and a highly extraverted person will not develop avoidant PD. In this way, genes "bend the twig" but do not determine the shape of the tree. Also, the specific symptoms associated with severe PDs, such as BPD, may not have the same heritable component as the traits that underlie the disorder. This additional complexity is why some PDs, contrary to theory, are only partially continuous with the domains of personality.

Finally, the idea that heritability means that traits cannot change is mistaken (M. Rutter, 2006). In fact, some of the most important effects

of heritable factors in psychopathology lie in individual differences in susceptibility to environmental risks (Rutter, 2006). Some people are highly resilient and are surprisingly unaffected by serious adversities in life. Others are highly sensitive and are affected by all kinds of life events, good or bad (Belsky & Pluess, 2013).

In summary, biological factors in PDs can only be understood in terms of gene–environment interactions. Nonetheless, it would be useful if we could identify biomarkers associated with these conditions. This would help to make diagnosis precise. It might also open up the possibility of defining high-risk populations are most likely to develop PDs. Finally, biomarkers would be useful in research.

Potential biomarkers for PDs could include abnormalities in genetic variations, neurotransmission, neuropsychological measures, or neuroimaging (volume, activity, and connectivity of specific brain areas). There is a fairly large body of research examining each of these strategies in major mental disorders. Almost all of the research on PDs has focused on the borderline category.

Although this literature is large, the state of current knowledge, as described in a number of review articles (Mauchnik & Schmahl, 2010; New, Goodman, Triebwasser, & Siever, 2008; Ruocco, 2005), can be summarized as follows:

- No specific allele has been found that can account for any PD, and those that have been studied in BPD account for no more than 1% of the variance (e.g., Ni, Chan, Chan, McMain, & Kennedy, 2009). It is likely that heritability is associated with the activity of a large number of interacting genes.
- 2. There is no evidence for chemical imbalances in PDs. There is some evidence that abnormal serotonin activity is associated with impulsivity (New et al., 2008), but this finding is neither fully consistent nor specific to any diagnosis or trait profile.
- 3. Neuropsychological testing shows that impulsive PDs are associated with a failure of higher centers (prefrontal cortex) to inhibit impulses and emotions coming from the limbic system and the amygdala (Ruocco, 2005).

#### ETIOLOGY

4. Neuroimaging studies in patients with BPD show decreased activity in prefrontal cortex (Mauchnik & Schmahl, 2010), reduced volume in the amygdala and hippocampus (Ruocco, Amirthavasagam, & Zakzanis, 2012), an overactivity of "alarm" circuits in the amygdala (Donegan et al., 2003), associated with a failure of cortical inhibition (Koenigsberg et al., 2014). These findings are interesting but do not go much beyond what we already know through clinical observation.

At this point biological findings in PD can be understood as first steps in a long journey toward understanding how the mental abnormalities in PD are reflected in the brain. Moreover, none of these results is strong enough to justify any specific biological treatment.

## PSYCHOLOGICAL FACTORS

We have a lot to learn about the role of the psychological environment in PD, and what we thought we knew is in doubt. It was long believed that PD is the direct result of an unhappy childhood. These ideas have been associated with psychoanalysis. It was also assumed that the more severe the disorder, the more likely it had its origins in the earliest stages of development—a theory so ubiquitous that prominent PD researchers have taken it for granted (Millon & Davis, 2011). These ideas have also had a vast influence on psychotherapy, encouraging clinicians to search for past traumas to account for present distress.

Yet the idea that the cause of PD lies in childhood is oversimplistic. It has a grain of truth in that early adversity is, statistically speaking, a risk factor for psychopathology (Rutter & Rutter, 1993). Yet most people with unhappy childhoods develop no mental disorder at all (Rutter, 2012). Resilience is not the exception, but the rule. Moreover, only some patients with mental disorders report psychological trauma as children (Paris, 2000). Although some experiences are particularly traumatic, even people with a history of childhood sexual abuse (CSA) do not necessarily suffer major sequelae (Fergusson & Mullen, 1999). PD is the outcome of complex interactions that we are only beginning to understand.

It follows that practitioners who routinely attribute adult psychopathology to childhood experiences are making a cognitive error. They are

confusing correlation with causation. For every patient who comes in with symptoms associated with a history of adversity, there are thousands of others who are functioning reasonably well and will never seek clinical attention. There are also people who have such an abnormal temperament that they are at risk for PD without having experienced severe childhood adversity. Nonetheless, because people who do develop severe PDs are more likely to have had adverse childhood experiences, particularly in the borderline (Paris, 2008a) and antisocial categories (Robins, 1966), clinicians still need to explore these histories carefully.

These relationships are best framed in terms of gene—environment interactions. Doing so can be difficult. We prefer simple explanations to complex ones. From childhood on, our minds see cause and effect in everything, even when there is no relationship (Bloom, 2013). We have to be trained to think multivariately, even if it is a strain to keep that perspective in mind.

Those patients with PD who have had a reasonably normal child-hood may have had special needs, based on abnormal temperament, that are hard for any family to meet. In adulthood, they may therefore perceive their upbringing as inadequate. As Dobbs (2009) suggested, some people are "orchids" (successful only in just the right environment), and others are "dandelions" (able to cope with almost any environment). This point of view is also consistent with the research of Belsky and Pluess (2013), which shows strong variations in environmental sensitivity, such that people who are sensitive to adversity also benefit more than most others from a positive environment. This is a hopeful message for clinicians who treat PD.

# CHILDHOOD ADVERSITY IN BPD

Strong evidence for environmental risk factors in PDs has come from studies of BPD. CSA is common in this disorder and, when present, makes its course more severe (Soloff, Lynch, & Kelly, 2002). Although about two thirds of patients with BPD report some form of CSA, these large numbers are misleading. This is because they include events that do not lead

#### ETIOLOGY

to sequelae in normal populations (Paris, 2008a). Thus, single incidents with strangers or nonrelatives, not involving physical contact, are not risk factors for any mental disorder. However, severe abuse, perpetrated by family members or caretakers, can lead to psychopathology (Fergusson & Mullen, 1999). The parameters of CSA (perpetrator, nature of the act, duration) are more important than its simple presence in a patient's history (Paris, 2008a).

Severe childhood abuse of any kind occurs in only about a third of BPD cases, with abuse from a caretaker in about a quarter of these patients (Zanarini, 2000). This finding points to risk, but clinicians should not assume that every patient with BPD must have been abused as a child. The disorder is a final common pathway, demonstrating what Cicchetti and Rogosch (1996) called "equifinality" (the same outcome arising from different causes). Different patients also receive different doses of adversity that have different consequences (Paris, 2008a). Some BPD patients have histories of severe child abuse that go a long way to explaining the severity of their symptoms. Others have only mild or intermittent abuse that should be seen as part of a larger picture of family dysfunction. Finally, about a third have never experienced abuse at all.

The most important parameters of CSA are the relationship to the perpetrator and the nature of the sexual contact. The most pathogenic form of CSA is parental incest (Finkelhor, Ormrod, Turner, & Hamby, 2005), but it is not the most common form of CSA reported by BPD patients. It more frequent to hear that stepfathers, boyfriends of the mother, or older brothers were perpetrators (Paris, 2008a).

The term *abuse* can sometimes be abused. It is used to describe all sorts of things that can go wrong in childhood. Physical abuse is clearer and is another common risk factor for BPD, although not as specific to the disorder as CSA. Researchers have also studied emotional abuse, that is, constant and hurtful criticism from caretakers, which is frequently reported (Zanarini, 2000).

These adversities are more common in dysfunctional families. In most ways, a chaotic and dysfunctional family is a more essential risk factor, and a more important focus for clinical attention, than any specific type of

adversity. Parents of patients with BPD can have serious psychopathology, including substance abuse, depression, and PDs (White, Gunderson, Zanarini, & Hudson, 2003). This does not make them ideal caretakers for vulnerable children.

When BPD develops in the absence of trauma, the failure of parents to understand children with special emotional needs may be important. Linehan (1993) conceptualized BPD as an interaction between a vulnerable temperament (emotion dysregulation) and the failure of parents to understand the problem (an invalidating environment). A good body of research has supported this seminal idea, and later chapters in this book examine its clinical implications.

Parker (1983) developed a self-report instrument, the Parental Bonding Index (PBI), based on a more general theory about what makes for successful parenting. The essential idea is that children need emotional warmth and empathy, as well as a respect for their autonomy. These dimensions tend to be orthogonal and can be measured (for each parent) by subscales of the PBI. Parker's most consistent finding was that depression is associated with "affectionless control" (low affection combined with overprotection). Our own research group found similar results in BPD using the PBI to study recollections of parenting, compared with patients without a PD (Zweig-Frank & Paris, 1991) and to patients who had other types of PD (Paris, Zweig-Frank, & Guzder, 1994a). The results also suggested that BPD patients can suffer from *biparental failure*, that is, the absence of validating responses from either parent.

Again, interactions between temperament and adversity are the key to understanding these pathways. A good example came from a research study (Laporte, Paris, Guttman, & Russell, 2011) designed by my colleague Lise Laporte. We studied 56 pairs of sisters in which one had BPD and observed that concordance for the disorder only occurred in three of these pairs. This finding was particularly striking because we had a sample of severely ill patients, and both sisters had been exposed to the same traumatic events (Laporte, Paris, Russell, Guttman, & Correa, 2012). However, when we gave sisters a measure of personality traits developed for research on clinical populations (Livesley, Jang, & Vernon, 1998), differences were

#### ETIOLOGY

dramatic: BPD probands had abnormal scores on every subscale, whereas the normal sisters did not. Our interpretation was that trauma and adversity are most likely to be associated with BPD in children with an abnormal temperament.

The research on antisocial PD (ASPD) is less well-developed but shows a number of similarities to findings in BPD. It has been known for decades that dysfunctional families, particularly those in which the father is also antisocial, are also strong risk factors for the disorder (Robins, 1966). The risks are essentially the same as for conduct disorder, an early form of antisociality in young children and a required antecedent for making the diagnosis in adults. Most likely, ASPD is the result of an adverse environment in children with an abnormal temperament.

Research about environmental risk factors in other PDs is too thin to reach any firm conclusions, but a similar model might be applied. Interactions between temperament and adversity is a general principle for understanding the origins of psychopathology, and it will guide much of the rest of this book.

# SOCIAL FACTORS IN PDs

One of the main mechanisms behind resilience to adversity is access to social support and to attachments that lie beyond the family (Rutter, 2012). Moreover, because PD involves a disjunction between personality and social demands, the risk for developing these disorders depends on social context. Traits that are adaptive in one society can be maladaptive in another. When social demands change over time, the risk for developing a PD can also change.

A good example is ASPD, which has a high prevalence in developed countries—about 3% in North America (Robins & Regier, 1991). In contrast, there is a low prevalence in traditional societies, such as Taiwan (Hwu, Yeh, & Chang, 1989). ASPD increased in prevalence in the United States and Europe after the Second World War, and these cohort effects may have been associated with the social and family breakdown that marked those decades (Rutter & Smith, 1995). Thus social risk factors operate

through two mechanisms: They further aggravate psychological risk, and they interfere with resilience factors that could be based on a supportive social environment.

It has been suggested that BPD has also increased in prevalence since the Second World War (Millon, 1993). It is certainly surprising that this important disorder was never described before 1937. I have long wondered why people in the past, who could also have had a difficult temperament and who could also have been exposed to psychosocial adversity, did not develop BPD. Yet although most severe mental disorders can be recognized throughout history, one does not read about people cutting themselves and taking recurrent overdoses until about 75 years ago. The explanation may be that patients with the same underlying traits presented in different ways in other historical periods (Paris & Lis, 2013).

Narcissistic PD is another disorder that may be socially sensitive. It has been suggested that trait narcissism has increased over the past several decades, largely due to a more indulgent style of parenting and a society that values self-esteem about achievement (Twenge, 2011). This intriguing hypothesis requires more research.

Finally, it is possible that PDs in general are on the increase because of the rapid social change associated with modernity over the globe (Paris, 2014a). The type of PD one develops would be determined by temperament, but whether one's traits become amplified to the level of disorder would be influenced by the social environment.

# DEVELOPING AN INTEGRATED MODEL

Let us now consider an integrated etiological model of PD that takes into account all these risk factors: biological, psychological, and social. The biological risks involve a difficult temperament, which could be impulsive and extraverted or anxious and introverted. Although children could also have other combinations of temperamental variables, these are the two basic dimensions of psychopathology in children: *externalizing* and *internalizing* (Achenbach & Ndetei, 2012). A distinction between symptoms associated with impulsive actions versus symptoms associated with inner suffering is supported by a large body of research (Achenbach & Ndetei, 2012).

#### ETIOLOGY

Temperamental risks also lead to higher environmental risks. For example, children with an externalizing temperament are more likely to come into conflict with parents, peers, and teachers (Rutter & Rutter, 1993). In a family where other children have a different trait profile, temperamentally vulnerable children may be scapegoated. On the other hand, good parenting can help children with a difficult temperament overcome many difficulties. What is needed is high structure, predictable expectations, and empathy (Gordon, 2000). In contrast, a pattern of inconsistent, disruptive parenting with insufficient monitoring is characteristic of parents with conduct-disordered children (Berg-Nielsen, Vikan, & Dahl, 2002).

Similarly, parental behaviors have a relationship to the development of internalizing symptoms. Affectionless control is a risk factor for depression and anxiety in children (Berg-Nielsen et al., 2002), and there is evidence that this style of parenting is related to all disorders in which anxiety is prominent (Parker, 1983). Conversely, sensitive parenting can help children overcome internalizing temperamental patterns and become healthily introverted adults.

The social environment plays a role in buffering the influence of temperament and parenting on personality. Living in a good neighborhood, going to a good school, and having access to community activities are all important factors in the development of a healthy personality. Conversely, bad neighborhoods, bad schools, and living in a socially disintegrated community will be risk factors for psychopathology (Rutter & Rutter, 1993). These risks will have more negative effects on those who are temperamentally vulnerable as well as on those who come from dysfunctional families. Social factors may be of particular importance in adolescence, when problematic traits can interfere with the developmental task of identity formation. We live in a society where each person has to find his or her own role and niche in society, and this may be more difficult for those who are temperamentally impulsive or unusually anxious (Paris, 2014a).

In summary, PDs are most likely to develop when all of these factors, interacting with each other, are present. No single risk factor is sufficient to produce a PD.



# Prevalence, Precursors, and Outcome

This chapter reviews research on the prevalence of personality disorders (PDs) in the community, childhood precursors of PDs, and clinically important research on PD outcome.

# **EPIDEMIOLOGICAL STUDIES**

Epidemiological research on the community prevalence of PDs was rare until fairly recently. Researchers were unsure whether these complex conditions could be reliably identified in large-scale surveys.

The first extensive project in the United States to examine the prevalence of the major mental disorders listed in the third edition of the *Diagnostic and Statistical Manual of Mental Disorders (DSM–III)* was the Epidemiological Catchment Area (ECA) study (Robins & Regier, 1991). However, the ECA only examined antisocial PD (ASPD), considered to be

http://dx.doi.org/10.1037/14642-005 A Concise Guide to Personality Disorders, by J. Paris Copyright © 2015 by the American Psychological Association. All rights reserved.

the one category that was defined well enough to be assessed in a community sample.

Over the past two decades, the situation has changed greatly. A major study conducted in Norway (Torgersen et al., 2001) examined all the PDs in the manual. The findings showed that PDs, as defined by *DSM–III*, are common, with at least 10% of the community population meeting criteria for one category or another. A second study, conducted in the United Kingdom and based on criteria from the fourth edition of the *DSM (DSM–IV;* Coid, Yang, Tyrer, Roberts, & Ullrich, 2006), came up with similar results. A third study, also based on *DSM–IV* and drawing on data from the large-scale National Comorbidity Study (NCS) in the United States (Lenzenweger, Lane, Loranger, & Kessler, 2007), came up with similar levels of prevalence.

Thus, a consensus has emerged that about one in 10 people meet criteria for a PD. In these studies, ASPD was found to be particularly frequent, with a prevalence of approximately 2% to 3%. Borderline PD (BPD) was less frequent, with rates ranging between something less than 1% and 2% (Paris, 2010a). However, all these numbers depend on the definitions in the *DSM*—if a higher bar had been set, prevalence could have easily have been halved.

Epidemiological research is usually more likely to overestimate prevalence than to underestimate it. In a survey of alcohol and drug abuse in the United States, the National Epidemiologic Survey on Alcohol and Related Conditions (NESARC; Grant et al., 2004) results differed dramatically from previous studies. NESARC had one big advantage: access to a particularly large sample. However, its estimates of prevalence seem to have been inflated by the use of too low a bar for diagnosis. The study found that as many as 15% to 20% of the population could meet criteria for one or more PD diagnoses. A reanalysis of the same data (Trull, Jahng, Tomko, Wood, & Sher, 2010), raising the bar for formal diagnosis, found that although the prevalence of PDs was still high, rates were much closer to what the earlier studies had found. Unfortunately, the higher numbers have been quoted ever since. This may reflect bias and self-interest. When researchers write grants or introductory sections of papers submitted for publication,

#### PREVALENCE, PRECURSORS, AND OUTCOME

they usually aim to prove that the problem under study is common in the community.

The lack of a clear boundary between PD and trait variation remains the real problem, and it is not easily solved. In the end, prevalence depends on the judgment of investigators who measure it (Paris, 2010a). For example, Grant et al. (2004) claimed that obsessive—compulsive PD has a community prevalence of 7%. This is an enormous number, dwarfing the prevalence of common disorders such as depression. Yet because obsessive—compulsive PD is an almost pure trait disorder, it is not at all clear that everyone who met criteria in this study had a mental illness. Perhaps all it tells us is that 7% of the population is more perfectionistic than is good for them.

By contrast, PDs are common in clinical settings. Zimmerman, Rothschild, and Chelminski (2005) estimated that up to half of all outpatients meet criteria for one category or another. The irony is that clinicians who had a clinical focus on mood and anxiety missed many PD diagnoses. In particular, patients with BPD were often diagnosed with bipolar disorder because they have mood swings, even when they have never had episodes of mania or hypomania (Paris, Gunderson, & Weinberg, 2007).

The clinic is where we should be sharpening our diagnostic skills, not in community populations where most people manage without any treatment. The failure to recognize PDs in these settings is one of the great tragedies of our time. Patients are being managed with interventions designed for anxiety or depression, and their personality pathology may not be addressed at all.

Another problem is that a continuous relationship between traits and disorders makes it difficult to determine cutoff points. This may explain why many estimates of PD prevalence in the general population have ranged as high as 10% (Paris, 2010a). In the largest study, obsessive—compulsive PD had the highest prevalence, even though this diagnosis fades imperceptibly into normality. In contrast, the prevalence of BPD, which has more symptoms that are distinct from normality, was much lower, similar to that found for major mental disorders such as schizophrenia (Paris, 2010a).

# CHILDHOOD PRECURSORS OF PDs

A large body of recent research has examined the precursors of PD in childhood (De Fruyt & De Clercq, 2014; Scott, Stepp, & Pilkonis, 2014; Tackett, Herzhoff, Reardon, De Clercq, & Sharp, 2014; Tackett & Sharp, 2014). If PDs are rooted in temperamental variation, one would expect them to be apparent in childhood. This sequence has been well documented in ASPD. Follow-up research (Robins, 1966) indicated that early-onset conduct disorder almost always precedes ASPD. In fact, it is the same disorder but at different stages of development. The main caveat is that not every child with conduct disorder ends up developing ASPD (Zoccolillo, Pickles, Quinton, & Rutter, 1992).

It has been accepted in all the *DSM* manuals since the third edition that one cannot even make an ASPD diagnosis without a childhood history of conduct disorder. This is one of the few examples in which a research finding has actually changed diagnostic criteria. Moreover, the earlier and more severe conduct symptoms are, the more likely the outcome will be ASPD (Zoccolillo et al., 1992). In a longitudinal community study (Caspi & Roberts, 1999), antisocial behavior at age 18 could be predicted with surprising accuracy by a 90-minute interview of mother and child conducted at age 3.

It would be helpful if we knew as much about the precursors of BPD as we do about ASPD. BPD usually begins in adolescence, a stage at which all its clinical features are already apparent (Chanen & McCutcheon, 2013). The idea that the disorder should not be diagnosed at that stage is plain wrong. It is based on the idea that adolescents grow out of problems like this, but they don't. Even so, the fifth edition of the *DSM* wisely advises clinicians to ensure that PD symptoms have been present for at least a year.

What are BPD patients like before adolescence? Although many patients report an unhappy childhood, they may or may not have had observable symptoms before puberty. If they do, the clinical presentation may be different. Research is just beginning to address this question. The most consistent finding is that children who later develop BPD have a mixture of internalizing and externalizing symptomatology.

#### PREVALENCE, PRECURSORS, AND OUTCOME

This principle was illustrated by the Pittsburgh Girls Study, a longitudinal follow-up of a high-risk group (Stepp, Pilkonis, Hipwell, Loeber, & Stouthamer-Loeber, 2010). The researchers found that externalizing disorders (attention-deficit/hyperactivity disorder, conduct disorder, and oppositional defiant disorder) as well as internalizing disorders (particularly posttraumatic stress disorder) tend to precede BPD in the childhood of women who develop symptoms in adolescence. It should be kept in mind, however, that there are multiple pathways to BPD, so that no single profile predicts the outcome (in the way that conduct disorder predicts ASPD).

An ongoing longitudinal study of prepubertal girls with symptoms resembling BPD (Zanarini et al., 2011) found similar psychosocial risk factors at age 11 to those documented in adults: abuse, neglect, and dysfunctional families. When these cohorts are followed up into adulthood, the study could provide further insight into the precursors of BPD in adolescence and adulthood.

In a study of 12-year-olds from a birth cohort, Belsky et al. (2012) identified a group with BPD features. The findings were that BPD characteristics are heritable but that children with these features also had a history of trauma and family dysfunction. This is further confirmation of the importance of gene–environment interactions.

My research group studied a group of prepubertal children in a day program who had symptoms resembling adult BPD and whom we followed into adolescence (Zelkowitz et al., 2007). We also found risk factors similar to adult BPD (Guzder, Paris, Zelkowitz, & Marchessault, 1996). However, most of our cases were male, and although these men continued to have life problems, they did not develop BPD over time.

The ideal study of childhood precursors of PD would need to follow large cohorts into adulthood. This has only rarely been carried out, largely because of the expense. The Children in the Community Study (Cohen, Crawford, Johnson, & Kasen, 2005) attempted to do so by following a group of children in the Albany–Saratoga area of New York State over the course of 30 years. However, due in part to attrition, they were not able to identify enough cases that met full PD criteria and so could only report predictors of PD symptoms, limiting the generalizability of the findings to subclinical populations.

In summary, there is evidence that BPD has precursors before puberty, but they may or may not come to clinical attention. Over the next 10 years, as longitudinal data come in, we should know more.

Much less is known about the childhood precursors of other PDs. A long-term study (Kagan, 2004) of infants with "behavioral inhibition" (high social anxiety) during infancy could have shed light on the antecedents of avoidant PD, marked by social anxiety and withdrawal. Most of Kagan's (2004) subjects did well in adolescence, but some continued to have symptoms. Unfortunately the follow-up did not last long enough to determine which diagnoses developed in the adult years.

# **OUTCOME**

PDs used to be thought of as chronic, lifelong, and hopeless conditions. This perception has been an important factor in the reluctance of clinicians to diagnose or treat them. However, long-term follow-up research has shown that most patients get better with time. The prognosis is actually better than for many other mental disorders.

Clinicians tend to overestimate chronicity mainly because patients who continue to have symptoms keep coming for help, whereas those who have no longer have symptoms do not. This is what Cohen and Cohen (1984) called the "clinician's illusion." You won't often see recovered patients in your office unless you have arranged for long-term follow-up. Moreover, the unfairly negative image of PD outcome is shaped by the cases that do not do well. Meanwhile, new patients who have not yet recovered, as well as the important minority of patients who fail to recover, continue to flood our clinics.

Moreover, as shown by Zimmerman and Mattia (1999), many patients are misdiagnosed and treated either with medication or standard therapy for their "comorbid" mood disorders. It is only when these conditions fail to respond that a PD diagnosis is even considered.

Twenty-five years ago, a group of retrospective studies following BPD patients into middle age showed a striking pattern of recovery after 15 years. This finding came as a surprise, but perhaps we could have anticipated it.

#### PREVALENCE, PRECURSORS, AND OUTCOME

Many of us had wondered why most of our patients were young, and why older patients were notable in their absence. It was hard to believe they had all died by suicide or from illness.

These studies of 15-year outcome were published in the 1980s. Two were conducted in private hospitals (McGlashan, 1986; Plakun, Burkhardt, & Muller, 1985), one in a state hospital (Stone, 1990), and one in a general hospital (Paris, Brown, & Nowlis, 1987). Despite differences in samples, attrition levels, and methodologies, the results were almost identical. By age 40, most patients were functioning better, and the majority were no longer in treatment.

A few years later, a long-term study in Iowa (Black, Baumgard, & Bell, 1995) reported similar findings for patients with ASPD, some of whom were followed for decades. They showed gradual improvement over time but continued to have difficulties. What BPD and ASPD have in common is impulsivity. It is known that most disorders characterized by this trait (substance abuse, bulimia nervosa) improve over time. In fact, most people are less impulsive at 40 than they were at 20. These changes are probably due to brain maturation, as well as to social learning. Patients with ASPD and BPD will no longer meet criteria if they are less impulsive after 5 or 10 years. This does not mean, however, that their interpersonal problems are behind them.

The findings of these follow-back studies needed to be supplemented by large-scale prospective follow-ups and needed to be generalized to a larger population suffering from a wider range of PDs. The National Institute of Mental Health made a major investment in sponsoring the Collaborative Longitudinal Study of Personality Disorders (CLPS), a 10-year prospective follow-up of patients with four diagnoses: borderline, schizotypal, avoidant, and obsessive—compulsive (Gunderson et al., 2011). In each category, symptoms remitted with time, and many patients eventually functioned within a normal range. These findings clearly showed that improvement over time is standard for PD. Even if some patients continue to have difficulties with work and interpersonal relations, the vast majority no longer met criteria when evaluated after 10 years.

Another important prospective study, the McLean Study of Adult Development, compares BPD patients with patients with other PDs, all of

whom had been hospitalized (Zanarini, Frankenburg, Reich, & Fitzmaurice, 2012). This research has now lasted 20 years and seems ready to continue to evaluate patients into old age. The results have been essentially identical to those of the CLPS study in documenting that remission is the usual outcome and that once patients remit, they rarely relapse. However, although recovery was the rule, many patients had residual difficulties in psychosocial functioning. (This suggests that some patients may need additional treatment later in adult life.)

Similar findings emerged from the Longitudinal Study of Personality Disorders, which followed college students into adult life (Lenzenweger, 1999). There was good evidence for improvement over time, even for highly symptomatic disorders such as BPD.

In summary, patients with PDs have a reasonably good prognosis, and most do better than patients with mood disorders. It would be useful if clinicians knew more about these findings, because they might be less pessimistic and less reluctant to make PD diagnoses. Although none of the studies was able to test whether treatment makes a difference, it seems reasonable to consider that therapy aims to speed up the process of recovery and to make it more complete.

I now make a point of explaining the BPD diagnosis to all patients and informing them that while their condition is serious, they can be expected to recover. This hopeful message is supported by the scientific evidence, and many patients are glad to hear it. One told me, "I thought I was just this weird person who didn't respond to antidepressants, but now I know I have a completely different problem."

# Antisocial Personality Disorder

Por as long as he could remember, Johnnie had always been in trouble. He was a defiant and aggressive child, and his parents could not control him. He had been tossed out of several schools, and only finished Grade 9. Johnnie began smoking pot before puberty and then moved on to heavy use of alcohol and cocaine. By age 16, he was a drug dealer. Johnnie was often brought to the attention of the police for incidents ranging from theft to violent quarrels when intoxicated. His first sexual experiences were at age 12, and he went on to have many brief affairs without making a commitment to anyone. Although peripherally involved with organized crime, he was never loyal to anyone.

At 25, Johnnie, who was unemployed, was charged with assault after a drunken encounter with a policeman. In the past, his parents had always bailed him out of trouble, but this was a second offense, so this time they hired a good lawyer. He came to the psychiatric clinic asking for support

http://dx.doi.org/10.1037/14642-006

A Concise Guide to Personality Disorders, by J. Paris
Copyright © 2015 by the American Psychological Association. All rights reserved.

for a defense of mental illness. He even asked if the clinic could do a brain scan, which could show that he was not responsible for his behavioral problems.

Johnnie is a typical example of antisocial personality disorder (ASPD), a condition that is common but more often seen in forensic settings.

# HISTORY

Two hundred years ago, the only generally recognized mental illnesses were psychoses and severe depressions that required hospitalization (Shorter, 1997). There was not yet any concept of personality disorder (PD). From time to time, however, nonpsychotic patients would be referred to mental hospitals after committing serious crimes if these crimes involved violent or bizarre actions without an obvious motive. The term *moral insanity* was used to describe such cases (Berrios, 1993).

In the 20th century, a number of different (and often confusing) terms have been used to describe offenders who are callous or chronically criminal in their behavior. *Sociopathy* and *psychopathy* have been used for at least a hundred years, but they are not interchangeable (Rutter, 2006).

The first edition of the *Diagnostic and Statistical Manual of Mental Disorders* (*DSM–I*; American Psychiatric Association, 1952) included a category of *sociopathic personality*, which the second edition (*DSM–II*; American Psychiatric Association, 1968) went on to rename ASPD. To make things more complicated, the *International Classification of Diseases* (World Health Organization, 1993) chose the term *dissocial*. Yet although *antisocial personality*, *dissocial personality*, and *sociopathy* all describe more or less the same construct, *psychopathy* is different.

Hervey Cleckley (1964) wrote a classic book on psychopathy, which, in a clever turn of phrase, he titled *The Mask of Sanity*. For Cleckley, psychopaths were mentally ill, even if that wasn't immediately obvious. These were patients who, in addition to a criminal history, were charming, manipulative, and (most particularly) callous about other people's feelings. Cleckley thought psychopaths were born that way, and he included many dramatic tales of bad children coming from good families. However, his experience

#### ANTISOCIAL PERSONALITY DISORDER

was that of a private practitioner attending to children of the wealthy; a forensic specialist would have observed a different pattern, in which dysfunctional families produce dysfunctional children. This is what Lee Robins (1966) found in a landmark prospective study.

The Canadian psychologist Robert Hare (1999) has carried out a lifetime of research on psychopathy in forensic populations. His Psychopathy Checklist (PCL–R; Hare, 2003) is a structured interview used in hundreds of research studies. Hare (2003) has shown that scores on the PCL–R require a two-factor model: one mainly describing criminal behavior, the other describing personality traits of callousness and manipulativeness. In contrast, ASPD describes a wider range of people with less severe behavioral patterns (Coid, 2009).

# **CHARACTERISTICS**

The criteria for ASPD in *DSM*–5 (American Psychiatric Association, 2013) are based on a disregard for the rights of others (callousness, criminality, impulsivity, irresponsibility) and on a past history of conduct disorder (CD). Unlike other PDs, this diagnosis can only be made after age 18 years. The reason is that CD sometimes "burns out" by the end of adolescence. However, it is possible in *DSM*–5 to score severe callousness as a *specifier*. This procedure could help researchers identify patients who meet the more stringent requirements for psychopathy.

The construct of ASPD derives from the work of a sociologist (Robins, 1966) who followed up a large cohort of children with CD into adulthood. The main finding was that if children did not have CD, they would not develop ASPD as adults. However, not all children with CD became antisocial. That outcome is more likely in the presence of an earlier onset and greater severity (Zoccolillo, Pickles, Quinton, & Rutter, 1992). Also, when CD starts in adolescence, the clinical picture is more sensitive to social context, is associated with gang membership, and often remits by young adulthood (Moffitt, 1993).

ASPD can be mistaken for other diagnoses. For example, if these patients are arrested or put in jail, they may threaten or attempt suicide,

but that need not be a picture of clinical depression. Another mistake is that ASPD, when it presents in an emergency department, can be misdiagnosed as mania. Taking a careful life history and obtaining additional information from family members helps avoid these errors.

#### **PREVALENCE**

ASPD is a highly prevalent disorder. Moran (1999) described its frequency as between 2% and 3%, and the National Epidemiologic Survey on Alcohol and Related Conditions estimated prevalence at 3.7%. In all studies, the diagnosis is much more common in males than females. Among prisoners, approximately half will meet criteria (Black, Gunter, Loveless, Allen, & Sieleni, 2010; Fazel & Danesh, 2002; Moran, 1999).

However, ASPD appears to be rare in non-Western countries in which a more traditional way of life has been maintained. The prevalence is also low in Western countries that have retained stronger traditions. Thus, in a Norwegian survey, the prevalence was only 0.7% (Torgersen, Kringlen, & Cramer, 2001). In a study in Taiwan conducted approximately 30 years ago (Hwu, Yeh, & Chang, 1989), almost no cases could be found, either in urban or in rural samples. The results might be different today because Taiwan has become more like Europe and North America.

# RISK FACTORS

The risk factors for ASPD are biological, psychological, and social. Biological factors definitely increase the risk for developing ASPD. Criminality, impulsivity, and callousness, the key characteristics of the disorder, are, like other personality traits, clearly heritable (Mednick, Moffitt, & Stack, 2009). Some of these traits can identified early in development; in one longitudinal study, observations of aggressiveness during a 90-minute interview at age 3 predicted antisocial behavior at age 18 (Caspi, Moffitt, Newman, & Silva, 1996).

No biomarkers are known to correlate with antisocial traits, but Adrian Raine (2013), a researcher at the University of Pennsylvania, conducted a series of studies that point to brain differences in violent criminals. His

#### ANTISOCIAL PERSONALITY DISORDER

group used imaging to identify abnormalities in frontal white matter, as well as more subtle anatomical changes. The research suggests a problem with brain connectivity, affecting regions associated with the processing of emotion and the control of impulsivity. However, most patients with ASPD are not consistently violent, and many have never been in prison (Robins, 1966). Therefore, these results cannot be generalized to all individuals who meet diagnostic criteria.

Gene-environment interactions are important in the risk for ASPD. An adoption study by Cloninger, Sigvardsson, Bohman, and van Knorring (1982) found that interactions between temperament and family adversity accounted for the risk for ASPD risk better than either alone. In one of the most widely quoted studies in the history of psychology, Caspi et al. (2002) found that a combination of heritable risk (the gene controlling monoamine oxidase) and child abuse predicted antisocial behavior, yet neither of these risk factors did so by themselves. This finding remains controversial because it has not always been replicated, and it is worth noting that only a subgroup in this cohort developed antisocial characteristics.

Family dysfunction is a strong risk factor for ASPD (Black, 2013a). Many patients grow up in families characterized by chaos and violence, and physical abuse is common (J. Hill, 2003). Others suffer from a lack of the structure and discipline they need to control their impulsivity (Robins, 1966). A chaotic family environment may also reflect common heritable vulnerabilities in both parents and children. Unfortunately, the literature on psychosocial adversity is not extensive, reflecting the fact that patients with ASPD do not often come for treatment in mental health settings, and when they do, they are difficult to recruit for research.

Social factors play an important role in ASPD. Callousness and criminality have always characterized certain people, as history clearly shows, but, as noted in Chapter 3, there was a notable increase in the prevalence of ASPD after the Second World War (Rutter & Smith, 1995), suggesting that rapid social change can make these problems more prevalent. This hypothesis is also supported by cross-cultural differences in the prevalence of the disorder (Hwu et al., 1989).

In summary, only the most severe cases of ASPD point to a "bad seed" theory. In those patients, callousness seems to represent a defect, either in

the capacity for empathy or in the emotional consequences that accompany empathy. This is why some patients do well in combat situations during times of war but fail to adapt to the expectations of a peacetime life. For less severe cases, heritable biological risks may account for a smaller portion of the variance. Many patients grow up in dysfunctional families and are raised in bad neighborhoods and associate with problematic peer groups that do not provide them with positive role models. Thus, interactions among biological, psychological, and social risk factors are required to explain the pathways to this form of psychopathology.

## **OUTCOME**

The outcome of ASPD is similar to those of other impulsive disorders, such as substance abuse or borderline PD. In a unique study, Black, Baumgard, and Bell (1995) followed 71 patients for several decades who had found their way, at least briefly, into a mental hospital. Thirty years later, they continued to have serious problems holding a job and maintaining intimate relations. Yet these former patients no longer showed the behavioral patterns that had led them to be in repeated trouble with the legal system. Thus, ASPD shows some of the features of burnout seen in impulsive disorders such as substance abuse. For this reason, it is generally (but not always) safe to release middle-aged patients from prison. Even so, the prognosis of ASPD remains guarded. Black et al. (1995) observed that these patients, even when they no longer met diagnostic criteria, continued to be difficult people.

# MANAGING ASPD

Unless you work in a forensic setting or a substance abuse clinic, you may not often see patients with ASPD. Help seeking is not the forte of this population. However, cases of ASPD do sometimes appear for assessment in clinics. Zimmerman, Rothschild, and Chelminski (2005) found that 3.6% of outpatients meet diagnostic criteria for the disorder. A common scenario is for them to come on the advice of lawyers who hope to get them off a charge.

#### ANTISOCIAL PERSONALITY DISORDER

There has been concern about the potential for violence among patients with this diagnosis. In prisons, up to half of the population will meet criteria for ASPD (Black, Gunter, Loveless, Allen, & Sieleni, 2010). In response to some notorious crimes, the National Health Service in the United Kingdom set up units for "dangerous and severe personality disorder" (Mullen, 2007). However, most people with the diagnosis never kill anyone.

By and large, patients with ASPD do not benefit from talking therapies. The only study, never replicated, suggesting that standard methods might help them was conducted 30 years ago by Woody, McLellan, Luborsky, and O'Brien (1985), but the sample was small, the patients were already in treatment for substance abuse, and they were all clinically depressed. Because patients with ASPD have a high rate of substance abuse (Grant et al., 2004), drug rehabilitation (Fridell, Hesse, Jaeger, & Kühlhorn, 2008) may be the useful intervention. Successful withdrawal from addictive drugs may improve functioning even when patients continue to show features of their PD.

Another twist in the story is that some of the women therapists treat are attracted to antisocial men, like the proverbial moth to the flame. These men may be callous, but they can be surprisingly good at using empathy to manipulate women (as well as other people in their lives). Many of these couples end up in situations of domestic violence.

There is no reason at this point to doubt the overall conclusion that patients with ASPD are not suitable for psychotherapy, at least in the standard forms by which it is currently applied. Thus, one reason why making the diagnosis is important is that recognizing ASPD allows therapists to save their limited time for more treatable cases. If we did have a treatment for ASPD, whether biological or psychological, the demand would be enormous, given the high social cost of the disorder. At this point, however, therapy for substance abuse may be the most reasonable option we have to offer.

# Borderline Personality Disorder

Laura was a 29-year-old woman with a long psychiatric history who came to the emergency department with suicidal ideas. She had been seen as an adolescent for recurrent overdoses, cutting, and substance use, but she had not engaged in therapy. Laura finished high school and then supported herself for some time working as a stripper before taking a job in customer service. She described her unstable mood as a roller-coaster. Relationships with men began quickly and ended badly. Many of her boyfriends were abusive, and at least two were active criminals. Breakups were difficult and had often led to visits to the emergency department or to crisis clinics. When highly stressed, Laura felt depersonalized and paranoid and sometimes heard voices telling her to kill herself. She thought of suicide every day but still hoped that she could be treated for her problems.

This vignette describes a typical case of borderline personality disorder (BPD), the most important personality disorder (PD) in clinical

http://dx.doi.org/10.1037/14642-007

A Concise Guide to Personality Disorders, by J. Paris
Copyright © 2015 by the American Psychological Association. All rights reserved.

practice and one of the most extensively researched. BPD is common and can be identified in approximately 10% of patients in mental health clinics (Zimmerman, Rothschild, & Chelminski, 2005). That is why it plays such a prominent role in this book.

# CLINICAL FEATURES

Adolf Stern (1938), the psychoanalyst who first described BPD, thought that this kind of psychopathology lay on some kind of border between neurosis and psychosis. Despite this misleading theory, and a confusing choice of a name, Stern's description of what BPD patients are like remains contemporary. The term *borderline* is a misnomer and has only been retained for lack of a good alternative, much as *schizophrenia* does not really describe a split mind. Although alternative names have been suggested, terms such as *emotional dysregulation disorder* capture only one element of the clinical picture. We should not change the name until the condition is much better understood.

BPD has a closer resemblance to severe mental disorders than to other PDs. It has an unusually wide range of comorbidity and is associated with intense suffering and a wide range of symptoms, including dysregulation of mood and anxiety, substance abuse, and micropsychotic symptoms (Gunderson & Links, 2012; Paris, 2007c; Zanarini, Frankenburg, Khera, & Bleichmar, 2001). Because patients with this diagnosis suffer, they are highly treatment seeking (Zanarini et al., 2001).

BPD is an amalgam of traits and symptoms. A large body of research shows that the most prominent of its underlying traits are affective instability and impulsivity (Crowell, Beauchaine, & Linehan, 2009; Siever & Davis, 1991). Thus, many of its clinical features can be understood as reflecting emotion dysregulation, combined with a wide range of impulsive actions, both of which lead to serious and chronic interpersonal problems. These underlying traits of BPD are only partially egosyntonic. Patients complain about unstable mood, described as a "roller-coaster" of emotions in response to environmental stressors, and most acknowledge that impulsive actions get them into difficulty (Linehan, 1993; Paris, 2008a). Although

## BORDERLINE PERSONALITY DISORDER

impulsive actions, such as substance abuse, sexual promiscuity, and shop-lifting, can be egosyntonic, this is not the case for self-harm behaviors such as cutting or for overdoses. Finally, the cognitive symptoms of BPD, such as depersonalization, paranoid ideas, and transitory auditory hallucinations (Zanarini, Frankenburg, Wedig, & Fitzmaurice, 2013; Zanarini, Gunderson, & Frankenburg, 1990), are almost always egodystonic.

None of these dimensions account for the most severe symptoms associated with BPD, such as chronic suicidality and self-harm, but as patients recover, those symptoms tend to remit. Even after recovery, patients often retain "subsyndromal" features related to dysfunctional traits that do not change, most particularly mood instability that interferes with occupation and interpersonal functioning (Zanarini et al., 2007).

Recurrent overdoses, a feature seen in most patients with BPD (Soloff, Lynch, Kelly, Malone, & Mann, 2000), contrasts with a frequency of only 5% for lifetime suicide attempts in the general population (Kessler, Berglund, Borges, Nock, & Wang, 2005). Patients with BPD can also have suicidal ideation for years on end, a phenomenon that, when present, strongly suggests the diagnosis (Paris, 2008a). Although adolescents experiment with cutting, the pattern is usually transient and remits over the course of development (Moran et al., 2012). In contrast, BPD patients persist with self-harm for years, and because this behavior has modulating effects on dysphoria, can become addicted to cutting (Brown, Comtois, & Linehan, 2002). These phenomena are only partially explained by high levels of neuroticism (Morey et al., 2007; Morey & Zanarini, 2000), which describe emotion dysregulation without specifying behaviors that could be used for emotion regulation. Finally, approximately 10% of BPD patients eventually commit suicide (Paris, 2008a).

Because BPD is associated with such a broad range of symptoms, the disorder can seem complex and mystifying. Given the range of its clinical features, BPD might be called psychiatry's chameleon. It does not easily fit into a single diagnostic niche, and because symptoms are so widespread, it overlaps with many other diagnoses. The meaning of these comorbidities has sometimes been interpreted as meaning that BPD is "really" something other than a PD. Perhaps all they show, however, is that the *Diagnostic and* 

Statistical Manual of Mental Disorders (DSM) system uses similar criteria to describe many disorders. Because severe PDs disrupt psychosocial functioning, it should be no surprise that patients are depressed, anxious, and moody.

The idea that BPD is a variant of some other disorder has been influential among clinicians who find the construct confusing and treatment options murky. It has long been proposed that BPD is a form of depression (Akiskal et al., 1985), a variant of bipolarity (Akiskal, 2004), or a "complex" form of posttraumatic stress disorder (PTSD; Herman, 1992). Each of these theories is like the blind men and the elephant—they consider only one domain of the disorder, not BPD as a whole. The disorder is complex, but this diagnosis allows us to use a single construct to describe mood symptoms, anxiety symptoms, substance abuse, cognitive symptoms, and problematic relationships.

In this way, a diagnosis of BPD covers more ground. As is so often the case, a PD construct explains more than a number of accompanying "comorbidities." The mood symptoms of BPD are intrinsic to the diagnosis but reflect affective instability rather than classical mood disorder (Gunderson & Links, 2012). Depressed mood in BPD also does not respond to standard treatments, such as pharmacotherapy (Stoffers, Völlm, et al., 2012), indicating that it is an entirely different phenomenon.

Two comorbid diagnoses are particularly important when one is considering how to treat BPD. The first is substance abuse: Although most patients with BPD use substances on a casual basis, some have serious addictions. The second is eating disorders: Severe bulimia or anorexia nervosa can take over patients' lives. When substance abuse and eating disorders are severe, they have to be treated first, before the PD can be usefully addressed (Gunderson & Links, 2012).

Practitioners who do not recognize BPD (because all they see is a mood disorder) will apply clinical tools that are not ideal for PDs. Psychiatrists may prefer pharmacological interventions, while clinical psychologists may apply their training in cognitive—behavioral methods to manage anxiety or depression. Often it is only when these treatments fail that consultation with a specialist in PD occurs.

#### BORDERLINE PERSONALITY DISORDER

Yet experienced consultants can usually (but not always) make a diagnosis of BPD in a single interview. What one has to do is move beyond current symptoms and obtain a detailed life history. There can be heterogeneity, but for me, patients who meet most of the diagnostic criteria seem like peas in a pod. The key point is that knowing how to recognize this disorder allows you to make a choice of therapy guided by diagnosis.

# THE MULTIPLE DOMAINS OF BPD

BPD describes multiple domains of psychopathology (Paris, 2007c). Many clinicians have been impressed by its prominent mood symptoms. These patients have strong emotional responses to life circumstances and take a long time to calm down. However, they are not always chronically depressed, and intense anger is a feature of the disorder. Moreover, instead of episodes of depression or hypomania, in which affect is stable and unresponsive to the environment, emotions have a mercurial quality. Unlike depression (defined in the fifth edition of the *DSM* [*DSM*–5] as lasting for at least 2 weeks) or hypomania (defined in *DSM*–5 as lasting for at least 4 days), patients with BPD can be in a different mood from hour to hour, depending on what is going on in their interpersonal life.

Thus, one of the most characteristic features of the disorder is affective instability (AI; Koenigsberg, 2010), which is essentially identical to emotion dysregulation (ED; Linehan, 1993). AI and ED result from hypersensitivity to the environment and show rapid and intense responses, followed by a slow recovery, and occur when interpersonal interactions are conflictual (Russell, Moskowitz, Zuroff, Sookman, & Paris, 2007). Again, note that AI is different from clinical depression (during which nothing can cheer up the patient) or hypomanic episodes (during which patients remain "high" even when things go wrong). Also, unlike patients with mood disorders, BPD patients are prone to becoming angry, often to the point of screaming, breaking things, or even becoming physically violent.

Affective instability in BPD patients, unlike classical mood disorders, responds inconsistently, or not at all, to pharmacotherapy, including anti-depressants, mood stabilizers, and antipsychotics (Stoffers, Völlm, et al.,

2012). Finally, mood in BPD is associated with feelings of emptiness and chronic suicidality (Gunderson & Links, 2012).

BPD should be easy to diagnose, but because of its prominent mood features, it is often mistaken for bipolar disorder. These days almost every patient with BPD will be called bipolar by someone, particularly when they come to an emergency department. Moreover, there is a group of zealots for bipolarity who describe the mood shifts of BPD as lying within a "bipolar spectrum" (Akiskal, 2004). This point of view leads to the treatment of BPD with multiple medications that do not work, as opposed to the use of specialized psychotherapies that do work.

There are a number of reasons to reject the conclusion that BPD is a form of bipolarity (Paris, Gunderson, & Weinberg, 2007). BPD and bipolar disorders can co-occur, but their relationship is neither consistent nor specific. There are important differences in phenomenology. They do not show the same response to medication. Family studies point to clear distinctions, and it is unusual for BPD to evolve into bipolar disorder. Finally, research has not shown that these disorders have a common etiology.

The second core feature of BPD is impulsivity. Moeller, Barratt, Dougherty, Schmitz, and Swann (2001) defined this construct as describing (a) decreased sensitivity to *negative* consequences of behavior; (b) rapid, unplanned reactions to stimuli before complete processing of information; and (c) lack of regard for long-term consequences. Although other mental disorders also show impulsivity, there are a number of characteristic impulsive behaviors seen in BPD. These consist of chronic and recurrent overdoses and self-harm behaviors such as cutting. These are the symptoms that often bring BPD patients to the emergency department, where BPD can be found in approximately 40% of patients presenting with repetitive suicide attempts (Forman, Berk, Henriques, Brown, & Beck, 2004). One also sees a wide range of other impulsive actions, most particularly substance abuse, bulimia, and shoplifting (Zanarini et al., 2001).

BPD can be frightening because patients with this diagnosis frequently have chronic suicidality—threatening to kill themselves or making multiple attempts. This is why some therapists avoid treating them. Yet although suicidality in BPD is a primary object of clinical concern, it is chronic. Some

#### BORDERLINE PERSONALITY DISORDER

patients will think about killing themselves on a daily basis for years, and up to 10% of patients with BPD eventually do commit suicide; some come close after making serious attempts (Paris, 2008a). This outcome is more likely in patients who are unusually impulsive or aggressive (McGirr, Paris, Lesage, Renaud, & Turecki, 2007). Those who kill themselves are usually older (approximately 35–40 years of age) and have failed to recover from the disorder even in the face of multiple attempts at treatment (Paris, 2003).

Despite the long-term risk, 90% of patients with BPD never kill themselves (Paris, 2006). This is crucial for frightened clinicians to keep in mind. For the most part, suicidality in BPD is communicative—a way of turning up the volume when one does not expect to be heard. Paradoxically, maintaining the option of suicide can also be comforting, particularly when quality of life is low. Nonetheless, overdoses are common in BPD. Fortunately, most incidents involve taking pills impulsively and at non-lethal levels, but some patients raise the stakes higher, and a scenario can develop that looks like a version of Russian roulette, with the stakes being life and death.

It is also important to know that self-harm, particularly cutting, is not suicidal behavior. BPD patients may cut their wrists and arms (or their thighs and abdomen if they want to avoid having other people notice). The purpose, as reported by patients themselves (Brown, Comtois, & Linehan, 2002), is to relieve tension, not to die. Self-harm is used to break the vicious cycles associated with emotional dysregulation. Its effectiveness in managing painful affects explains why these behaviors, however problematic, can sometimes become addictive (Linehan, 1993).

In addition to affective instability and impulsivity, patients with BPD have highly unstable intimate relationships. Their closest attachments, usually to romantic partners, are characterized by clinging, fear of abandonment, and intense conflict (Gunderson & Links, 2012). Gunderson and Lyons-Ruth (2008) suggested primacy be given to the interpersonal aspects of the disorder. However, these features are also a consequence of emotion dysregulation. If what patients need from an intimate partner is a soothing response, and they don't get it, they may respond with anger, leading to further escalation of the conflict.

The most "borderline" aspect of BPD is its cognitive symptoms (Zanarini et al., 2013). About half of all patients experience transitory auditory hallucinations under stress, but they do not develop delusional elaborations of such experiences. One of the most common symptoms is hearing a voice saying the patient is bad and should die. Sometimes the experience seems temporarily real, but on reflection, patients realize that they are hearing their own thoughts spoken aloud.

Many BPD patients experience paranoid feelings: They are suspicious and think others, even strangers, are talking about them, but these features do not reach delusional proportions. Finally, depersonalization and derealization in BPD can be severe and prolonged. These symptoms are sometimes termed *dissociative*, but it is not clear that is a helpful term. By and large, they are not related to specific life experiences but reflect emptiness and interpersonal disconnection.

The clinical features of BPD are listed in *DSM*–5 (American Psychiatric Association, 2013), with five of nine required to make a diagnosis (i.e., a *polythetic* system). No single feature is necessary, and *DSM*'s approach has sometimes been called a "Chinese menu." The problem is that there are too many ways for patients with different symptoms to meet criteria for the same diagnosis. For that reason, some researchers prefer a more restrictive definition, grouping symptoms into symptomatic or trait dimensions and requiring more characteristic features to be present.

I have had long experience in using a semistructured interview developed at McLean Hospital in Boston, based on criteria developed 40 years ago by Gunderson and Singer (1975). The Diagnostic Interview for Borderlines, Revised (DIB–R; Zanarini, Gunderson, Frankenburg, & Chauncey, 1989) describes a more homogeneous group of patients than *DSM–5*. Symptoms are scored on four subscales (affective, cognitive, impulsive, and interpersonal), and diagnosis requires having most of them, with the last two subscales (impulsive and interpersonal), reflecting the core aspects of the disorder, being more heavily weighted. Patients cannot reach the cutoff point (a score of 8 of 10) without having multiple features of the disorder, and thus heterogeneity is greatly reduced.

Some patients who meet *DSM* criteria will not meet DIB–R criteria, either because they never had or no longer have a group of symptoms. For

#### BORDERLINE PERSONALITY DISORDER

example, patients who have partially recovered from BPD but have come to avoid intimate relationships will not score on the DIB–R, which requires a score of 8 of 10 for problems present over the past 2 years. One can modify this requirement to describe a group of "lifetime" borderline patients who no longer meet criteria and have "graduated" to a diagnosis of PD—not otherwise specified. As we will see, this kind of partial recovery is one of the most common outcomes for patients with BPD.

Because research on the characteristic features of BPD seems clearcut, why is it so often missed? To make a diagnosis, one must differentiate a PD from symptomatic conditions that resemble it or that may sometimes accompany it. The first of such conditions is major depression. There is little doubt that patients with BPD often present with symptoms that meet DSM-5 criteria for major depression. This leads to the common practice of diagnosing a depression but either ignoring the PD, or, under previous editions of DSM, writing a comment such as "Axis II, deferred." However, consider the low bar that DSM-5 makes for a diagnosis of "major depression." It only requires the presence of five of nine symptoms for a period as brief as 2 weeks. This leads to confusion between sadness or unhappiness and depressive illness (A. V. Horwitz & Wakefield, 2007). In BPD, patients can be depressed and then shift rapidly into another mood (Gunderson & Phillips, 1991). Even when they do cross the 2-week bar for a major depression, they do not respond to the same treatment as patients who have depression without a PD (Newton-Howes, Tyrer, & Johnson, 2006). Finally, depression in BPD is not an episode but a chronic state that is similar to a feeling of emptiness (Gunderson & Phillips, 1991).

A second confounding condition is bipolar disorder. As discussed earlier, because mood swings in BPD are rapid and mercurial, they do not meet the 4-day requirement for hypomania, which is, in turn, a requirement for a diagnosis of bipolar II disorder. However, for those who remain convinced that the PD is a bipolar variant, *DSM*–5 allows them to make a diagnosis of *bipolar disorder*, *unspecified* or *bipolar disorder*, *other specified*. The manual allows this category to be applied to patients who have the typical affective instability associated with BPD. (It can also be used for atypical presentations such as bipolar II without depression.) Although

one rarely sees this diagnosis in practice (no one really wants to use these residual categories), the very presence of this option reflects a belief in the validity of a bipolar spectrum.

The third is attention-deficit/hyperactivity disorder (ADHD). This diagnosis has become extremely common in adults and is considered by Batstra and Frances (2012) to be a fad producing an "epidemic." ADHD has become a frequent diagnosis in adults largely because it is the basis for drug prescriptions or specific interventions by psychologists (Paris, 2013d). However, not all attention problems in adults justify a diagnosis of ADHD. For one thing, you cannot make the diagnosis without documenting a childhood onset, which many clinicians fail to do. Moreover, PDs such as BPD can be associated with an unfocused style of life in which goals are not set or reached, combined with a pattern of impulsivity. This picture can be confused with the specific deficits of ADHD, but because there is no evidence that stimulants have any value in BPD, this is likely to be an error.

## **EPIDEMIOLOGY**

Most studies of the community prevalence of BPD find a rate somewhat less than 1% (Coid, Yang, Tyrer, Roberts, & Ullrich, 2006; Lenzenweger, Lane, Loranger, & Kessler, 2007; Torgersen, Kringlen, & Cramer, 2001), about the same as for schizophrenia. As already noted, higher frequencies are probably an artifact of conducting large-scale surveys with research assistants as interviewers.

Studies have been conducted in which men with BPD were examined separately, with few clinical differences from women (Paris, Zweig-Frank, & Guzder, 1994b). Although most BPD patients in clinical settings are female, community studies have found an equal prevalence in men and women (Lenzenweger et al., 2007). The question is how to account for this difference. The most likely explanation is that women are more likely to seek help. Notably, patients with BPD who die by suicide are usually male, and most have only intermittent contact with the mental health system (Lesage et al., 1994). Men with the same level of psychopathology are more likely to live in the world of substance abuse.

#### BORDERLINE PERSONALITY DISORDER

## A BIOPSYCHOSOCIAL MODEL OF BPD

The biopsychosocial model (Engel, 1980) proposes that the etiology of mental disorders is based on complex interactions among biological vulnerability, psychological adversity, and social stressors. BPD is a good example of its usefulness.

## **Biological Factors**

The heritability of BPD accounts for more than half of the variance (Reichborn-Kjennerud et al., 2013). Yet how do genes affect brain function to make people vulnerable to the disorder? We do not know the answer. There is no such thing as a gene "for" BPD, and many variations probably play a role.

What we do know is that the brain functions differently in patients with BPD (New, Goodman, Triebwasser, & Siever, 2008). This is a large and complex literature, but the most consistent findings are related to the traits of affective instability and impulsivity that underlie the diagnosis. Neuroimaging findings have been well summarized by New, Goodman, Triebwasser, and Siever (2008). As discussed in Chapter 3 of this volume, there is a diminished top-down control of affective responses related to deceased responsiveness of specific regions of prefrontal cortex, as well as functional and volumetric changes in subcortical structures such as amygdala, hippocampus, and caudate. Neuroendocrine research also points to a role for serotonin in the impulsivity that characterizes BPD.

Evidently, BPD patients are wired differently. However, there is no predictable relationship between abnormalities in connectivity or neuro-transmission and the disorder. Without psychosocial adversity, temperamental characteristics remain temperamental. Patients who have recovered from BPD may continue to be highly emotional, but they learn with time to contain intense feelings and not act on them in ways that are self-destructive (Paris, 2003).

## **Psychological Factors**

The psychosocial adversities that convert temperamental vulnerability to a diagnosable PD have been widely documented, with most findings

pointing to child abuse, emotional neglect, and dysfunctional families (Paris, 2008a; Zanarini, 2000). Again, there is no predictable relationship between these risks and the development of the disorder. As shown by our own research on sibling pairs that are discordant for BPD (Laporte, Paris, Guttman, & Russell, 2011), childhood adversity has a different effect on people with a vulnerable temperament than on those who have traits that promote resilience. Also, about a third of patients with BPD will not have experienced serious adversity in childhood (Paris, 2008a).

There is no simple relationship between trauma and BPD. This is why the concept of *complex PTSD* or *complex trauma* (Herman, 1992) is misleading. It supposes that disorders like BPD are a direct result of childhood experiences, which is not the case. Moreover, throwing around words such as *complex* is little more than hand-waving and explains nothing about the truly complex pathways that lead to BPD.

To understand BPD, one must take gene–environment interactions into account. For example, a recent study of BPD traits that used a twin sample to assess genetic influences failed to find a direct relationship between child abuse and BPD traits (Bornovalova, Huibregste, & Hicks, 2013).

To understand these mechanisms, one must consider that heritable traits of affective instability make children more sensitive to their environment, which can then lead to vicious cycles in which negative events further feed instability. Also, traits of impulsivity make children more likely to choose behaviors that reduce emotion dysregulation in the short run but increase it in the long run. Moreover, traumatic events are processed differently in those who are temperamentally vulnerable, but similar adversities produce either minor or no sequelae in most children (Fergusson & Mullen, 1999; Paris, 2000).

Our research group conducted a family study (Laporte et al., 2011) in a sample of severely ill women with BPD and their female siblings. Of 56 pairs of sisters, only three were concordant for the disorder. Yet both reported having been exposed to the same traumatic events. The sister who did develop BPD was distinguished by a different personality trait profile, pointing to the role of a problematic temperament.

In an earlier set of studies (Paris, Zweig-Frank, & Guzder, 1994a, 1994b), we found that approximately a third of BPD patients report severe abuse

#### BORDERLINE PERSONALITY DISORDER

and neglect, a third had milder forms of adversity, and another third had a relatively normal childhood. These findings support a model in which developing BPD reflects what researchers refer to as a *common final pathway*, or what Cicchetti and Rogosch (2002) termed *equifinality*. Just as the same outcome can arise from different sources, different outcomes can arise from the same risk factors (i.e., *multifinality*).

Although many patients with BPD have a traumatic childhood, others will have suffered from having their intense emotions misunderstood. This is why Linehan (1993) hypothesized that BPD arises from interactions between temperamentally based emotion dysregulation with *invalidation* (the failure of family members to understand and support children when they are upset). Most research (Paris et al., 1994a, 1994b; Zanarini, 2000) suggests that in patients who have not had a traumatic childhood, feelings of emotional neglect are more important.

Thus, BPD patients can come to the same point after different experiences. Some come from highly dysfunctional families, and when we obtain such histories, we are tempted to say, "of course, this is why they have BPD." However, research showing the ubiquity of resilience suggests otherwise. When BPD patients come from relatively normal families in which their siblings do not have significant psychopathology, invalidation of emotions is a more likely mechanism. Yet for many years, some clinicians assumed that patients with BPD *must* have suffered child abuse; when patients did not actually remember such events, it was claimed that they had repressed these memories (Herman, 1992). There was a time in the late 1980s and into the 1990s when an epidemic of false memories of abuse, encouraged by therapists who believed in such theories, spread through the psychotherapy community (McNally, 2003). This was one of the most malignant fads in the history of psychiatry and clinical psychology. Although the tide of error has receded, there are still therapists who believe in these premises.

#### **Social Factors**

There is indirect evidence, based on increases in associated symptoms (e.g., suicidality, substance abuse), that BPD is more likely to develop when the social environment is stressful or changes too rapidly (Paris &

Lis, 2013). This mechanism could explain why BPD is a relatively recent diagnosis in psychiatry. It is also worth noting that some of the symptoms associated with the disorder are socially contagious, as in the current epidemic of self-harm in which some adolescents imitate these behaviors (Klonsky, 2007).

## An Integrated Model

None of these risk factors produce a disorder by themselves, and all need to be placed in the frame of a biopsychosocial model. Only the effects of multiple and interacting risks can predict outcome, particularly one that can arise from several pathways (Beauchaine, Klein, Crowell, Derbidge, & Gatzke-Kopp, 2009).

Interactions between genes and environment are also consistent with a stress—diathesis model of mental disorders (Monroe & Simons, 1991). This leads to a general principle: Patients do not develop BPD unless they have an underlying temperamental vulnerability. In this respect, BPD resembles most severe mental disorders. There is an important role for environmental risks, but children who are resilient to childhood adversity probably have a different temperament from those who are not.

In summary, the recipe for cooking a case of BPD has many ingredients. This is why treatment requires multiple interventions that can also be biological, psychological, or social.

## BPD OVER THE LIFE SPAN

## **Childhood Precursors and Adolescent Onset**

BPD usually becomes clinically apparent during adolescence, and it may actually be more frequent at that stage than in young adult populations (Chanen & McCutcheon, 2013). Zanarini, Frankenburg, Khera, and Bleichmar (2001) found that most patients with BPD develop symptoms soon after puberty, even if they do not seek help until a few years later. Our own group (Biskin, Paris, Renaud, Raz, & Zelkowitz, 2011) followed a group of patients treated for BPD in mid-adolescence for 5 years and found that

#### BORDERLINE PERSONALITY DISORDER

about half continued to meet criteria for the diagnosis in young adulthood. This is consistent with the course of BPD in young adulthood, in which some patients remit early but others take 10 to 15 years to recover (Gunderson et al., 2011).

Many clinicians are reluctant to diagnose PDs in teenagers because they may only be showing "adolescent turmoil" and may see problems as "just a phase." Actually, it is important to make the diagnosis in adolescent patients and not to dismiss these symptoms. It is also important not to attribute them to mood disorders without PD.

Although I often hear that *DSM* does not permit clinicians to make a valid PD diagnosis in adolescents, this idea is based on a mistaken reading of the manual. *DSM*–5 allows for early diagnoses of any PD, with the exception of antisocial PD, but makes a point of requiring a chronic rather than an episodic course (at least a year of symptoms). Unfortunately, not everyone has read this passage carefully.

The onset of BPD at puberty is similar to the onset of mood disorders, psychoses, and addictions, all of which tend to present for the first time at that developmental landmark. Yet although the childhood precursors of ASPD are well known, we do not know what BPD patients were like as children. If we did, we might be in a position to consider strategies for prevention.

One possibility is that symptoms have been present before puberty but are more internalizing than externalizing, and thus they do not attract attention in the same way as conduct disorder. Some patients will describe feeling suicidal as children, although this is far from universal. Others state that life was manageable until puberty, when "all hell broke loose." Different developmental pathways can lead to a similar outcome.

In recent years, researchers have designed studies to examine the developmental precursors of BPD in children. Because women are more commonly seen in practice, the studies have focused on high-risk populations of girls. They have identified symptoms appearing before puberty that resemble BPD and that may suggest an early onset of the disorder (Stepp, Pilkonis, Hipwell, Loeber, & Stouthamer-Loeber, 2010; Zanarini et al., 2011). By age 11, some girls are affectively unstable, impulsive, and

suicidal. The results of research thus far suggest that these girls have behavioral symptoms that can also be diagnosed as conduct disorder, oppositional defiant disorder, or ADHD and that they grow up in high-conflict dysfunctional families. Long-term follow-ups of these cohorts should shed light on whether women identified at this earlier stage are at risk for an adult diagnosis of BPD.

## **Long-Term Outcome**

BPD, once believed to be a lifelong problem, turns out to have a better prognosis than severe mood disorders. Most patients stop meeting diagnostic criteria by middle age. This favorable outcome was first shown by retrospective (follow-back) studies (McGlashan, 1986; Paris, Brown, & Nowlis, 1987; Stone, 1990) and later confirmed by large-scale prospective studies (Gunderson et al., 2011; Zanarini, Frankenburg, Reich, & Fitzmaurice, 2012).

To summarize these findings, most patients with BPD do not meet criteria for the diagnosis at age 40, and many recover before age 30. Impulsive symptoms remit early, whereas affective symptoms are slower to change and are associated with some degree of interpersonal and cognitive problems. Some BPD patients recover completely, but many continue to have some degree of residual psychosocial dysfunction (Zanarini et al., 2012). Thus, even when patients are no longer acutely symptomatic, they may not find stable employment, stable partners, or become parents. In our 27-year follow-up (Paris & Zweig-Frank, 2001), although most BPD patients were working at age 50, only half were living with another person, and only half had ever had children.

Some of the most interesting and clinically relevant findings of outcome research concern suicide. The frequency of death by suicide in BPD is approximately 5% in prospectively followed cohorts and approximately 10% in follow-back studies (Paris, 2006). Although younger BPD patients frequently threaten to kill themselves, and frighten us by making such threats, the vast majority of suicides occur much later in the course of the illness. At least 90% of BPD patients, after being chronically suicidal for years, choose to go on living.

#### BORDERLINE PERSONALITY DISORDER

Our own research found that after 27 years of follow-up, the mean age of death by suicide was 38 (Paris & Zweig-Frank, 2001). Stone (1990) found a similar result: After 15 years, the mean age of suicide was 30. Thus, the highest risk of suicide in BPD does not lie in the early 20s, the period when patients are most likely to threaten suicide—and to alarm therapists. This was one of the most important research findings of my career. It confirms that in most patients with BPD, suicidal behaviors are largely interpersonal and communicative. The time when patients are most likely to kill themselves is after many treatments have been tried and have failed. By that point, while others have recovered from BPD, those most at risk have failed to improve.

#### TREATMENT

Unlike antisocial PD, BPD is treatable, and the main method of treatment is psychotherapy. Not any old psychotherapy will do, however. Standard methods do not work well. There are now a number of specific programs designed to meet the challenge of BPD. The most important of these is dialectical behavior therapy (DBT; Linehan, 1993). This is because DBT has the most evidence behind it and because most other methods apply rather similar principles. Anyone who wants to treat BPD has to understand this method and apply it to some extent.

However, despite the seminal contributions of DBT, it is lengthy, expensive, and not the only way to manage patients with BPD. This book argues for an integrative model that makes use of all the best ideas from all therapies currently available. It also argues for a briefer, more streamlined approach that makes therapy accessible for more patients. I discuss these recommendations in Chapter 10.



# Narcissistic Personality Disorder

George was a 27-year-old man who had recently separated from his wife. She was an unusually attractive woman whom George considered a great catch; he was often gratified to see how other men envied him. However, their relationship floundered on his inability to meet her emotional needs or to compromise his lifestyle. Within a few years, he had become chronically unfaithful and had begun to use cocaine regularly. Although George had a good education, his performance at work could best be described as disappointing. Initially his employer was impressed with his creativity and offered him significant responsibility in the business, but George had a way of making big promises and not following through.

Although George was not clinically depressed, he sought therapy, which he saw as necessary for "self-realization." He also wanted to be treated by a therapist with a notable name and reputation.

This vignette describes a typical case of narcissistic personality disorder (NPD), a condition that is common in clinical practice but that presents a challenge for standard methods of psychotherapy.

#### HISTORY

NPD is a controversial diagnosis. It has never been listed in the various editions of the *International Classification of Diseases*. It was included in the third edition of the *Diagnostic and Statistical Manual of Mental Disorders* (*DSM–III*; American Psychiatric Association, 1980), but as had been the case for borderline personality disorder (BPD), its close association with psychoanalysis limited its acceptance in the clinical community. Moreover, those who wrote about treating patients with NPD (Kernberg, 1976; Kohut, 1970) offered complex metapsychological speculations that were difficult to understand and not subject to empirical testing. More recently, however, NPD and narcissistic traits have been the subject of formal research (Campbell & Miller, 2011; Ogrodniczuk, 2013).

NPD had a rocky course in the development of the fifth edition of the *DSM* (*DSM*–5; American Psychiatric Association, 2013). Some said, jokingly, that this category had to be in the manual because narcissists can't stand being ignored, but the strength of the supporting data for the construct was questioned. At first, as the *DSM*–5 proposals were being prepared, NPD was slated for elimination. Later, it was restored within the hybrid model, in which diagnosis depends on a characteristic trait profile. This definition can be found in Section III of *DSM*–5. However, in Section II of the manual, NPD was listed, along with other PDs, in the same way as in the fourth edition of the *DSM* (*DSM*–*IV*; American Psychiatric Association, 1994).

The future of the NPD diagnosis will depend on research. Studies have long been thin on the ground. Surveys also suggest that these patients do not present in large numbers in hospital centers (Zimmerman & Mattia, 1999). If they seek therapy at all, NPD patients will be found in private offices. Moreover, unlike BPD, NPD does not cause enough trouble to inspire clinicians to embark on systematic investigation.

Nonetheless, NPD is currently enjoying a revival, in part due to research by psychologists on narcissistic traits that are continuous with the disorder

#### NARCISSISTIC PERSONALITY DISORDER

(Campbell & Miller, 2011). The diagnosis describes difficulties that are almost purely egosyntonic, which helps explain why NPD is not common in hospital clinics.

Even so, research on narcissistic traits provides stronger support for the diagnosis than for some of the other categories that were never under threat of removal from *DSM*–5. Because traits are more continuous with NPD than is the case for either antisocial PD or BPD, what is found in people who have these traits will be applicable to those who meet criteria for a formal diagnosis.

## CLINICAL FEATURES

NPD is described in *DSM*–5 as characterized by an expectation to be recognized as superior and special (without superior accomplishments); a need for constant attention and admiration associated with envy; and with a preoccupation with success, attractiveness, power, and intelligence. NPD patients lack the ability to empathize with the feelings or desires of others, are arrogant, and expect special treatment. This seems like a witches' brew of egotism. Yet narcissistic patients can be initially attractive to other people, even if they always end up disappointing them.

As discussed in Chapter 2, NPD is an example of a disorder that is highly egosyntonic and that represents a pathological exaggeration of personality traits. Research supports the idea that narcissism and NPD lie on a continuum (Campbell & Miller, 2011). However, if patients do not recognize that their traits are problematic, that is just what makes NPD a clinical problem.

## RISKS AND BENEFITS OF NARCISSISM

Narcissistic traits often lead to significant problems in work and relationships (Cain, Pincus, & Ansell, 2008). People with these characteristics overvalue themselves, are insufficiently self-critical, and respond to reasonable criticism with hurt and rage. When they suffer reversals in life, they are more likely to crash than to bounce back.

Even so, a small dose of narcissism can be adaptive in some contexts (Beck & Freeman, 2002). Being ambitious can be a good thing, and one cannot entirely fault people with unusual talents for being self-promoting, but people need to be willing to put in the effort required to make themselves a success, rather than expect rewards out of entitlement alone. We also need the interpersonal skills to work collaboratively with others.

Pathological narcissism differs from normal self-esteem. It is based on feelings of entitlement, as opposed to grounding one's sense of value in objective accomplishments (J. D. Miller, Gaughan, Pryor, Kamen, & Campbell, 2009). Narcissism is associated with manipulativeness and a lack of empathy, and this is why patients with NPD inevitably get into conflict with people who do not support their grandiosity. That is also why they lose jobs and intimate partners, even as they complain of being unappreciated.

## **PREVALENCE**

Researchers have examined the prevalence of NPD in community and clinical populations. The National Epidemiologic Survey on Alcohol and Related Conditions study (Stinson et al., 2008) reported a surprisingly high prevalence (6.2%), yet once again the numbers seem inflated. The cutoff between traits and disorders requires clinical judgment, and ratings in this study were made by research assistants rather than by trained clinicians. A reanalysis of the same data applying a different cutoff for diagnosis (Trull, Jahng, Tomko, Wood, & Sher, 2010) yielded a much lower estimate: 0.7% for males and 1.2% for females. A systematic review of other published studies (Dhawan, Kunik, Oldham, & Coverdale, 2010) also found a mean community prevalence of 1%. The lower figure makes sense but still indicates that NPD is far from uncommon.

In clinical populations, NPD is seen somewhat more frequently. Some clinicians believe that these patients do not often present to clinics. Yet using systematic interviews, the largest study reported a prevalence of 2% (Zimmerman, Rothschild, & Chelminski, 2005). This suggests that despite the highly egosyntonic nature of NPD, patients with this diagnosis seek help when their lives go wrong. Research shows many patients with

#### NARCISSISTIC PERSONALITY DISORDER

NPD are distressed, lonely, and have poor social functioning (J. D. Miller & Campbell, 2010). They can become demoralized when their attempts to get others to meet their needs prove unsuccessful.

Fortunately, narcissism declines with age (Foster, Campbell, & Twenge, 2003). Most of us learn life's lessons and realize that we are not the center of the universe—or of anybody else's life, for that matter. For patients with a clinical diagnosis of NPD, that kind of learning may not take place. For this reason, they are more likely to be divorced and to become unemployed (Ronningstam, 2011).

Some researchers have suggested that not all NPD patients are overtly grandiose and that this disorder can also take a "covert-vulnerable" form (J. D. Miller, Hoffman, Campbell, & Pilkonis, 2008; J. D. Miller & Maples, 2012). That construct may have merit, but it expands the definition of NPD considerably. If grandiosity is not required, the diagnosis will have a much higher prevalence. (This could be one more example of how *DSM* diagnoses tend to expand with time.)

There has been some discussion in the literature as to whether narcissism is increasing in contemporary culture. Based on cohort changes in self-report data on the Narcissistic Personality Interview (NPI) from the 1960s to the 1980s compared with recent times, Twenge (2011) hypothesized a rapid increase in narcissism over recent decades. Twenge proposed that overindulgent parenting, a fad for-self-esteem, and a set of social expectations that apply to celebrities have created a culture that promotes narcissism. This idea has attracted attention, both in the professional community and in the popular media. One recent study (Park, Twenge, & Greenfield, 2014) found a recent decrease in narcissism scores associated with the economic recession that began in 2007.

However, in the absence of confirmation by prospective studies, the level of narcissism in the general population is difficult to measure. The jury is still out on this issue. My own view (Paris, 2013b), in accord with a concept developed by the American historian Christopher Lasch (1979), is that the decline of collectivism and the rise of individualism, which started to accelerate about a hundred years ago, was the beginning of our present "age of narcissism."

## RISK FACTORS

Behavioral genetic studies of NPD (Torgersen et al., 2000), as well as of narcissistic traits (Vernon, Villani, Vickers, & Harris, 2008), have identified a heritable component that accounts for about 40% of total variance. This suggests that individuals are not likely to develop NPD without having a trait profile that makes them vulnerable. It is therefore logical that narcissistic traits, which can be observed in children (Twenge & Campbell, 2009), should precede the disorder. The existence of narcissism in prepubertal children has been documented by several research groups (P. L. Hill & Roberts, 2011; Kerig & Stellwagen, 2010; Tackett & Mackrell, 2011; Thomaes, Stegge, Bushman, Olthof, & Denissen, 2008). We need prospective follow-up of these cohorts into adulthood. Even if some children are prone to be boastful and grandiose (as many are), that would only be of concern if such characteristics do not mature in the course of development.

The psychological risk factors that amplify narcissistic traits to pathological narcissism have also been examined. Preliminary findings suggest a role for permissiveness in grandiose narcissism, and for cold overcontrol in vulnerable narcissism (Horton, 2011; Horton, Bleau, & Drwecki, 2006). These familial influences could also be augmented by cultural trends associated with the social reinforcement of grandiosity (Twenge & Campbell, 2009).

#### **TREATMENT**

There are surprisingly few data telling clinicians what works (and what doesn't work) for NPD. No clinical trials have ever been published. A large handbook of NPD and narcissism (Campbell & Miller, 2011) and an edited book on treatment (Ogrodniczuk, 2013) describe many potential methods of therapy, but none have yet been supported by clinical trials. In my view, we need a therapy package that is specific for NPD and that targets the most problematic areas of this disorder. This is what Linehan (1993) did for BPD, but NPD may require its own set of methods.

Psychoanalysts developed the NPD construct but usually relied on case histories with suspiciously happy endings. One might think that cognitive behavior therapy, which has a stronger research tradition, would have come

#### NARCISSISTIC PERSONALITY DISORDER

up with some data, but no studies exist to support any of the ideas that have thus far been proposed.

These problems are all too common in the mental health field. The result is that we are forced to fall back on clinical experience. I can offer a little of my own, for what it is worth. In my younger days I treated quite a few patients with NPD. Like other people, I found them intriguing, at least initially. I saw them as people with potential who were wasting their talents. I was too inexperienced to see that mediocre abilities can be inflated by skills in self-promotion. At the time, I lived in an intellectual climate in which several of my colleagues were intensely interested in NPD, and several were commuting regularly to Chicago to get supervision from Heinz Kohut (the guru of NPD in the 1970s).

I gradually came to realize something every aspiring therapist should know. Patients with NPD like to talk and to be listened to attentively, but they are reluctant to change. Psychotherapy meets some of their needs but does not reduce their sense of entitlement. More generally, patients don't take therapy seriously unless they take ownership for their problems and suffer enough to feel psychic pain. These characteristics are deficient in NPD. Today, if I were asked which category of PD I would prefer to treat in therapy, it would almost certainly be BPD. These patients may be difficult, but they are unhappy with themselves, making them more motivated for treatment.

What might a program specifically designed for NPD look like? It would need to focus on the core traits and behavioral patterns that characterize this diagnosis. For example, patients would need to be taught why entitlement is a bad strategy and how to replace grandiosity with commitment, compassion, and a passion for making this a better world.

The problem with therapy for NPD is that people with this diagnosis may come for treatment because they are unhappy, but they are unlikely to recognize the nature of their problem. Narcissistic traits are highly egosyntonic, and it may not even be possible to educate patients about their diagnosis. Moreover, treatment that is not well focused may do a better job at feeding emotional needs than at producing life changes. To counteract this tendency, we need to develop a model of individual therapy that is specifically designed to modify narcissistic traits. This is an ambitious goal, but no more so than in BPD, a condition long considered to be untreatable.

# Other Personality Disorders

This book has devoted separate chapters to personality disorders (PDs) with a good research base. Most of the others have a smaller literature, so I discuss them more briefly here.

Schizotypal PD has a large empirical literature, mostly stimulated by a group led by Larry Siever of Mount Sinai Hospital in New York (Chemerinski, Triebwasser, Roussos, & Siever, 2013). However, schizotypal PD may not actually belong in the group of PDs; in the *International Classification of Diseases (ICD)* it is considered a mild form of schizophrenia (World Health Organization, 1993), whereas the fifth edition of the *Diagnostic and Statistical Manual of Mental Disorders (DSM–5*; American Psychiatric Association, 2013) lists it in both groupings.

The other PDs listed in *DSM*–5 have attracted little research interest. Four (histrionic, schizoid, paranoid, and dependent) were removed in the proposals for *DSM*–5 but restored to the classification when the hybrid

proposal was rejected. The retention of two other categories (avoidant and obsessive—compulsive) may have been based on their having been examined for long-term outcome in the Collaborative Longitudinal Personality Disorders Study (CLPS), even though research on these categories remains thin.

## SCHIZOTYPAL, SCHIZOID, AND PARANOID PDs

These three categories all lie on the schizophrenic spectrum and are associated with a family history of psychosis (Siever & Davis, 1991). The hybrid proposal in Section III of *DSM*–5 folds them into a single category of schizotypal PD. However, they remain separate in Section II.

Gerald was a 27-year-old man who lived alone and was unemployed. He came to clinical attention after an altercation with the police, following a complaint that he had failed to clean up his yard. Gerald had never trusted people and had a vague feeling of threat around strangers. He had no friends and had never had an intimate relationship of any kind. He spent his days watching videos and sometimes went out to the mall. He was a visibly strange man, who rarely made eye contact, but Gerald did not complain that he felt lonely. In his mind, he simply marched to a different drummer.

This vignette is a typical example of schizotypal PD, characterized by social isolation and odd behaviors and thoughts. It is often considered a subclinical form of schizophrenia, but since patients with these diagnoses rarely go on to develop frank psychosis, even though schizotypal PD is cross-referenced in the schizophrenia section of *DSM*–5, it continues to be classified as a PD. This may or may not be a wise choice. For example, in earlier editions of *DSM*, a category called *cyclothymic personality* was considered a PD but was eventually moved to the bipolar section. The *ICD* system never considered schizotypal disorder to be a PD, and in the second edition of the *DSM*, a similar condition was called *simple schizophrenia*, that is, schizophrenia without overt psychotic symptoms.

Research on schizotypal PD shows that it shares biomarkers with schizophrenia. Both conditions show abnormal saccadic eye movements, which is a biomarker associated with psychosis, and neuroimaging findings are similar in both diagnoses as well (Chemerinski et al., 2013). What

#### OTHER PERSONALITY DISORDERS

makes schizotypal patients different is outcome. In the CLPS (Gunderson et al., 2011), their symptoms had the same tendency seen in other PDs for improvement over time and did not progress to psychosis.

Schizoid PD is characterized by a lack of interest in social relationships and emotional coldness. In other words, it has features of schizotypy without resembling psychosis. However, the distinctions between schizoid and schizotypal PDs may only be a matter of degree. In fact, the *DSM*–5 proposal in Section III eliminates schizoid PD, folding it into the schizotypal category.

Paranoid PD is characterized by pervasive, long-standing suspiciousness. The alternative model also folds it into schizotypal PD. This would have been a good decision, but implementation will have to wait for another edition of the *DSM*. As for community prevalence, the National Comorbidity Survey Replication (NCS–R; Lenzenweger, Lane, Loranger, & Kessler, 2007) found 3.3% for schizotypal, 4.9% for schizoid, and 2.3% for paranoid. However, some community surveys suggest differential heritability. Torgersen et al. (2000) reported a heritability of .61 for schizotypal, .29 for schizoid, and .28 for paranoid.

In clinical settings, these PDs are not common. Zimmerman et al. (2005) found a prevalence of 0.6% for schizotypal and 1.4% for schizoid; in the same study, 4.2% met criteria for paranoid PD. However, if patients with these diagnoses do not stay long in the mental health system, that might explain why so little research has been done on them.

The definitions of these categories are somewhat arbitrary and depend on clinical tradition. Although paranoid PD can be found in all editions of the manual, dating back to the first edition of the *DSM*, it has nine criteria, each of which describes suspiciousness in a different context. We are probably looking at a trait that lies on a continuum, not a disorder describing interactions between multiple trait dimensions.

#### HISTRIONIC PD

This category tells a story about the history of psychiatry. Decades ago, a category of *hysteria* was popular, used to describe a wide range of patients, some of whom had conversion or other somatic symptoms, and some of

whom had personality traits associated with stimulus-seeking, high extraversion, and a dramatic style of communication (Scull, 2009). Eventually, hysteria disappeared from psychiatry and was split into several pieces, one of which was hysterical personality, a category seen in earlier editions of *DSM*. In the third edition of the *DSM*, (*DSM–III*) the disorder was renamed *histrionic* to emphasize the dramatic flair believed to characterize these patients. One of the criticisms often applied to this category is that it might reflect a gender bias (Funtowicz & Widiger, 1999). Diagnoses of histrionic PD are usually made in women, with seductiveness as a diagnostic criterion.

Typical cases of histrionic PD may occasionally be seen in the clinic, but after 50 years in *DSM*, it remains unsupported by research, so it was time to remove it from the manual, and its continued presence is a historical anomaly. There are also no data on treatment for histrionic PD. The Cochrane group has published a protocol for reviewing this literature (Stoffers, Ferriter, et al., 2012), but there were no systematic studies for them to review and was withdrawn in 2014.

## AVOIDANT PD

Joy was a 28-year-old woman working in a call center. She could cope with this task, which gave her distance from difficult clients. She lived alone and spent weekends at the gym or doing photography. Joy had a few old friends but was too anxious about rejection to seek new ones. Joy would not accept invitations to parties or to other social events where she would have to chat with new people. However, when Joy remained single as her friends went on to marry and raise children, she felt lonely and left out of life.

This is a typical example of avoidant PD, a diagnosis introduced in *DSM–III* and that was a brainchild of the psychologist Theodore Millon. It describes patients who are socially anxious, and who, unlike patients with schizoid PD, desire relationships but are too afraid of rejection to take the chance of forming them. Torgersen et al. (2000) found the avoidant category to be the most common PD in Norway, with a prevalence of 5%, and the NCS–R (Lenzenweger et al., 2007) had exactly the same number. Not only are these patients common in the community, but Zimmerman

#### OTHER PERSONALITY DISORDERS

et al. (2005) found that 14% of patients in a large psychiatric clinic met criteria for the diagnosis, much more than patients with a Cluster A diagnoses. Torgersen et al. (2000) found the heritability of this disorder to be .28, much lower than other disorders in Cluster C.

Avoidant PD has a large overlap with social anxiety disorder (Sanislow, da Cruz, Gianoli, & Reagan, 2012). Thus, the small literature on treatment, as reviewed by the Cochrane group (Ahmed et al., 2012), describes cognitive behavior therapy methods developed for social anxiety. There is no specific protocol for treating avoidant patients, and there have been no clinical trials, although it is possible that some of the methods developed for social anxiety are also effective in avoidant PD.

#### DEPENDENT PD

This category is defined in a way that might lead one to ask, "How many ways can people be overly dependent?" Like paranoid PD, it does not really describe a complex of traits, as in other PDs, but a single trait in multiple contexts. This is a heritable dimension, as shown by Torgersen et al. (2000), who found a coefficient of .57. Community prevalence is low, with Lenzenweger et al. (2007) reporting 0.6%, and clinical prevalence is only 1% (Zimmerman et al., 2005).

A research group led by Adelphi University psychologist Robert Bornstein (2005) has been consistently interested in dependent PD. His studies examine the consequences of dependent personality traits, which lead to difficulties later in life as social networks narrow. However, this literature remains thin (Disney, 2013). There is no specific protocol for treating dependent patients, and there have been no clinical trials.

## OBSESSIVE-COMPULSIVE PD

William was a 37-year-old accountant, valued by his firm for precision and commitment. However, he often worked overtime to make sure he had done his job properly. This led to conflict with his wife, who found him emotionally unavailable, both to herself and to the children. The marriage was in a

fragile state, but William could not see why. He made a good living, did not drink, and had never been unfaithful. He saw his wife as making unreasonable demands for attention that interfered with his personal space.

This is a typical example of obsessive—compulsive PD (OCPD). This category is another example of a pure trait disturbance that fades imperceptibly into normality. In terms of the five-factor model, it mainly describes high Conscientiousness (Widiger & Costa, 2013). OCPD was the most frequent PD identified in the National Epidemiologic Survey on Alcohol and Related Conditions Study, even when the data were reanalyzed by Trull et al. (2010). OCPD has attracted little other research, except for the CLPS (Gunderson et al., 2011), which showed that it improves with time. Its heritability is .78 (Torgersen et al., 2000). There is no specific protocol for treating patients with OCPD, and no clinical trials of any method of treatment. Because of their emotional rigidity, these patients do not always do well in standard psychotherapy, so one might consider developing specific protocols for OCPD, as well as conducting research on the efficacy of group or couple therapy.

## PD, UNSPECIFIED

PD, not otherwise specified (now called PD, unspecified) is the most common PD clinicians see (Coccaro, Nayyer, & McCloskey, 2012; Verheul, Bartak, & Widiger, 2007). Zimmerman et al. (2005) found it in 14.1% of their sample. Any patient who meets overall criteria for a PD but does not meet criteria for any specific category will qualify for this category. Yet the diagnosis is not always made because the clinician must carefully follow the rules set down in *DSM*–5.

The question is whether this is a meaningful entity or a wastebasket. Patients meeting criteria for a PD can still be very different from each other. Putting them in the same category says more about problems in the *DSM* system than it does about clinical characteristics. The one common factor is that, like other PD patients, they suffer from psychosocial dysfunction, although symptoms are less severe than they are in specific PDs (Wilberg, Hummelen, Pedersen, & Karterud, 2008).

## OTHER PERSONALITY DISORDERS

The 10 categories of PD, retained in *DSM*–5, generally do a poor job of describing the full range of psychopathology described by the PD construct as a whole. They are traditional labels that do not correspond to the factors identified by trait psychology. One of the innovations suggested in the model in Section III of *DSM*–5 was to use trait dimensional profiles to classify these cases. This would have helped clinicians, and I hope this useful idea will eventually be revived.

# Psychopharmacology

This chapter reviews the evidence for the usefulness of medication for patients with personality disorders (PDs). By and large, there is no such evidence. Most patients with PD would be better off taking no medication at all. (The main exception is for managing symptoms such as severe insomnia.) Thus, clinical psychologists are not "missing" anything if they fail to obtain psychiatric consultation on their patients with PDs. Furthermore, because psychiatrists, once consulted, inevitably suggest a prescription, even when it is not evidence based, these procedures are likely to do patients a disservice. This is a sad story, in which my own profession of psychiatry has gone badly wrong, but as with any clinical error, it is important to understand why it has happened.

# WHY DRUGS ARE PRESCRIBED FOR PATIENTS WITH PDs

Clinical practice in mental health is more and more dominated by psychopharmacology. Clinical psychologists do not prescribe drugs, but many of their patients receive them anyway. This is partly because physicians believe in their efficacy and partly because patients themselves, influenced by advertising and the media, insist on them. There is also pressure from insurance policies, which reward rapid assessments and pharmacological treatment methods.

I have written (Paris, 2010c) a detailed critique of the evidence for "aggressive" psychopharmacology. Medication is essential for severe mental disorders (schizophrenia, bipolar illness, melancholic depression). However, many drugs on the market are vastly overprescribed, and not just for patients with PDs. They are routinely given to patients with common mental disorders (anxiety, depression) who may receive only marginal benefit or no benefit from them at all.

A majority of the prescriptions for psychiatric drugs are now being written by primary care physicians. Seven percent of all patient visits to a family doctor lead to treatment with one or more of these agents (Mojtabai & Olfson, 2011). The frequency of these prescriptions has doubled in part because patients are now being given antidepressants on a long-term basis (Moore et al., 2009) but also because pharmacological intervention has become a knee-jerk reaction for physicians. The prevalence of antidepressant prescriptions now exceeds the prevalence of depression itself (Mojtabai & Olfson, 2011), showing that these drugs are being prescribed for distress rather than for treatment of specific diagnoses. Given the comorbidity between depression and PD, many patients with PD will receive antidepressants, even though their efficacy in this population is doubtful at best (Newton-Howes, Tyrer, & Johnson, 2006).

These observations raise questions about whether patients would benefit if clinical psychologists, who treat so many depressed patients, should gain prescribing privileges. Many in the profession have supported this proposal (McGrath, 2010). However, I would not like to see this happen. My views have nothing to do with defending a medical

#### **PSYCHOPHARMACOLOGY**

guild and everything to do with protecting patients from unnecessary treatment.

Today, 11% of the population is taking antidepressants (Pratt, Brody, & Gu, 2011). If psychologists, who outnumber most other mental health professionals, begin to prescribe these agents, the current rate of use would probably increase even further. Nonmedical health professionals would be as reluctant as physicians are to "take the chance" of failing to write a prescription. McGrath (2010) acknowledged this possibility but downplayed it. To be fair, we do not actually know how this scenario would play out, but the current climate is one in which patients expect pharmacological intervention. I worry that psychologists, once allowed to prescribe, would be under enormous pressure to do so, at much the same rate as physicians. They would also make more money, while the problem of overdiagnosis and overprescription could become even worse than it is today.

Another concern is that psychiatrists' sharply declining use of psychotherapy (Mojtabai & Olfson, 2008) would spread to clinical psychology. I am concerned that the most skilled therapists in the mental health system will stop doing what they do best and end up carrying out the same mindless treatment that currently afflicts psychiatry. This would constitute a serious threat to public health. It is already the case that clinical psychologists feel required to obtain a medical consultation to protect themselves against any criticism that antidepressants have not been prescribed. When patients are not doing well, referrals to physicians become even more common. If psychologists prescribed, it would be difficult for them to resist patients' demands for drugs, and they might well be seduced by the increased income that is possible when professionals see more patients for briefer appointments rather than spend an hour conducting a psychotherapy session.

Thus, the problem with the overuse of psychopharmacology does not depend on professional discipline. It is rooted in a widespread belief that mentally troubled people cannot be treated successfully without drugs. Although there is no doubt that pharmacological agents are essential for patients with psychoses and severe depression, their efficacy for common

mental disorders, such as mild to moderate depression, is often hardly better than a placebo (Kirsch et al., 2008).

The belief that psychologically troubled patients must receive pharmacological treatment has become part of both our medical culture and the wider culture, replacing the older idea that everybody should be in psychotherapy. When psychologists ask for consultations on their patients from psychiatrists because they are afraid of missing an indication for drugs, the result is guaranteed. A consultant feels required to suggest something. If the first agent tried is not effective, patients end up trying at least one more drug, if not several more, and often in combination. Yet as shown by effectiveness research (Valenstein, 2006), the benefits of antidepressants have strikingly diminished returns when physicians switch from one drug to another or carry out augmentation strategies with additional drugs.

PDs reflect complex problems and have effects on work and interpersonal relationships. Reducing them to symptoms, especially mood symptoms, almost always misses the point. So how do psychiatrists, who should know better, get this so wrong? By invoking the concept of "comorbidity," they are not treating the patient as a whole but aiming only for symptom relief. Unfortunately, this often means treating every symptom with a different drug. So beware of psychiatrists who speak of comorbidity; they already have pen in hand waiting to write a prescription, whether for depression, anxiety, or putative attention-deficit/hyperactivity disorder. This unhappy scenario plays out all too often in patients with PDs.

## PSYCHOPHARMACOLOGY FOR PDs

The evidence that psychopharmacological agents are effective in the treatment of PDs is weak. Most of the research has concerned borderline PD (BPD), and there is a Cochrane report summarizing the evidence (Stoffers et al., 2010). The best that can be said is that drugs that are sedating have a temporary calming effect on agitation, but patients with BPD have comorbid symptoms of depression, mood instability, and impulsivity, and thus the temptation to prescribe agents targeted for those problems often proves irresistible. That aggressive pharmacological treatment and polypharmacy

#### **PSYCHOPHARMACOLOGY**

have become the rule in BPD is well documented (Zanarini, Frankenburg, Khera, & Bleichmar, 2001). The situation has only gotten worse since the article by Zanarini et al. was published in 2001.

Most of my colleagues continue to prescribe in this way, even those who have done research on PDs and should know better. When I question them on the subject, they respond that the patients they see are too sick for psychotherapy. In other words, they feel they have to do *something*, and who knows whether another drug might not work? Moreover, experts in PD research are not always skilled in psychological treatments and may be reluctant to obtain access to therapists who are.

I am particularly critical of the practice of polypharmacy, which is not at all evidence based; no clinical trials on drug combinations for PDs have been conducted. To prescribe five drugs to one patient suggests an absence of judgment and understanding. Mental illness is not a set of symptoms, each of which needs a separate magic bullet. Actually, providing a different drug for each symptom is not a new phenomenon in medical practice. This is an approach that the famous physician William Osler (1898) described over a century ago as "shotgun medicine."

I am painting a grim picture, but it corresponds to the way psychiatry is currently being practiced. I spend much of my own time in a treatment team for PDs getting patients *off* medication and ensuring they receive evidence-based psychotherapy. By and large, PD patients are better if they take nothing at all. I would now like to examine the empirical evidence behind these conclusions.

## Antidepressants

Even in depressions not complicated by a PD, one sees only a small advantage of drug over placebo, with effects most marked in severe depressions (Kirsch et al., 2008). Yet because patients with BPD are often comorbid for dysthymia or major depression (Zanarini et al., 1998), they are frequently offered antidepressants. Consistent evidence shows that the presence of *any* PD makes drug treatment of depression much less effective (Newton-Howes et al., 2006). This is the real meaning of *treatment-resistant depression*. In a

Cochrane review, Stoffers et al. (2010) found no clinical trials supporting the use of antidepressants in BPD. The National Institute of Clinical Excellence guidelines in the United Kingdom did not recommend them at all (Kendall, Burbeck, & Bateman, 2010). Fortunately, these agents do little harm. So when I reduce previously prescribed medication, I usually leave antidepressants for last.

Tricyclics were the first drugs to be widely used for depression. The problem with prescribing them for patients with PD who are chronically suicidal is that taking a week's supply can be fatal. Moreover, tricyclics have anticholinergic side effect profiles that often lead to noncompliance. Similar concerns about the danger of overdose arise for monoamine oxidase inhibitors, and patients may fail to follow the dietary restrictions that are required to avoid hypertension. These drugs are rarely used today for PD patients.

Because selective serotonin reuptake inhibitors (SSRIs) are much safer, they have been widely used, especially in BPD. However clinical trials suggest that their efficacy is limited. By and large, SSRIs have more consistent effects on aggression and anger than on depression or mood swings (Kendall et al., 2010). Antidepressants temporarily "take the edge off" and are also sedating enough to reduce anxiety. That can be useful, but one never sees remission of the PD.

## **Mood Stabilizers**

BPD is associated with marked affective instability (Koenigsberg, 2010) or emotion dysregulation (Linehan, 1993). Patients show rapid mood swings and do not easily calm down once upset. The idea that affective instability lies in the bipolar spectrum is not supported by research (Paris, 2012), nor does research support the idea that BPD is a form of bipolarity (Paris, Gunderson, & Weinberg, 2007). This helps explain why mood stabilizers—lithium and antiepileptic agents—do not produce a remission in BPD, as they often do in bipolar I and bipolar II disorders. The idea that BPD patients with emotion dysregulation benefit from these agents is not based on evidence but derives from the misleading term *mood stabilizer*. However,

#### **PSYCHOPHARMACOLOGY**

as with SSRIs, mood stabilizers that are sedating often yield some reduction in anger and aggression. Claims that mood stabilizers have a larger effect on impulsivity than either SSRIs or antipsychotics (Ingenhoven, Lafay, Rinne, Passchier, & Duivenvoorden, 2010) are misleading, given that the meta-analyses on which these conclusions are based include randomized controlled trials (RCTs) with small samples and dubious methodologies.

It is not surprising that mood stabilizers have, at best, only a partial effect in BPD. If affective instability arises from entirely different pathological mechanisms from bipolar disorder, then these symptoms may require entirely different drugs that have yet to be developed. If and when they are invented, I will be happy to prescribe them.

## Antipsychotics

Antipsychotics, usually in low doses, have long been used in BPD. However typical neuroleptics have many troubling side effects, particularly when used long-term, and clinical guidelines have specifically warned against using them in PDs (Kendall, Burbeck, & Bateman, 2010). The development of atypical neuroleptics, which are better tolerated, has made the prescription of antipsychotics much more common. However, these agents have a serious side effect, one that develops after patients are under treatment for several months: a metabolic syndrome associated with weight gain and diabetes (Newcomer & Haupt, 2006). This is a good reason for caution in prescribing these agents and for quickly discontinuing them once administered.

Atypical antipsychotics are effective for the psychotic symptoms that accompany BPD (Rosenbluth & Sinyor, 2012). Yet once again, their main effects on BPD are a reduction in impulsivity and anger, without remission of the disorder. Nonetheless, short-term use can be of benefit. The problem is that once the patient is on an antipsychotic, physicians are afraid to discontinue it, but the longer patients take antipsychotics, they more side effects they will have. Add to that the high doses of atypical antipsychotics that physicians sometimes prescribe when patients do not have an initial response.

Most clinical trials thus far have been conducted on olanzapine, all of which were sponsored by Eli Lilly (the company that manufactures this agent). Yet perhaps the most common drug used in practice these days is quetiapine, which thus far has only one short-term clinical trial in BPD (Black et al., 2014). This agent may be preferred because it is a little less likely to cause weight gain and metabolic syndrome. Another popular alternative is aripiprazole, which has never been tested systematically in BPD.

Overall, the effects of neuroleptics are occasionally useful but usually marginal. Given their side effects, they should only be used on a short-term basis.

## Polypharmacy

When patients show a short-term improvement after being prescribed a drug but the improvement does not last, there can be several possible explanations. One is that the original response was a placebo effect that had to be temporary. Another is that a drug can be sedating in a state of great agitation but did not affect any of the other symptoms of a disorder. Then this scenario can lead to the prescription of additional drugs, usually by adding a new agent (without subtracting what the patient is already taking). This sequence helps explain why so many BPD patients are on a polypharmacy regime of four or five drugs (Zanarini et al., 2001), with at least one agent from each major group.

For some time there have been published algorithms for drug treatment of various conditions, suggesting a sequence of prescriptions, each of which would target different symptoms. The American Psychiatric Association (2001) guidelines for BPD, now out of date, were deeply flawed from the beginning because they adopted this approach, doing real damage to practice by offering algorithms for pharmacotherapy that were not based on RCTs, which carried a heavy burden of side effects, and guaranteed polypharmacy. The more conservative conclusions of the National Institute for Clinical Excellence (Kendall et al., 2010) in the United Kingdom recommended that drugs should be prescribed with great caution in BPD.

#### **PSYCHOPHARMACOLOGY**

In summary, drugs for BPD are nonspecific stopgaps with limited value that could eventually be replaced by better and more precise alternatives. We do not have agents that specifically target the traits that underlie BPD: impulsivity and affective instability. If we had an effective pharmacological treatment for these trait dimensions, psychiatry would be revolutionized.

There is even less evidence for prescribing drugs for other PDs. However, there may be a benefit in using low-dose antipsychotics for schizotypal PD (Chemerinski, Triebwasser, Roussos, & Siever, 2013).

What is discouraging about current practice is that drugs are being given to patients with PDs only because psychotherapy is not available. If these agents are used, we need to apply the same principles for evaluating them as for any medical illness—that is, a number of RCTs that is sufficient to conduct meta-analyses that yield replicable findings. These are the standards of reports by the Cochrane group, the most reliable and comprehensive source of information about the efficacy of pharmacological and psychotherapeutic treatments. Cochrane's conclusions about efficacy in BPD (Stoffers et al., 2010) were typically cautious and tentative. The existing literature, marked by small samples, short follow-up periods, and lack of replication, does not meet accepted standards. Drugs for PDs do not have any effect on personality pathology, and most have potentially serious somatic side effects. We need to consider alternatives—that is, the forms of psychotherapy that have been specifically developed for PDs. That is the subject of the next chapter.

## 10

# Psychotherapies

Psychotherapy for personality disorders (PDs), like any other form of treatment, should be evidence based. For many years, if you wanted to read about treatment, the only choices you had were to read books describing the clinical experience of an "expert." You could also go to workshops to learn about these ideas. Recommendations were not based on empirical evidence, however, because there wasn't any.

Almost all research on the treatment of PDs has studied patients meeting criteria for borderline PD (BPD), which is also the condition that most interests clinicians. Starting with the seminal work of Linehan (1993), a number of innovative methods of treatment have been tested in clinical trials and shown to be effective. There are now half a dozen therapies for patients with BPD, each described by an acronym. I am not convinced, however, that they work in different ways and have different effects.

Three caveats should be considered. First, even if one follows a tested method of psychotherapy, many decisions still need to be made that derive from experience and skill. Second, research on the therapy of BPD does not have the heft of literature on other mental disorders, such as depression or anxiety. Third, there is no evidence that any one method that has been tested is better than any other. Any therapy that is well planned will be better than unstructured treatment as usual (TAU) because it provides patients with external structures that make up for their inner chaos.

A further limitation, one that applies to any psychotherapy, is that not every patient can be expected to benefit from treatment. Even in the most seriously ill, however, rehabilitation can have partial effects. Yet some are sicker than others, and those who do best usually have "ego strengths"—a job, an intimate relationship, or both. These areas of positive functioning provide patients with a base on which to build skills in other areas. Although there are always surprises and some patients who are seriously ill may recover, the observation that better functioning is predictive of outcome in psychotherapy is a well-known and consistent finding in research (Bohart & Greaves-Wade, 2013). If you are treating patients who have no job, no relationship, and no life to speak of, what is there to work on? If, on the other hand, patients have a life, then therapy benefits from a laboratory setting in which people can practice what they learn in treatment sessions.

Although specialized therapies may not be different from each other, not all psychotherapies applied in practice are equal. What researchers call TAU tends to be a mess in which patients talk about their problems to a sympathetic professional but are not given specific guidance in overcoming dysfunctional emotions, thoughts, and behaviors. This is why clinical trials always find that specific methods do better than TAU: It is not hard to do better. Yet when comparisons are made between two well-structured approaches, differences usually disappear (McMain et al., 2009; Zanarini, 2009). Patients need planned and structured forms of therapy, but the brand name may make no difference.

#### **PSYCHOTHERAPIES**

## SPECIFIC METHODS OF PSYCHOTHERAPY FOR BPD

## **Dialectical Behavior Therapy**

Dialectical behavior therapy (DBT) was the first evidence-based treatment for BPD. Developed by Marsha Linehan (1993), this was the first psychotherapy for BPD to undergo successful clinical trials (Linehan, Armstrong, Suarez, Allmon, & Heard, 1991; Linehan et al., 2006). The introduction of DBT was a turning point in the treatment of the disorder, and its principles lie at the core of all successful therapy in this population. Here, at last, was a practical approach that targeted the key traits and symptoms of BPD.

Today, DBT remains the leading evidence-based method of therapy for patients with BPD. It is an adaptation of cognitive behavioral therapy (CBT), combined with interventions common to other approaches, but specifically designed to target the emotion dysregulation that characterizes BPD, and to reduce impulsive behaviors. It applies chain analysis to incidents leading to self-injury and overdoses—that is, showing patients what emotions lead up to impulsive behaviors and teaching them alternative ways of handling dysphoric emotions. DBT also emphasizes empathic responses to distress that provide validation for the inner experience of patients. The program consists of weekly individual therapy, group psychoeducation, telephone availability for coaching, as well as support through consultation for therapists undertaking these procedures. The method is an eclectic mix of behavior therapy, CBT, mindfulness based on Zen Buddhism, and original ideas such as radical acceptance (Linehan, 1993). These techniques have been described in some detail (Linehan, 2014; Linehan & Koerner, 2012).

The first published trial (Linehan et al., 1991) compared 1 year of DBT with TAU and found DBT to be superior, especially in regard to reductions in self-harm, overdose, and hospitalization. The question was whether it was too easy to do better than TAU. For this reason, Linehan et al. (2006) conducted a second clinical trial in which the comparison group was "treatment by community experts"—therapists who identified themselves as interested in BPD and experienced in its treatment. The results again

favored DBT, with reductions in overdoses and subsequent hospitalizations within a year, although this time there were no differences between the groups in the frequency of self-harm. Replication studies in other centers produced similar results, albeit with higher rates of attrition (Linehan & Koerner, 2012). A meta-analysis (Kliem, Kröger, & Kosfelder, 2010) supported the conclusion that DBT is an effective and specific method that is superior to traditional ways of treating BPD patients.

Although several specific methods of therapy designed for BPD symptoms have been supported by randomized controlled trials (Paris, 2010b), the strongest evidence supports DBT. The method is a clinical application of psychological research on emotion regulation (Gross, 2013). The dysregulation in BPD leads to unstable emotions that are abnormal responses to interpersonal conflict (Koenigsberg, 2010). That conclusion has been confirmed by studies of BPD patients using ecological momentary assessment, a technology that allows researchers to track emotional instability more closely by immediate recording of affective and behavioral responses to life events (Russell et al., 2007; Trull et al., 2008). In DBT, patients are taught better ways of calming down, during and after emotional storms, which then reduces the frequency of self-harm and overdoses.

There are some important unanswered questions about DBT. Although the original cohort received therapy 20 years ago, it has never been followed up, so we do not know whether treated samples maintain their gains and continue to improve beyond a 1-year posttreatment follow-up. Also, given the resources required to conduct DBT, it needs to be determined whether this complex program can be dismantled or streamlined for greater clinical impact. One report found that a 6-month version of the therapy can also be effective (Stanley, Brodsky, Nelson, & Dulit, 2007). A treatment lasting for a year (and often more) becomes quickly inaccessible as waiting lists grow and most patients and their families cannot afford the expense.

Finally, there is the question of whether DBT is a uniquely efficacious treatment for BPD or whether other well-structured approaches can produce the same results. To address this issue, McMain et al. (2009)

#### **PSYCHOTHERAPIES**

administered DBT for 1 year, with random assignment to a comparison condition called *general psychiatric management*, a manualized version of the American Psychiatric Association (2001) guidelines for the treatment of BPD. The results of this comparative trial found no differences between the groups in overdoses, hospitalization, or self-harm. This negative finding had important clinical implications. It suggests that although DBT is better than most treatments, it can be matched by other therapies that are designed for this population and that are equally well-structured. Further, because results were good in both groups, the treatment package used for the comparison has now been studied on its own, under the name of *good psychiatric management* (Gunderson & Links, 2014).

A key question about DBT is whether the results of the treatment are specific to the method or to the structure. By and large, psychotherapy research supports a common factors model in which all well-structured treatments yield similar outcomes (Wampold, 2001). The positive results of DBT could be due to its high level of structure rather than to its specific interventions. This supposition was supported by the study by McMain et al. (2009). So although DBT is clearly better than TAU and somewhat better than treatment by therapists with experience in treating BPD, it is not necessarily better than a well-thought-out program of clinical management.

The popularity of DBT depends on its comprehensiveness, as well as on its commitment to conducting research to demonstrate its efficacy. DBT is not the only evidence-based therapy on the market, but it is the only method that has been tested in multiple clinical trials outside the center where it was developed, showing that its efficacy cannot be accounted for by allegiance effects. The ideas behind DBT are fundamental for any therapist seeing patients with BPD. It is not a narrowly focused form of treatment that only deals with cognitive schemata; it also provides validating responses to current emotional upsets and offers education about emotion regulation.

However, there is a serious problem with DBT: Its expense makes it inaccessible. This is mainly because of the length of therapy. DBT has been tested for a year, but even that length of time is beyond most insurance

policies or the financial resources of most families. Of even more concern, Linehan (1993) suggested that even this lengthy period may be only the first phase of a treatment that could go on for several years. I am reminded of the story of psychoanalysis in which inevitably incomplete results led to an interminable course of therapy. DBT is effective but is accessible only to those who can pay for it. Even if it were properly insured, its length would still make access a problem: Clinics offering the treatment, even for 12 months, often have extensive waiting lists. It is important to shorten DBT or to make it intermittent (or do both). This is the only way to provide service to more patients.

We all owe a debt to Marsha Linehan. I have learned an enormous amount from her and have applied her principles in all the clinics I lead that treat BPD. Moreover, Linehan's recent public statements, acknowledging that she herself once suffered from BPD but recovered, were courageous and have done a great deal to reduce the stigma associated with this disorder. Nonetheless, treatment for BPD suffers from the perception that DBT is the only brand that works. Therapists should not feel badly if they are not in a position to provide DBT in a formal way or to refer patients to a DBT clinic. In the next chapter, I show that its principles can be incorporated into normal clinical practice. In my view, brand names are bad for therapy. As cognitive theory evolves, it has become a more general term for what might be called simply *psychotherapy* (Beck & Haigh, 2014). Livesley (2012) recommended that DBT give up its brand name and incorporate its best ideas into a general model of treatment for BPD.

## Other Evidence-Based Psychotherapies for BPD

Although other methods have been devised, they do not differ from DBT in any essential way (Paris, in press). We now examine those that have undergone clinical trials.

Mentalization-based treatment (MBT; Bateman & Fonagy, 2006) is rooted in attachment theory, but the method also has a strong cognitive component. Its assumption is that BPD patients have trouble recognizing emotions (their own and those of other people), that is, *mentalization*.

#### **PSYCHOTHERAPIES**

MBT teaches patients how to do that better. Like most effective programs, it uses a combination of group and individual therapy. Although developed by psychoanalysts, it uses a number of cognitive methods similar to DBT in that patients are taught to recognize their emotions, learn how to tolerate them, and manage them in more adaptive ways.

MBT was first tested in a randomized controlled trial (RCT) in a day program lasting 18 months (Bateman & Fonagy, 2001) and found to be superior to TAU. A second study in a larger sample of outpatients given 18 months of treatment found a decline of both self-reported symptoms and clinically significant problems, including suicide attempts and hospitalization (Bateman & Fonagy, 2009). This is the only method for which researchers have followed up a cohort for 8 years to determine if the effects of treatment remain stable, which turned out to be the case (Bateman & Fonagy, 2004).

MBT needs successful clinical trials in centers outside the hospital where it originated. Thus far, the only attempt at replication outside the United Kingdom reported few differences from standard therapy (Jørgensen et al., 2013). However, Bateman and Fonagy (2008) do not consider MBT as a "one-and-only" approach but encourage mental health workers to learn its principles and then apply them in their own clinical settings, without necessarily following a strict protocol. One can only applaud such openmindedness and flexibility. Finally, Bateman and Fonagy (2008) have stated that the results of their research are not specific to their method but support any structured approach to psychotherapy. This may be the most consistent finding in this literature (Choi-Kain & Gunderson, 2008).

Transference-focused psychotherapy (TFP; Clarkin, Levy, Lenzenweger, & Kernberg, 2007) is based on the theories of the psychoanalyst Otto Kernberg. It differs from other methods in that its focus is on distortions between therapist and patient in the session, used to illustrate interpersonal problems elsewhere in the patient's life. It has thus far undergone two clinical trials, one comparing it with DBT, with only minimal differences (Clarkin, Levy, Lenzenweger, & Kernberg, 2007), and one comparing it with TAU, to which it was superior (Doering et al., 2010). TFP aims to generalize what happens in therapy to outside relationships.

Given the long record of failure for psychodynamic therapy in BPD, one might consider this approach with caution. Nonetheless, at this point TFP has about as much support as MBT. It shares the advantage of being structured and well thought out.

Cognitive analytic therapy is based on similar concepts and can be considered as another psychodynamic—cognitive hybrid. It applies object relations theory to establish a firmer sense of self in patients. It has been tested in a population of adolescents (Chanen & McCutcheon, 2013), where it was effective, albeit not superior, to a manualized version of "good clinical care."

Schema-focused therapy (SFT; Young, Klosko, & Weishaar, 2003) is another mixture of cognitive and psychodynamic approaches that aims to modify how patients think about their world (i.e., cognitive schemata), but it also focuses on the distorting effects of negative childhood experiences. It has undergone one clinical trial comparing it with transference-focused psychotherapy (Giesen-Bloo et al., 2006), with only minor differences in outcome and one trial in which it was superior to TAU (Bamelis, Evers, Spinhoven, & Arntz, 2014). The problem with schema-focused therapy is that it is designed to last for 3 years, making it even more inaccessible than DBT.

Standard CBT has been tested in a study conducted in the United Kingdom (Davidson, Tyrer, Norrie, Palmer, & Tyrer, 2010). After an average of 26 sessions, BPD patients did better with cognitive therapy than with TAU. It also seems likely that CBT for BPD is now being conducted on broader, more flexible principles. When a Cochrane review (Stoffers, Völlm, et al., 2012) concluded that the data for cognitive therapy were "promising," they were not thinking of standard CBT. Linehan had developed DBT because of her impression that standard CBT was not effective for BPD. Yet a large RCT (Davidson et al., 2010) found manualized CBT, modified to target PD symptoms, was superior to TAU for the treatment of recurrent deliberate self-harm.

It is important to know that therapy lasting for a few months can be effective. The evidence for this conclusion was recently reviewed by Davidson and Tran (2014). What was most striking about these findings

#### **PSYCHOTHERAPIES**

was that the mean length of treatment was 16 sessions. This suggests that BPD might be treated more rapidly, and less expensively, than by treatments designed to continue for a year or two. Perhaps the most chronic and severe patients with BPD require several years of therapy, but it makes no sense to make a long duration the standard of care.

Systems training for emotional predictability and problem solving (STEPPS; Blum et al., 2008) is a brief and practical program that closely resembles DBT in its focus on emotion regulation skills. It is designed to supplement TAU, particularly in settings where specialized individual therapies are not available. STEPPS, based on psychoeducation in groups, has been supported by clinical trials, with a 1-year follow-up (Blum et al., 2008).

STEPPS is a short-term intervention with psychoeducation conducted in groups, designed to supplement standard psychotherapy or management conducted elsewhere. It is particularly suitable for populations living in regions where specialized treatment is not available. It is inexpensive and offers ready accessibility. STEPPS has been subjected to a successful clinical trial in BPD (Blum et al., 2008), with one replication (Bos, van der Wel, Appelo, & Verbraak, 2010). It has also undergone one test in the treatment of antisocial PD (Black, Gunter, Loveless, Allen, & Sieleni, 2010), although one cannot conclude that this makes antisocial PD as treatable as BPD.

#### COMMON FACTORS IN THE TREATMENT OF BPD

Although these individual approaches each have useful ideas and techniques, they work through common mechanisms (Paris, in press). A vast literature shows that the effective factors for outcome in any form of psychotherapy are common rather than specific (Wampold, 2001). Nor is it necessarily true that patients with BPD can only be seen in specialized clinics; most benefit from what Gunderson and Links (2014) termed *good psychiatric management*. The ideas behind effective therapy are spreading to the wider therapeutic community, and interventions (e.g., teaching emotion regulation) are becoming part of the armamentarium of therapists of all persuasions.

It is unfortunate that psychotherapy as a field continues to be defined by competing methods, many of which use a three-letter acronym. It is even more unfortunate that clinicians define themselves as practitioners of any single method. Research can help us get beyond these unnecessary divisions.

Thus, I agree with Livesley (2012), who suggests that psychotherapy should be evidence based, not acronym based. Even if some interventions are partly specific to BPD, we need a single model of therapy to make use of the best ideas for all sources. To have multiple methods competing for market share may be good for book sales, but it is not the way to develop evidence-based practice.

We do not need so many forms of psychotherapy, most of which resemble each other more in practice than in theory. If therapies based on so many different ideas and using many different techniques can produce the same results, they must have a lot in common. One of the main ingredients is structure. Traditional therapies for PD failed because they rely on unstructured techniques that leave patients adrift. These are not patients who get better just by being heard and supported. People with BPD also need specific instruction about emotion regulation, control of impulsivity, and life skills that can be used to find a job and build a social network.

Although different methods seem to target different aspects of PD, the failure of comparative trials to find large differences in outcome also suggests that common factors are of crucial importance. Again, consider the large body of research supporting the view that common factors (also called nonspecific factors) are the best predictors of results in all forms of psychotherapy (Wampold, 2001). By and large, when different forms of therapy are compared head to head, researchers almost always find equivalent results. The most important common factors are a strong working alliance, empathy, and a practical problem-solving approach to life problems (Baldwin & Imel, 2013; Crits-Christoph, Gibbons, & Mukherjee, 2013).

With a complex and challenging disorder like BPD, psychotherapy needs to maximize these mechanisms and find ways to make them more specific. The best-validated methods offer a defined structure, focus on the regulation of emotions, and encourage the solution of interpersonal

#### **PSYCHOTHERAPIES**

problems through self-observation. Empathy and validation are essential elements of any therapy but are particularly important for BPD patients, many of whom are sensitive to the slightest hint of invalidation (Kohut, 1970; Linehan, 1993). In other words, these are patients who can easily feel that their emotions are being dismissed. They will not listen to anything else you have to say unless they perceive that their feelings are accepted.

Self-observation is a skill that therapists need to teach all their patients. When one learns to know feelings better (and not be derailed by them), one can stand aside from emotional crises or even begin to think about alternative solutions to problems. Clinicians who provide treatment following these principles do not necessarily need to refer patients to specialized programs.

#### IMPORTANCE OF STRUCTURE IN PSYCHOTHERAPY

Psychotherapy is the backbone of treatment in BPD, but clinicians in the past were not trained to apply structure to treatment sessions. That is probably why open-ended therapies have been associated with large dropout rates (Skodol, Buckley, & Charles, 1983). Moreover, therapies that focus too much on the past have a way of encouraging patients to regress.

The key to recovery from a PD is to "get a life." That usually means finding a job or going back to school to prepare for a job. Without a social role, recovery from a PD is less likely (Zanarini et al., 2012). Unfortunately, some of our patients make the mistake of trying to solve their life problems through an intimate relationship that gives the illusion, for a time, of unconditional love. In the absence of work, that strategy only makes them dependent on another person, seriously impeding self-mastery.

Finally, because PDs usually improve with time, therapy aims to hasten naturalistic recovery. Because patients can get better on their own, determining whether change is the result of a specific intervention requires testing through RCTs. Thus far, these trials have provided strong support for a few psychotherapy methods and tentative support for others.

In summary, even if a well-structured approach works well for most patients, generalized methods might be enough, and patients with BPD

have done much better since DBT and other methods specific to the disorder were developed. We await the day when effective packages of this kind can be developed for other PDs.

### WHAT WORKS, WHAT DOESN'T WORK, AND WHY

Because of their prominent mood symptoms, patients with BPD are often put on medication and can end up being prescribed four or five different drugs (Zanarini, Frankenburg, Khera, & Bleichmar, 2001). The review in Chapter 9 showed that these practices are not evidence based. The role of pharmacology in BPD treatment is limited and is most effective for short-term management of insomnia. Because most medications in current use (antipsychotics, antidepressants, and mood stabilizers) are sedating, they can "take the edge off" BPD symptoms through nonspecific effects on impulsivity. However, as shown by the most recent Cochrane report (Stoffers, Völlm, et al., 2012), none of these agents have specific effects on BPD itself. Most patients can be managed with minimal medication or with no medication at all.

Clinical psychologists treating BPD patients should therefore be cautious about obtaining psychopharmacological consults to "cover themselves." I understand why this happens; these are difficult, scary cases. However, when you ask for a consult with an MD, your patient may be put on an aggressive drug regime. If you read the literature, with its conservative conclusions, it will become clear that although drugs are palliative in the short term, no pharmacological agent produces remission in BPD. If you need consultation on difficult cases, I suggest you choose clinicians with expertise in the psychotherapy of BPD.

It must be acknowledged that psychotherapy for these patients has not always had a good reputation. More than 75 years ago, Stern (1938) described BPD as treatment-resistant (i.e., it didn't respond to psychoanalysis). Ever since, therapists have struggled with the obstacles the disorder presents. It is not easy to manage people who don't follow your advice, don't always come to appointments, and frequently threaten suicide.

#### **PSYCHOTHERAPIES**

Even so, many problems can also be understood as artifacts of well-meaning but insufficiently structured therapy. TAU is often the comparison point in research studies, but it might be better described as "the usual mess." Patients with BPD don't fit well into normal practice, in either clinics or offices. They need therapists trained to provide more specific interventions.

Yet even when therapy is based on a theory, it can still falter. In the past, patients with BPD were offered regressive psychoanalytic approaches that were unproductive or counterproductive. In BPD, therapy fails when too much time is spent talking about the past. Of course, if childhood was marked by trauma, life histories need to be validated and understood, but patients need to move on and deal with their current problems in relationships and work.

Standard methods of behavioral therapy and CBT may also run into difficulties in this population. Linehan (1993) developed DBT because standard CBT did not seem to be effective for treatment of BPD. For example, patients with BPD are not always willing to do the homework that CBT requires. Linehan's discovery was that therapy works best when offering specific strategies for emotion regulation. This was the great breakthrough that has made BPD a treatable disorder.

Moreover, the "supportive" techniques used in TAU (sessions that review the week and provide nonspecific encouragement) are not evidence based. Research on therapy for BPD shows that almost any specific method is better than TAU, underlining the limitations inherent in the reality of all these "usual" clinical practices. Yet, as more therapists become aware of more specific methods, TAU itself may be changing for the better.

Being an effective therapist for these patients may not depend that much on your theory about BPD. It is more important to understand people whose communication style can be difficult and problematic, to be comfortable with knowing that you cannot prevent suicide. If you want to treat BPD, the first requirement is a thick skin.

Yet psychotherapies designed for BPD have a stronger evidence base than any form of pharmacotherapy. As noted earlier, medications have never been shown to lead to the sustained remissions documented for

psychological treatments. However, not "any old" psychotherapy will do. In the past, mistaken methods leading to poor results have given therapy for patients with BPD a difficult reputation.

Although no research has specifically examined TAU, it is not difficult to see why it doesn't work. Patients come to their sessions and tell stories about stressful events that have occurred over the week. Therapists validate feelings, but by itself, that does little for patients who misunderstand and distort their interpersonal environment. The danger is that patients will perceive that their therapists agree with them—that other people are to blame and that they are victims.

Empathy has to be linked to tactful confrontations to help patients learn new ways of understanding and dealing with problems, what Bateman and Fonagy (2006) called the *capacity to mentalize* (similar to the concept of mindfulness). Thus, using what Linehan (1993) called a *dialectical* approach, one must validate as one teaches new skills. The absence of such an approach is why supportive therapy has limited value. The aim must be to have a strong enough alliance with patients that they are willing to see their problems in a different light.

These principles help us to understand why classical psychodynamic therapy was often ineffective for BPD. Patients who cannot mentalize and who are constantly in the throes of emotion dysregulation cannot make use of procedures such as free association with a relatively silent therapist who only intervenes to make "interpretations." Moreover, when therapy focuses on the past rather than the present, patients are more likely to be mired down in their grievances than to move on (this is what Linehan meant by *radical acceptance*). People move on more easily when they feel understood, independent of a therapist's theories (Strupp, Fox, & Lesser, 1969). Reexperiencing traumatic events from childhood can be particularly counterproductive. A neuroimaging study helps to show why. Koenigsberg (2010) found that patients with BPD do not habituate to stressful thoughts but become increasingly activated and disturbed. Thus, therapies that focus on trauma produce regression and increase symptom levels.

In summary, therapies that are present oriented, have a strong cognitive component, balance acceptance and change, offer a predictable structure,

#### PSYCHOTHERAPIES

and in which therapists are active and engaged are most likely to succeed. We need to place more importance on the present than the past to help patients to get a life. In most cases this means getting a job or an education. It can also mean raising a family. For some people, it may involve hobbies or volunteer work. In whatever form, one must engage with the world to get better. Patients also need to be told that they have to work on getting a life now, not wait for therapy to somehow make doing so easier.

#### INTEGRATED BPD TREATMENT

Although evidence-based treatments for BPD have emerged from specialized treatment programs, these clinics tend to be too expensive or inaccessible. Yet even though these are not the settings where most therapists work, the same principles can be applied to ordinary practice.

An example is the use of group therapy to teach patients behavioral and cognitive skills, which is part of the package offered by the methods that have been most systematically tested. Most clinicians in practice do not carry out this kind of treatment. This is why the STEPPS program was developed: to augment individual therapies conducted by therapists in the community by providing a group setting based on the principles of psychoeducation. Yet because few communities have access to STEPPS, therapists should consider doing more group therapy in their own practices or in group practices. Another example is the use of psychoeducation to teach life skills and emotion management. CBT has been doing this for decades. These methods can also be transferred to the setting of individual therapy.

Another implication of research on BPD treatment is that therapists need to move out of the primarily receptive mode they may have been taught to adopt. There is no contradiction between empathic listening and therapeutic activity.

Psychotherapy for BPD is being held back by the existence of multiple competing methods, each with a three-letter acronym. The results of these methods tend to be overinterpreted by therapists with allegiance to one or another of them. Yet although all well-structured methods are

superior to TAU, none is clearly superior. There should be only one kind of psychotherapy for PD: the one that works. An integrated method would use the best ideas from everyone and put them together into one package (Livesley, 2012).

This conclusion, consistent with research on common factors in all therapies, should be reassuring. Therapists need not be overly concerned that they haven't been trained in the latest method or the latest twist on existing methods. Psychotherapy is placed in a bad light by the endless competition between approaches. In medicine, there is no such thing as a school of treatment specific to any drug; therapeutic agents are used when appropriate and when they complement other interventions. Even so, treatment of BPD cannot be generic but needs to be more specific. For some clinical problems, such as severe substance abuse (W. R. Miller & Rollnick, 2013), new and different methods have been developed. Several of the therapies developed for BPD offer unique interventions that go beyond what clinicians do for most of their patients.

Linehan unlocked a crucial door by placing emphasis on skills for emotion regulation. BPD patients do not recognize their emotions or know how to deal with them, nor do they know how to self-soothe when experiencing difficult feelings. They often do not even know they have had an emotion and move directly to impulsive actions to get rid of a bad or uncomfortable feeling. That is why reviewing the sequence of events before a cut or an overdose is so crucial. Also, even though mindfulness is a difficult technique for most people to learn, even the simplest forms of self-observation can be useful. STEPPS offers a practical method, with down-to-earth pictograms of boiling pots to help people rate the intensity of their emotions. MBT also teaches people to recognize what they feel, but adds an emphasis on the need to recognize what other people are feeling.

Radical acceptance is another important element that is common to all effective therapies. Patients are not encouraged to feel like victims but to come to terms with the past. Most will have had difficult childhoods with adverse events of various kinds. Yet they need to accept the hand that life has dealt them and to accept themselves with all their flaws. Radical acceptance

#### **PSYCHOTHERAPIES**

is similar to the famous motto of Alcoholics Anonymous, which advises people to accept what they cannot change.

Rather like the proverbial tale of the blind men and the elephant, each of the specific methods developed to treat these patients looks at the problem of BPD from a different angle, and all have some degree of validity. One can combine the management of emotion dysregulation emphasized by DBT, the ability to observe feelings emphasized by MBT, and the focus on negative thought patterns that characterizes SFT, together with a limited level of exploration and understanding of life histories. These tools are all part of a broad therapeutic armamentarium, nested in an empathic and practical approach, aimed at maximizing the common factors that produce success.

# APPLYING EVIDENCE-BASED PRACTICE TO PD TREATMENT AS A WHOLE

When specific methods are developed for the other PDs, they will probably follow many of the same principles described for BPD. What is needed is to define trait domains that can be modified by psychological interventions and to develop interventions that can be used to increase interpersonal skills and reduce negative patterns of behavior. For example, a treatment package that had an effective way of modifying grandiosity could open the door to effective treatment of narcissistic PD. Similarly, a package modifying perfectionism would be the key to treating obsessive—compulsive PD. It is known that CBT programs for treating social anxiety have some effect on avoidant PD (Ahmed et al., 2012), but they have not been extensively tested.

In many ways, psychotherapy for PDs is just beginning. BPD, because of its great clinical burden, will continue to take precedence. However, other PDs, common in clinical and community settings, need specific interventions of their own. Inspired by the success of treatment for BPD, such programs are bound to be developed in the coming decades.



## Management

In this chapter I must, of necessity, depart to some extent from the evidence-based perspective of this book. The reason is that practitioners have to make decisions about the details of everyday clinical practice that have not been examined in research. Also, although psychotherapies as a complete package have been tested in clinical trials, they are a mixed bag. Most probably, some interventions are highly therapeutic, and others are merely idiosyncratic. Clinicians who want to adopt an eclectic approach need not follow any procedure with absolute fidelity.

I am not proposing a comprehensive treatment model of my own, nor do I want to develop one. However, although an integrated therapy would eventually need its own clinical trials, it can at least be consistent with available evidence. I want to apply common principles derived from the literature, combining the best ideas from many sources. I must acknowledge that like other specialists in personality disorder (PD) treatment,

some aspects of my approach to management are based on my experience (as well as the experience of working in a clinical team).

This chapter examines some of the problems that trouble therapists of patients with PD and make management difficult and considers principles that could be used to minimize them. The first focus is the tendency of patients to resist change and be in interminable treatment. The second is the frightening chronic suicidality that therapists need to manage in patients with BPD.

# PSYCHOTHERAPY: TERMINABLE AND INTERMINABLE

Patients with PDs are often lonely. This leads them to become overly attached, either to therapists or to the mental health system. They need to understand that recovery requires finding a place in the community and a social role. Otherwise, therapy becomes not a means to an end but an end in itself. Goals for change need to be set and agreed on and used to monitor progress. If rehabilitation is not a patient's goal, then there is no point in undertaking treatment.

Patients with PD are known to be at risk for interminable therapy. This problem arises because their lives are unsatisfying, making them anxiously attached to their therapists, and using the protected treatment setting to compensate for deficient social networks. Patients with borderline PD (BPD) sometimes see the mental health system as a setting where they cannot be rejected. Patients with narcissistic PD, who believe they are interesting, may enjoy treatment but fail to commit themselves to change; in common parlance, they talk the talk, but don't walk the walk. Patients with avoidant PD, who only feel safe with people they know well, may not translate what they learn in therapy to an interpersonal context. Patients with compulsive PD, who maintain emotional control through intellectualization, are better at theorizing than at putting ideas into practice.

Interminability is a problem for all forms of talking therapy, but practitioners need to be aware how personality traits can stand in the way of

effective therapy. This is why we need to make every session count and avoid drift or stalemate. Every therapy must also include, from the first sessions, a plan for termination.

In any treatment, there will always be work left undone, but therapy that goes on too long becomes counterproductive. This problem was first recognized and described by Freud (1937/1962). What the founder of psychoanalysis did not understand, however, is that patients who are lonely and fulfilled may be reluctant to "graduate" from therapy. Most eventually get past this obstacle, and it is possible to make the termination phase of treatment a productive experience. Yet unfortunately, the "Woody Allen scenario" of endless therapy is not unusual, at least in patients who are either rich or have generous insurance coverage. I have seen quite a few "lifers," as one researcher for the Menninger study of long-term psychotherapy (L. Horwitz, 1974) called such patients.

The best way to get around the problem is never to offer open-ended therapy. All psychotherapies need a contract with a time frame, and careful review must take place before renewing that contract. This focuses the mind of both participants and encourages active collaboration for change. It also protects patients from wasting their time and therapists from feeling trapped into keeping patients until they move away or die.

Another option is to provide intermittent rather than continuous courses of treatment. This idea, first proposed 70 years ago by Alexander and French (1946), is designed to work against the tendency of openended therapy to be addictive. It also leaves the door open for "retreads" when circumstances require another course of treatment (Paris, 2007b). It is best to assume that almost all periods of therapy for PD will be brief.

Why, then, do so many mental health practitioners believe that patients with PD need long-term therapy? The main reason is that the problems associated with these conditions are so severe, it seems impossible to treat patients in a few months. Yet there is little empirical evidence that a long course of illness necessarily requires a long course of treatment. In most cases, even in the more severe PDs, there will have been periods of successful functioning on which treatment can build. Even when this is not the case, empirical data fail to support therapy that goes on for years. In

contrast, a large body of evidence shows that shorter interventions can be effective for a wide range of psychopathology (Crits-Christoph, Gibbons, & Mukherjee, 2013). Therapist time is a scarce resource that should never be wasted.

Psychotherapists need realistic goals. If they are out to "cure" a PD, they will feel impelled to invest much time in each patient, but if they are realistic and accept that a significant reduction in symptoms is a good enough result, they will treat more people and be more effective with all of them. Therapy can stop when patients' own healing mechanisms begin to kick in and when they have learned enough from treatment to continue the recovery process on their own. This principle is consistent with the idea that patients need to develop a sense of agency in their own lives and not to depend on therapists to do everything for them. Similarly, therapists need to be comfortable with the idea that life is a journey and that patients can be discharged without being "cured."

#### THREE BASIC PRINCIPLES FOR MANAGEMENT

## 1. Making a Contract

All therapies require a mutually agreed contract. One should not embark on an expensive, demanding, and time-consuming venture without defining goals and setting a time frame. Like a business plan, a therapy contract encourages focus and reduces drift.

#### 2. Learning New Skills

Much of the work in psychotherapy consists of reassessing problematic behaviors and suggesting alternatives that work better. Instead of detailed exploration of the past, or blanket support, treatment needs a large component of psychoeducation. Teaching skills usually involves unlearning dysfunctional behaviors, and learning more effective interpersonal strategies. As discussed in the previous chapter, Marsha Linehan (1993) pioneered this skill-based approach. These ideas are designed for the treatment of BPD, but the same methods could help a wider range of patients who are

dysregulated and impulsive. Dialectical behavior therapy (DBT) has also been tested for managing substance abuse (Dimeff & Linehan, 2008), a problem often comorbid with BPD and in which emotion dysregulation and impulsivity are also striking. Systems training for emotional predictability and problem solving (Black, Blum, McCormick, & Allen, 2013) and mentalization-based treatment (Allen, Fonagy, & Bateman, 2008), both of which are discussed in Chapter 10, have been tested in forensic populations in which some patients with antisocial PD have similarities to BPD.

## 3. Getting a Life

This is a key principle for treating PD (Zanarini, 2008). No matter what else is addressed in therapy, patients need something to commit to. Freud, when asked what mental health is, said it consists of love and work, but work almost always needs to come first.

Many patients mistakenly believe that "love is the answer," but you have to be somebody before you can love somebody. You cannot find stable love without having an identity of your own. Searching for love without finding yourself raises impossible expectations and usually ends in disaster. Moreover, investing all your emotional capital in one relationship is risky—it gives you nothing to fall back on.

For this reason, patients who come to treatment after the loss of a relationship can be encouraged to take a break from intimacy until they feel better about themselves, and then consider trying again. Although not everyone takes this advice, many will. It is much more important to develop a social role and a social network. Thus, every patient should be encouraged to work, or to go to school to prepare for work. Moreover, to recover, patients need *social capital*, that is, friends and a community (Paris, 2014b).

For these reasons, putting patients with PDs on disability is poison for therapy. Certifying patients for long-term benefits gives up any possibility of rehabilitation. When you sign, you are agreeing that your patient will never be able to work. It can be argued that this can sometimes be true. Yet even the most disabled patients can be asked to volunteer as a way to determine if they have that potential.

# A STEPPED CARE MODEL FOR THE TREATMENT OF BPD

My rationale for treating patients briefly is based on the principle of *stepped care* (Paris, 2013b). This is a model of service delivery in which one begins with less intensive and time-limited treatment to see how helpful that can be. Only when that fails should one consider longer and more expensive therapy.

Since 2001, I have been in charge of a clinic that treats patients with BPD within a 12-week time frame. Each patient receives 24 sessions (12 individual and 12 group). We use a combination of group with psychoeducation and individual sessions, which has been found in most studies to be better than either alone. This short-term program is designed for patients with BPD who are more acute and less disabled and therefore able to benefit from a few months of treatment. The content is a kind of "DBT light" with an eclectic admixture.

Our experience is that among the patients who stay with us for 12 weeks, most will make a fair degree of progress. We have pre—post data on 130 patients seen between 2005 and 2007 that documents improvement in most cases (excluding the 30% of patients who dropped out). Scores on a number of standard measures (Symptom Check List—90, Barratt Impulsivity Scale, and Beck Depression Scale) went down significantly after treatment. Although patients might have improved to the same extent without treatment, most had multiple courses of unsuccessful therapy in the past. Further, if it were true that all patients with BPD need long-term therapy, we would not have seen measurable change within a few months.

We used these findings in 2008 when the hospital asked us to justify the value of our programs; however, pre–post data lack a control group, so we have not published these results. For the same reason that reports claiming to show improvement in long-term therapy require comparisons with briefer forms of treatment, improvement in short-term therapy needs to be compared with naturalistic remission. Nonetheless, we found that emergency department visits and hospital admissions (of particular interest to the hospital administration) went down by about half. We were

able to convince the hospital to fund us, even if we could not reach firm conclusions.

Follow-up telephone calls made a year later to most patients also tended to document further progress. Although there were continuing difficulties, most continued to apply skills they had been taught and felt they were continuing to learn new ones. Thus, although we cannot be sure how effective we are, we tend to doubt that the patients we treat would have improved just as much with no intervention at all.

Treating patients with PD briefly has attracted some research support. Most of the gains in DBT occur fairly quickly (Stanley, Brodsky, Nelson, & Dulit, 2007), and many patients with BPD do well after 25 sessions (Davidson, Tyrer, Norrie, Palmer, & Tyrer, 2010). We therefore treat most of our BPD patients in the 12-week program, particularly those who are younger and less chronic.

Patients who are more disabled can be referred to a longer program designed for the rehabilitation of chronic patients. We also use this as a backup option in a stepped care model for those who fail to respond to 12 weeks of therapy. The separate clinics for extended care have a time scale up to 18 to 24 months. Although results are necessarily more limited in this highly dysfunctional population, pre—post data suggest that most are less symptomatic after treatment. The difference is that although most patients coming to the brief program are working or attending school, many of those coming to long-term treatment are on welfare or long-term disability. We encourage all patients to get back to work (or go to school), and some of them do. If this is not possible, we can still encourage our patients to get more involved in activities, either in the family or in the community.

The main advantage of a stepped care model is that it allows us to manage a large number of patients. I work in general hospitals where there is a constant inflow of patients with BPD through emergency departments and crisis teams. There is certainly no lack of work to be done. In the absence of specialized but brief programs, these patients tend either to move in and out of the system or to be stuck in it without making progress.

I am interested in reaching out to the community, so I carry out several hundred consultations a year. Once seen, patients who are able to

engage in our programs do not have to wait for more than a few weeks for therapy. Most, particularly when younger and more acute, need rapidly accessible interventions. Our program opens up access to patients who would otherwise have to be put on waiting lists for treatment (as happens in other settings offering specialized services). This is not to say that all patients who have been to the emergency department are ready for psychotherapy. They may refuse treatment, fail to come to sessions, or drop out. However, these cases can always be reevaluated at a later time to see whether motivation has become stronger.

A stepped care model allows for intermittent courses of therapy, and patients who come back with serious problems after completing a 12-week program can be offered the longer option. However, if there is no progress after 24 months, we do not offer further treatment because continuing beyond that point is unlikely to be profitable, and we return these patients to community care.

These principles can also be applied to clinical practice outside specialized settings: Most practitioners can apply the stepped care model. It has been established that this is a good way to allocate scarce services in all forms of chronic disease, and the model has been applied to other disorders in medicine and psychiatry that have a variable prognosis (Paris, 2013c). Because prognosis is generally variable, it makes little sense to offer the same treatment to every patient. Many will recover rapidly, and only some become chronically disabled. One cannot know the prognosis at intake, and thus the treatment model moves by steps, with brief therapy as the default condition, reserving resource-heavy interventions for those who fail to recover. The model also allows for intermittent therapy, in which patients are discharged to see how well they make use of the first step. A good rule of thumb is to ask patients to apply what they have learned and to wait 6 months before returning for reassessment.

The discovery that psychotherapy, when properly conducted, helps most people with BPD is one of the most promising developments over my lifetime as a clinician. Younger clinicians may see the same principles applied to all PDs. Yet there will always be patients who do not do well in psychotherapy, or in any other form of treatment. This is a fact we

have to live with: "Radical acceptance" is as necessary for us as it is for our clients. It does not make sense to continue to see treatment failures on a regular basis, because that blocks access to new patients who will gain more benefit.

Although there is only a small literature on the brief treatment of PD (Paris, 2007b), Anthony Bateman, who developed mentalization-based treatment, has told me that he now thinks it may be more useful to treat BPD intermittently rather than continuously. This option works against stalemates, while maintaining availability for later consultation or therapy "retreads."

Although we lack treatment protocols for PDs other than BPD, a stepped care approach might also be applicable for patients with other trait profiles and for the mixed picture seen in PD, not otherwise specified. One common element in all disorders is interpersonal and occupational dysfunction that seriously interferes with quality of life. A stepped care model for all these patients would also emphasize rehabilitation and the accumulation of social capital (Paris, 2014b).

Psychotherapy is too dependent on market forces. Paying patients who can afford to come regularly for months or years may be seen as a plus by clinicians. In contrast, providing evidence-based therapy for patients with PD requires a commitment to accessibility. An all-too-common scenario is that specialized treatment is only available to a small minority: those who have money, those whose families have money, and those who are lucky enough to get in before a clinic is forced to put prospective patients on a waiting list.

Where I work (Canada), psychiatric treatment is insured by the government, but most psychiatrists do little psychotherapy. The main portal of entry into the mental health system is the hospital emergency department. Many patients receive a few sessions of follow-up after that or end up in general follow-up, receiving medication instead of treatment from a team well trained in the specific management of their disorder. By and large, PD patients do not do well in standard outpatient follow-up. Stepped care provides an alternative model, allowing for the allocation of scarce but effective resources to these complex clinical problems.

#### MANAGING SUICIDALITY IN BPD

Chronic suicidality is a hallmark of BPD, and fear of losing a patient to suicide can get in the way of conducting effective treatment. When you are constantly afraid your patients will kill themselves, it is hard to do any useful work. If therapists respond to every suicidal threat as if the patient's life is in danger, treatment comes to a complete standstill.

To treat chronically suicidal patients, clinicians need to understand that the possibility of suicide is part of the territory. Some patients do actually kill themselves, but that is more likely to happen when they are out of treatment. Most of the time, suicidality is a way of coping with pain. Paradoxically, preoccupation with death can itself become a way of life (Paris, 2006).

It follows that clinicians need not overreact to self-harm incidents or to small overdoses that are immediately brought to the attention of significant others. This does not, of course, mean that one should ignore such events. Rather, they need to be understood as ways of expressing distress, albeit dramatically. Instead of a panicky referral to the emergency department, one can conduct a careful inquiry about the source of distress and the sequence of events that led up to the crisis. Sending patients with PD to hospitals has never been shown to have benefit or to prevent suicide. Moreover, hospitalization is often counterproductive because it puts patients in an artificial environment and interferes with life tasks that need to be mastered. It interrupts therapy without accomplishing anything useful; it takes patients away from their problems temporarily but also removes them from the domains in which they are functioning. Hospital wards may offer a breathing space but staying there solves nothing.

Ironically, some patients insist on going to hospital. For most people, the emergency department would be an unpleasant place to spend an evening or to stay overnight. Yet patients whose quality of life is low may prefer this environment to going home. Although most patients in the emergency department agree to leave after being evaluated, some escalate their threats to force staff to keep them. If the cycle continues, patients tend to be fully admitted.

In the United States, managed care has done patients with PD a real favor by refusing to provide coverage for more than a few days of hospital care. However, in the fully insured system in which I work, I have seen patients remain on hospital wards for months when they continue to threaten suicide every time discharge is brought up. When this happens, admission becomes a purely custodial procedure.

Some governments in Europe pay for extended hospital stays for patients with PD, usually in specialized inpatient programs. This is an expensive option that can almost certainly be carried out in other ways. Specialized programs for PD can be better located in day hospitals, as has been shown by Piper, Rosie, and Joyce (1996), as well as by Bateman and Fonagy (2004).

The use of hospitalization for PD is rooted in confusion between the acute suicidality seen in classical mood disorders and the chronic suicidality that characterizes BPD. In the first case, one applies specific methods of treatment that can best be handled on an inpatient unit, where drugs can produce a remission of a temporary state of suicidality. In the second case, patients are admitted but not treated and continue being chronically suicidal once they are discharged. There is no evidence that keeping patients on a hospital ward, even under constant observation, prevents from them from killing themselves in either the short or long term. When suicidality is chronic, death by suicide is a risk that must be accepted (Maltsberger, 1994). In any case, chronically suicidal patients who do commit suicide rarely die in the acute crises that bring them to emergency department. Instead, they are more at risk later in the course of the disorder, when they give up hope after a long series of unsuccessful treatments (Paris, 2003). No therapist wants to lose a patient to suicide, but we do not know how to predict it, even using standard algorithms based on known risk factors (Paris, 2006). In BPD, where some of the risk factors for suicide, such as serious previous attempts and substance abuse (Stone, 1990), are known, statistical relationships do not translate into prediction because these effects are too small to have any useful clinical application.

Understanding chronic suicidality as a form of communication offers a way out of these dilemmas. Patients who threaten and attempt suicide

feel unheard and invalidated; but when they talk of death, significant others and mental health professionals will listen. This is not to dismiss suicidality as mere attention seeking. Suicidal ideas need to be taken seriously, but not in the way that one might think—that is, doing something radical to prevent suicide. Instead we need to provide empathy for the suffering that makes patients consider ending their lives.

Thus, one of the most useful responses therapists can make to a suicidal threat is to reflect on how bad the patient must be feeling to consider that option. It is important to validate emotions, even when a therapist feels under pressure. Empathy also makes it easier to move to a problem-solving mode. This can be framed in statements such as, "You can always kill yourself, but let's see if we can come up with alternatives to help you through the crisis you're in now." There are also times when patients accuse a therapist of not caring enough to stop them from committing suicide. The answer has to be, "In the long run, I can't stop you, but I believe we can work together to find another solution."

These responses provide implicit validation of autonomy. Patients can be attached to their suicidality and find it comforting. They are, to borrow a phrase from John Keats, "half in love with death." Their life is more tolerable if they know that, if necessary, they can leave it. For this very reason, therapists need to accept suicidality to treat it.

Therapists are understandably terrified about suicide, but it is a tragic experience that will happen, at least occasionally, to almost any clinician who works with seriously ill people (Chemtob, Hamada, Bauer, Torigoe, & Kinney, 1988). It is part of the landscape of psychotherapy. Because no one knows how to predict suicide, clinicians working in this context must live with the risk. The only effective ways of preventing suicide that have been supported by research are population-based interventions that make it more difficult to access the means to kill oneself (Paris, 2006).

In our litigious society, therapists are also afraid of being sued by families if their patients die by suicide. Although this possibility cannot be ruled out, there are ways to make such an outcome less likely (Gutheil, 2004). The first is to keep careful notes, explaining in detail why one is intervening (or not intervening). A second is to obtain consultation from a trusted colleague. The third, and most important, is to bring families

into therapy at an early point to explain the risk of suicide and gain their cooperation in a plan that aims to treat the patient rather than consign him or her to a cycle of repeated admissions.

Chronic suicidality is wearing for therapists, some of whom try to avoid taking on patients who appear to be at high risk. Yet patients who improve sufficiently to discard the option of suicide can be one of the most validating experiences in the practice of psychotherapy.

In summary, constantly worrying about suicide reinforces pathology by putting the therapist in the position of an anxious caretaker who is responsible for the patient's welfare (just the sort of relationship that chronically suicidal patients tend to have on the outside). In contrast, empathizing with the suffering that makes people want to leave the world is helpful and avoids agreeing with suicidal patients that feelings can be replaced with actions.

#### CLINICAL EXAMPLES

I now provide illustrations of how some of the principles described in this chapter can be applied to patients with PD. I have drawn all these examples, which have varying outcomes, from our 12-week program for BPD.

#### **Patients Who Made Significant Progress**

Arlene was a 22-year-old woman referred following a severe overdose, after which she made further suicidal threats. The precipitant was a breakup with a boyfriend, part of a long-term pattern of infatuation with unsuitable men. She was, however, able to be competent and stable in her work as a secretary.

In individual therapy, Arlene came to understand that her problems associated with feeling rejected by men were related to an absent father. Her biological father had moved to another city when she was a child and forgot about her. But Arlene was also in chronic conflict with her mother and had a difficult relationship to her stepfather. In group therapy, she was taught a number of skills that follow the principles of DBT and are part

of our treatment module: emotion regulation, appropriate assertiveness, and thinking before acting out impulsively.

Arlene made a change by moving out of the house. She also realized she needed a break before becoming involved with another boyfriend and developed a social network with other women. At 1-year follow-up, she had returned to school and was no longer contemplating suicide.

Shirley was a 25-year-old nurse. Symptoms included cutting, suicidal ideas, and chronic depression with insomnia. She was living with her mother and stepfather. Her relationships to men were impulsive and unstable. In individual therapy, Shirley described a problematic relationship with her father, a physician who was both inconsistent and rejecting, much like her boyfriends. In group therapy, she was taught how to manage her emotions to avoid cutting and was able to find better ways of soothing herself, primarily through increasing the number of female friends in her life. On 2-year follow-up, Shirley had a good job and had found a much more supportive man. Although she continued to have periods of low mood, she no longer carried out self-destructive actions.

Norma was a 24-year-old student who grew up in an Inuit community in the Arctic. She was referred for cutting, suicidality, unstable relations, and olfactory hallucinations (i.e., she thought she had a bad smell). Symptomatic change was associated with finding a stable social network, partly through the Inuit community in the city, where she found role models, and partly from the women she met at school. Norma stopped cutting, and decided to take a break from men. On 1-year follow-up, she maintained these gains and planned a career in social services.

## Patients Who Improved Up to a Point

Maureen was a 24-year-old farmworker referred for suicidality, cutting, and chronic depression. Eighteen months earlier, Maureen had gone through a divorce, followed by a series of unsatisfying relationships with men, finally moving back with her parents. She was only able to work part time on a dairy farm.

Maureen was motivated enough to do a 2-hour drive to come to the city for treatment. In individual therapy, one key issue was a history of

sexual abuse by a cousin that lasted over 4 years. This had occurred because of emotional isolation within the family, in which she felt entirely misunderstood, but she also felt scarred and stigmatized.

In group therapy, Maureen was encouraged to overcome her difficulty in being intimate and trusting and to overcome the feeling that no one could ever find her special. Maureen was somewhat improved after therapy: She was able to keep her job and stopped seeing an uncommitted boyfriend. On 6-month follow-up, however, she felt lonely and pessimistic about stable intimacy.

Maria was a 26-year-old woman living alone on welfare, having recently dropped out of university. Her symptoms include cutting, suicidal threats, substance abuse, and chronic shoplifting (for which she faced a court date). After her parents divorced, her mother was murdered by a lover. Maria had also been involved with abusive men, and although she currently had a supportive boyfriend, she was not faithful to him.

Maria attended the program regularly and at discharge was less impulsive, particularly in relation to alcohol intake. On 6-month follow-up, she had made plans to go to school to learn how to be a hairdresser. However, it was not clear that she would be able to establish a stable intimate relationship.

#### Patients Who Failed to Respond to Treatment

Jean was a 25-year-old woman referred to our PD program for chronic suicidality. Jean has not been able to make any commitments in life and became heavily involved in drinking and drugs. She broke up with a boyfriend and grieved for him, although they had spent most of their time together taking cocaine. Jean came from a middle-class family but rejected "conventional" values without being able to come up with an alternative or a direction.

Jean regarded treatment with ironic detachment, not getting emotionally involved. Although she was able to moderate her substance abuse, she did not seriously look for work and continued to depend on her parents financially.

Karen was a 42-year-old unemployed woman who was referred after an overdose. The crisis was associated with the loss of a boyfriend who

moved away to another city. Karen had little life of her own and never had. She had long been involved with substance abuse, and her only work history was as an escort. Karen did not feel a strong need for treatment and dropped out of our program after a few weeks.

Although the results of a program such as ours are inevitably variable, our team is satisfied to be able to help a good percentage of those who enter it. We know that not every patient with a PD is treatable at any given time and that this is a difficult population, so we accept limitations. Good responses to brief treatment are gratifying if therapists keep expectations reasonable and value small victories. By and large, those who do best are those who have shown previous strengths in work and in relationships. Motivation for change and persistence, however difficult to measure or predict, may also determine outcome.

Most patients in research studies have been documented to show symptomatic improvement, but we are skeptical of claims of personality change, usually based on a few highly successful cases. Rather, we aim to help patients find a way to make their personality work better. To be emotionally labile is not necessarily a bad thing, particularly when one no longer responds to one's sensitivity by cutting, overdosing, or using substances.

### A PHILOSOPHY OF MANAGEMENT

The management of PD need not be an exercise in perfectionism. Many patients go into remission and may even offer therapists heartwarming gratitude. Those who make a partial recovery should still be counted as successes. I am trained in medicine, in which most pathology is chronic and cures are exceptional. As for cases who fail to respond to reasonable efforts at treatment, there is no point banging our heads against walls. (As some of our patients do.)

This is the philosophy that makes me happy to treat difficult patients. I want challenging work that requires the skills I have learned over several decades, but I do not want to trap myself in unreasonable expectations. My hope is that the readers of this book will give up looking for absolute answers and accept that if they can help most of the people they see, they are doing well.



## Summary and Future Directions

# WHERE WE HAVE COME FROM AND WHERE WE ARE NOW

The concept of a personality disorder (PD) is relatively recent but is consistent with theoretical and empirical work in psychology. The construct emerged from psychoanalytic theories about the structure of the mind but is independent of that model, now rooted in trait psychology. The principle that both personality and PD arise from gene—environment interactions has been confirmed by behavior genetics, and outcome research has shown that these diagnoses are stable. Finally, the concept of PD is consistent with cognitive neuroscience and with research on emotion regulation.

Although much still needs to be learned about the causes of PDs, what we know generally fits into a biopsychosocial model. Thus, biological temperament, life experiences, and the social environment all play

a role in etiology. Research has to take all of these factors into account. It also requires a better understanding of the relations among heritable temperament, the development of stable traits, and the risk for PD. This line of research requires longitudinal data. For example, one of the more important recent findings is that temperament can produce *differential susceptibility* to the environment, so that some personality characteristics can lead to either disorder or to better functioning, depending on what happens to people over the course of their lives (Belsky & Pluess, 2009). These findings confirm the view, best developed by Beck and Freeman (2002), that our personality can either work for us or against us, depending on circumstance.

The other recent development that has led to controversy about PD concerns the *Diagnostic and Statistical Manual of Mental Disorders* (5th ed.; *DSM*–5; American Psychiatric Association, 2013). Although the current classification of PDs leaves a great deal to be desired, it is neither better nor worse than the systems we use to classify other major mental disorders. I have a larger concern. Clinicians have not yet embraced the complexity of personality and PD. Too often they prefer the simplicity of symptoms seen as a clearer focus for treatment.

The greatest progress in PD research in recent decades has concerned outcome and treatment. The idea that PDs are incurable can now be firmly rejected. It was based almost entirely on the "clinician's illusion," in which illness is seen as more chronic than it really is. Careful follow-up of patients with a range of PDs shows that most improve with time. This finding could reduce the stigma of PD in which perceptions of hopelessness have interfered with the recognition of these conditions. This is not to say that patients with PDs do not continue to have problems—most of them do—but we all have problematic traits, and most of us manage to live productive lives in spite of them.

Research showing that borderline PD (BPD) can be successfully treated was a dramatic breakthrough for the field. Most of the credit belongs to Marsha Linehan, who rethought the problem from scratch and came up with creative and highly effective solutions. Others have developed parallel or similar methods, and we now understand the basic principles of

#### SUMMARY AND FUTURE DIRECTIONS

therapy for this population. Of equal importance, ineffective methods of therapy that dominated the field for too long are being gradually discarded.

# FUTURE DIRECTIONS AND UNANSWERED QUESTIONS

Further progress in research on PD is not predictable, but new paradigms could emerge to change the field. Our knowledge could be further extended by following some of the most promising current developments.

#### **Childhood Precursors and Prevention**

Research on the childhood precursors of PD has been developing rapidly, focusing largely on BPD. The etiological pathways that drive personality pathology should be further illuminated by longitudinal studies of populations at risk, whether that risk is due to abnormal temperament or to psychosocial adversity. Although it may never be possible to write an equation or a formula for "cooking" a PD, we could be in a position to define which children are most likely to develop one.

Doing so would allow us to consider methods of prevention and early intervention. For example, early psychoeducational interventions in schools and in the home have been shown to reduce drug use among adolescents at risk (Castellanos-Ryan, Séguin, Vitaro, Parent, & Tremblay, 2013). If addictions, a problem as stubborn as PD (and highly comorbid with PD), can be at least partially prevented, then the same might be accomplished for symptoms of antisocial PD and BPD.

### **Adolescent Onset**

Because we now know that PD begins early in life and because it may have a higher prevalence in adolescence that it does later, it makes little sense to wait until patients are 18 years old to diagnose it. Although adolescents are difficult to engage in therapy, it has been shown that at least some of

them can benefit from early interventions, particularly when they are well thought out and systematic.

Because not everyone examines the manual closely, here is what *DSM*–5 has to say:

For a personality disorder to be diagnosed in an individual younger than 18 years, the features must have been present for at least 1 year. The one exception to this is antisocial personality disorder, which cannot be diagnosed in individuals younger than 18 years. (American Psychiatric Association, 2013, pp. 247–248)

We have an important job to do in convincing clinicians not to dismiss PD as a "phase" that adolescents can be expected to grow out of. Although that sometimes happens, most of these patients continue to have serious problems in adulthood and require both accurate diagnosis and evidence-based treatment.

## **Community Prevalence**

Over the past few decades, epidemiological research has taken PD seriously. If the word can get out that antisocial PD affects 2% to 3% of the population and that BPD can be found in almost 1%, this will increase clinical awareness. The main problem is that all these estimates are based on dicey criteria derived from the *DSM* system. We also need to be careful about overestimates of community prevalence, which could be used to discredit rather than to support PD diagnoses.

#### Clinical Prevalence

One of the most ironic aspects of current practice is that although PDs are common, they often go unrecognized. We need to develop simpler and more user-friendly ways of assessing PDs. It may be even more important to teach clinicians that they ignore them at their peril and that patients suffer if life histories are not taken and long-term psychopathology is not properly considered.

#### SUMMARY AND FUTURE DIRECTIONS

## **Biological Research**

It is difficult to treat PDs if we do not understand them. To solve these problems, much hope has been placed on neuroscience. We now know much more about how the brain functions. Unfortunately, despite the billions of research dollars spent on biological research in the past 20 years, it has told us little of direct relevance to the treatment of mental illness. The investment may pan out eventually, but there are good reasons to be cautious. The clinical phenomena that characterize PD lie at a level of complexity that will be difficult to correlate with changes in brain structure of functioning as measured by imaging. We need more basic knowledge to come up with hypotheses that are specific to any disorder. In the meantime, more needs to be done to study key variables such as affective instability.

#### **Gene-Environment Interactions**

The phenomena that characterize PDs are difficult to correlate with genes or with changes in the epigenome. Yet not enough research has been conducted on the gene—environment interactions that drive psychopathology. The obstacles are that genetic knowledge is still fairly primitive and we lack sophisticated and accurate ways of measuring environmental factors. Simplistic approaches, in which genetic variants are linked to broad developmental risk such as "abuse," have failed to address this problem. Further, because an outcome of PD is marked by equifinality, we will need large samples to sort out the interactions that raise risk for serious psychopathology.

## **Making Treatment Accessible**

Current treatments of PD, even when evidence based, have a long way to go to become accessible. The best studied method, dialectical behavioral therapy, is expensive and can potentially go on for years, much like psychoanalysis. If it has not been tested for more than 1 year, then it

should be offered for only 1 year. Briefer interventions, along the lines of systems training for emotional predictability and problem solving, need to be further developed. Putting patients on waiting lists or requiring them to be wealthy is no way to apply the progress that has been made in treating BPD.

There are few things as inexpensive as prescribing a drug. This is one of the reasons for the decline of psychotherapy as a whole. If we want practitioners to treat PDs, we need to develop methods that help patients within a reasonable time. If there is flow in the system, more patients will be able to enter it.

### **Extending Treatment Research to Other PDs**

It is understandable that research on treatment has focused on BPD, but many of the same principles could be applied to other disorders. What we need are treatment packages specifically designed for the specific problems that each category of disorder presents. One prime candidate for this development is narcissistic PD.

### **Developing a Better Classification**

I have left this issue almost for last. Some researchers want to start with a new system and see where it leads. I would argue that it could be a waste of time to tinker with classification until we know much more. When physicians of the past studied swelling and pain, they made little progress; it was only when they understood the mechanisms behind disease that they could begin to classify it scientifically. Until we can begin to answer such questions, attempts to develop a new diagnostic system may be futile.

#### **Raising Awareness and Recognition**

The most urgent need for PD researchers and clinicians who treat these conditions is to raise public awareness of the problem. There is no drug to

#### SUMMARY AND FUTURE DIRECTIONS

treat these patients, and thus the pharmaceutical industry is not interested in helping with this agenda. Psychotherapy is the most effective form of treatment for PD. Yet as long as treatment can only be carried out in specialized settings, caregivers, patients, or families may not see it as relevant. These are all reachable goals, and we can expect slow but steady progress in the coming decades. Our patients depend on us to do so.

## FOUR MYTHS (AND REALITIES) ABOUT PD

## Myth 1: Personality Cannot Change.

Reality: Although personality traits are relatively stable over time, they do show gradual change, mostly for the better. With time, most impulsive people learn to control themselves, anxious people gradually expose themselves to what they fear, and compulsive people learn to give up some degree of control. This is not to say that these trends are universal or that there is no such thing as PD in old age; of course there is. By and large, however, people change for the better as they grow old. Life's greatest distresses occur in the young and the middle-aged, and old age can temper the fires of youth with acceptance and serenity.

## Myth 2: PDs Are Not Mental Illnesses.

*Reality:* The suffering associated with personality pathology is as severe as in other major mental disorders. There is a tendency to dismiss the problem, or redefine it as due to more familiar constructs such as depression or anxiety. We do this only because we are not always familiar with the specific approaches that can work for these patients.

#### Myth 3: Personality Disorders Are Incurable.

*Reality:* As we have seen, most people with PD improve gradually, albeit with some degree of residual dysfunction. The idea that PD does not get better is due to the clinician's illusion in which recovered patients stop coming for help, whereas unrecovered patients keep asking for help.

## Myth 4: Patients With PDs Cannot Improve in Brief Therapy.

*Reality:* This belief has led to lengthy, expensive, ineffective, and frustrating therapies. The best becomes the enemy of the good. If you are a seasoned clinician, you will probably have learned to accept small victories, to not try for ideal outcomes, and to be satisfied with helping people to function in life.

The realities listed here are fundamental to a philosophy of treatment for PD. We will learn how to help patients more effectively in the future, but we already know enough to take on difficult cases and to make a real difference.